

MONA OFFSHORE WIND PROJECT

Book of Reference F06_F07 (tracked)

Deadline: 7 **Application Reference: EN010137 Document Reference: D4** Document Number: MOCNS-J3303-DMC-10005 14 January 2025 F06_F07



Document status

Version	Purpose of document	Authored by	Reviewed by	Approved by	Review date
F01	Application	Dalcour Maclaren	Mona Offshore Wind Ltd	Mona Offshore Wind Ltd	June 2024
F02	Submission at S51	Dalcour Maclaren	Mona Offshore Wind Ltd	Mona Offshore Wind Ltd	July 2024
F03	Submission at 1	Dalcour Maclaren	Mona Offshore Wind Ltd	Mona Offshore Wind Ltd	August 2024
F04	Submission at D2	Dalcour Maclaren	Mona Offshore Wind Ltd	Mona Offshore Wind Ltd	27 August 2024
F05	Submission at D3	Dalcour Maclaren	Mona Offshore Wind Ltd	Mona Offshore Wind Ltd	30 Sept 2024
F06	Submission at D6	Dalcour Maclaren	Mona Offshore Wind Ltd	Mona Offshore Wind Ltd	20 Dec 2024
F07	Submission at D7	Dalcour Maclaren	Mona Offshore Wind Ltd	Mona Offshore Wind Ltd	14 Jan 2025
Prepared	Prepared by:		ed for:		
Dalcour Maclaren		Mona	Mona Offshore Wind Ltd.		



Introduction

- 1. This Book of Reference accompanies the application for a Development Consent Order for the Mona Offshore Wind Farm ("the authorised project"), as required by Regulation 5(2)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 ("the Regulations").
- 2. The land described in this Book of Reference, which is to be subject to powers of compulsory acquisition and/or temporary possession ("the Order Land"), is required to deliver the works comprising the authorised project which are described in detail in Part 1 of Schedule 1 of the draft Development Consent Order ("the dDCO" or "the Order") and shown on the Works Plans (Document reference C1, C2, B3 and B4). This Book of Reference is part of the application documents for the authorised project and should be read in conjunction with the Land Plans, the Crown Land and the Special Category Land Plans (Document references B5, B6, and B7), the Statement of Reasons (Document reference D3) and the dDCO (Document reference C1 and C2).
- 3. Every parcel of land that is affected is identified on a plot by plot basis and a unique reference number has been ascribed to each plot. The plots are shown on the Land Plans which accompany the dDCO. For each plot the Book of Reference identifies whether the Applicant (Mona Offshore Wind Limited or "the undertaker") is seeking the power to acquire that plot outright, the power to create and/or acquire permanent rights, or the power to create and/or acquire temporary rights of possession and use. It also lists the persons with land outside the order limits who may be entitled to make a relevant claim for compensation due to the effects of construction or when the authorised project is operational.
- 4. The dDCO seeks powers to compulsorily acquire land and new rights (both temporary and permanent) for the purposes of the construction, operation and maintenance of the authorised project. Some of the plots identified in Part 1 of the Book of Reference will be subject to acquisition of permanent rights (including imposition of restrictive covenants) pursuant to Article 22 (Compulsory acquisition of rights) and Schedule 8 (Land in which only new rights etc. may be required) of the dDCO. These plots are shown coloured blue on the Land Plans.
- 5. Plots that are subject to powers of temporary possession only, such as for the purpose of access and/or use as a temporary construction compounds or areas only required during construction are listed in Schedule 7 of the order and shown coloured yellow on the Land Plans.
- 6. Plots that are subject to rights are shown coloured blue on the Land Plans, with those required for the purpose of hedgerow enhancement coloured green. All such plots are listed in Schedule 8 of the Order.



7. The colours shown on the Land Plans indicate the type of acquisition sought as set out in Table 1 below.

Table 1:

Colour of the plot on the land plans	Description of Acquisition sought in Book of Reference	Acquisition sought	Principal Relevant dDCO Article (s)
Pink	"Acquisition of the Freehold"	Compulsory acquisition of all interests and rights in land (including as required, subsoil, surface land)	Article 20
Blue	"Acquisition of new rights and imposition or restrictions" & "Acquisition of new rights"	Creation and compulsory acquisition of new rights (including where necessary, a right to impose restrictive covenants)	Article 22
Yellow	"Temporary possession"	Temporary possession and use of land, primarily during construction for up to 12 months from completion of the relevant work	Article 29
Green	"Acquisition of new rights for Hedgerow Enhancement"	Creation and compulsory acquisition of new rights (including where necessary a right to impose restrictive covenants) for hedgerow enhancement and maintenance.	Article 22

8. The Applicant has taken the cautious approach of seeking powers of compulsory acquisition (or rights of use) in respect of all plots of land required for the authorised project even where it already holds an interest or presumes it holds an interest in the land. This approach has been taken to ensure that the Applicant has the ability to acquire the interests it needs in the whole of the Order Limits, even where an unidentified owner later asserts an interest in land which the Applicant believes it owns or has rights.

Rights which may be acquired

9. For plots in which new rights are to be acquired and/or restrictive covenants are to be imposed, the rights sought for the authorised



project have been categorised as shown in Table 2 below. This means that where the entry in the Book of Reference specifies for example that "all cable rights" are to be acquired, all of the rights shown in the cable rights entry of the table will be sought. Where only some of a category of rights is required, the specific rights sought will be listed for the relevant plot to ensure that these are not greater than is required.

10. The dDCO (document reference C1 and C2) seeks powers to override existing easements and other rights and for the extinguishment of private rights in land which is compulsorily acquired. The purpose of these powers is to enable the construction, operation, maintenance and decommissioning of the authorised project.

(1) Number of plot shown on the Land Plans	(2)Purpose for which rights may be acquired
02-033; 03-037; 03- 045; 03-047; 03- 049; 03-050; 03-	Cable rights and restrictive covenants Cable rights
060; 03-062; 03- 063; 04-067; 04- 070; 04-074; 04-	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
078; 05-080; 05- 083; 05-084; 05- 087; 05-088; 05-091; 05-093; 06-097; 06- 100; 06-103; 06-105; 06-106; 06-106a; 06- 108; 07-109; 07-111; 07-117; 07-120; 07- 125; 07-127; 07-129;	(a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables (including the removal of materials including spoil) in, under, over and/or on the land, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers, cable clamping and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling;

Table 2:



07-133; 08-136; 08-146, 08-150; 08-154; 09-157; 09-158; 09-159; 09-168; 09-169; 09-171; 09-172; 09-174; 09-175; 09-176; 09-177; 09-178; 10-179; 10-180; 10-183; 10-184; 10-185; 10-188

(b) enter, be on, and break open and break up the surface of the land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables;

(c) to benefit from continuous vertical and lateral support for the authorised project;

- (d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting, removing, testing, upgrading and replacing the cables and connection into any adjacent cables and associated works, to take plant and equipment on to adjoining land;
- (e) construct and install the cables and thereafter use the land for all necessary purposes for the construction, commissioning, testing, repair and maintenance of the cables in, on or under the land;
- (f) place and use plant, machinery, structures and temporary structures within the land for the purposes of the installation, construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of the cables and to erect temporary signage and provide measures for the benefit of public and personnel safety;
- (g) erect fencing, gates, walls, barriers or other means of enclosure, and create secure working areas and compounds including trenchless installation technique compounds and working areas;
- (h) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair or renewal is being carried out;
- (i) effect access to the highway including creation of temporary visibility splays;
- (j) make such investigations in or on the land as are required;
- (k) fell, lop, cut or remove or coppice wood, uproot trees or hedges or shrubs which now or hereafter may be standing on the land or other land which would if not felled, lopped, cut or removed would obstruct or interfere with the installation and operation of the cables;
- (I) remove and discharge water from the land, and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains,



conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or cleanse drainage schemes on the land or reinstate any existing drainage scheme on the land;

- (m) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus, public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers);
- (n) remove fences and structures within the land during any period in which construction, maintenance, repair or renewal is being carried out (subject to erection of any temporary stock-proof fencing as is reasonably required and the re-instatement or suitable replacement of the fences or structures following the exercise of the rights);
- (o) store and stockpile materials (including excavated material);
- (p) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and monitoring the use of any trenchless installation technique, to keep in place and monitor the same through construction, maintenance, repair, replacement or decommissioning and reinstatement of the land;
- (q) to excavate materials below ground level, including soils, and to store and re-use or dispose of the same, and in so excavating to undertake any works, including works of protection or removal of archaeological remains as may be required by any written scheme of investigation approved under this Order;
- (r) lay out temporary paths and bridleways for public use as temporary diversions for public rights of way which are interfered with during any period in which construction, maintenance, repair, renewal or decommissioning is being carried out;
- (s) to install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, and remediation works for environmental or ecological mitigation or enhancement works, including temporary works for noise alleviation measures and the installation of temporary barriers for the protection of fauna;



	Partners in UK offshore
	(t) carry out such works (together with associated fencing) required by a planning permission and/or consent now or to be granted over the land and/or in accordance with any necessary licences relating to protected species and/or wildlife;
	(u) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, maintain and inspect replacement underground cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and other conducting media together with conduits or pipes for containing the same in and under the land;
	(v) to construct, use, maintain and improve a permanent means of access including visibility splays, and retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the land, adjoining land and the highway; and
	(w)erect temporary bridges and supporting or protective structures for the purposes of access to adjoining land.
2. F	Restrictive covenants
A r	restrictive covenant over the land for the benefit of the remainder of the Order land to:
	(a) prevent anything being done in or upon the land or any part thereof for the purpose of the erection of any buildings or construction, erection or works of any kind (including the foundations or footings thereto);
	(b) prevent anything being done by way of hard surfacing of the land with concrete of any kind or with any other material or surface whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage to relevant part of the authorised project nor make it materially more difficult or expensive to maintain the authorised project);
	(c) to prevent anything to be done by way of excavation of any kind in the land or any activities which would alter, increase or decrease ground cover or soil levels in any manner whatsoever save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.6m for the purposes of arable farming) or are required to be carried out by any statutory



	undertaker in order to exercise their statutory functions or rights in relation to their apparatus (if any) within the land without the consent in writing of the undertaker;
	(d) to prevent the planting or growing within the land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised project nor make it materially more difficult or expensive to access and maintain the relevant part of the authorised project);
	(e) to prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised project;
	(f) to prevent the carrying out of operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised project; and
	(g) to prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or areas of habitat creation or enhancement including any ploughing or grazing without the prior written consent of the undertaker.
02-012; 02-021; <u>02-</u> 021a 02-022; 02-023;	Cable rights and restrictive covenants under existing infrastructure
02-024; 02-025; 02- 026; 02-027; 02-028;	1. Cable rights
02-029; 02-031; 02- 034; 02-035; 02-036; 03-061; 04-065; 04- 073; 05-082; 06- 096; 06-098; 06-101; 07-110; 07-126; 08- 135; 08-140; 08-153; 09-173; 09-173a; 09- 173b; 10-182; 11-189	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
	(a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables (including the removal of materials including spoil) in or under the land, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers, cable clamping and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling;



- (b) enter, be on, and remain on the land with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables;
- (c) to benefit from continuous vertical and lateral support for the authorised project;
- (d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting, removing, testing, upgrading and replacing the cables and connection into any adjacent cables and associated works, to take plant and equipment on to adjoining land;
- (e) construct and install the cables and thereafter use the land for all necessary purposes for the construction, commissioning, testing, repair and maintenance of the cables in or under the land;
- (f) place and use plant, machinery, structures and temporary structures within the land for the purposes of the installation, construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of the cables and to erect temporary signage and provide measures for the benefit of public and personnel safety;
- (g) erect fencing, gates, walls, barriers or other means of enclosure, and create secure working areas and compounds including trenchless installation technique compounds and working areas;
- (h) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair or renewal is being carried out;
- (i) effect access to the highway including creation of temporary visibility splays;
- (j) make such investigations in or on the land as are required;
- (k) fell, lop, cut or remove or coppice wood, uproot trees or hedges or shrubs which now or hereafter may be standing on the land or other land which would if not felled, lopped, cut or removed would obstruct or interfere with the installation or operation of the cables;
- (I) remove and discharge water from the land, and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts,



install, use, inspect, maintain, adjust, alter, renew, repair, test or cleanse drainage schemes on the land or reinstate any existing drainage scheme on the land;

- (m) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers);
- (n) remove fences and structures within the land during any period during which construction, maintenance, repair or renewal is being carried out (subject to erection of any temporary stock-proof fencing as is reasonably required and the re-instatement or suitable replacement of the fences or structures following the exercise of the rights);
- (o) store and stockpile materials (including excavated material);
- (p) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and monitoring the use of any trenchless installation technique, to keep in place and monitor the same through construction, maintenance repair, replacement or decommissioning and to reinstate the land;
- (q) to excavate materials below ground level, including soils, and to store and re-use or dispose of the same, and in so excavating to undertake any works, including works of protection or removal of archaeological remains as may be required by any written scheme of investigation approved under this Order;
- (r) lay out temporary paths and bridleways for public use as temporary diversions for public rights of way which are interfered with during any period in which construction, maintenance, repair or renewal or decommissioning is being carried out;
- (s) to install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance and remediation works for environmental or ecological mitigation or enhancement works, including temporary works for noise alleviation measures and the installation of temporary barriers for the protection of fauna;
- (t) carry out such works (together with associated fencing) required by a planning permission and/or consent now or to be granted over the land and/or in accordance with any necessary licences relating to protected species and/or wildlife;



- (u) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, maintain and inspect replacement underground cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and other conducting media together with conduits or pipes for containing the same in and under the land;
- (v) to construct, use, maintain and improve a permanent means of access including visibility splays, and retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the land, adjoining land and the highway; and
- (w)erect temporary bridges and supporting or protective structures for the purposes of access to adjoining land.

2. Restrictive covenants

A restrictive covenant over the land for the benefit of the remainder of the Order land to:

- (a) prevent anything being done in or upon the land or any part thereof for the purpose of the erection of any buildings or construction, erection or works of any kind (including the foundations or footings thereto) other than those related to works for the benefit of existing highway or railway infrastructure;
- (b) to prevent anything to be done by way of excavation of any kind in the land nor any activities which would alter, increase or decrease ground cover or soil levels by greater than one metre whatsoever without the consent in writing of the undertaker, save where such works are reasonably required for the exercise of statutory functions or rights in relation any public highway or railway on the land and will not damage, undermine or interfere with the cables;
- (c) to prevent the planting or growing within the land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised project nor make it materially more difficult or expensive to access and maintain the relevant part of the authorised project);
- (d) to prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised project; and



	(e) to prevent the carrying out of operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised project.
02-032	Cable rights, transition joint bay rights and restrictive covenants
	1. Cable rights
	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
	(a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables (including the removal of materials including spoil) in, under, over and/or on the land, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers, cable clamping and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling;
	(b) enter, be on, and break open and break up the surface of the land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables;
	(c) to benefit from continuous vertical and lateral support for the authorised project;
	(d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting, removing and replacing the cables;
	(e) construct and install the cables and thereafter use the land for all necessary purposes for the construction, commissioning, testing, repair and maintenance of the cables in, on or under the land;
	(f) place and use plant, machinery, structures and temporary structures within the land for the purposes of the installation, construction, maintenance, repairing, renewing, upgrading,



inspecting, removal and replacing of the cables and to erect temporary signage and provide measures for the benefit of public and personnel safety;

- (g) erect fencing, gates, walls, barriers or other means of enclosure, and create secure working areas and compounds including trenchless installation technique compounds and working areas;
- (h) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair or renewal or decommissioning is being carried out;
- (i) effect access to the highway including creation of temporary visibility splays;
- (j) make such investigations in or on the land as are required;
- (k) fell, lop, cut or remove or coppice wood, uproot trees or hedges or shrubs which now or hereafter may be standing on the land or other land which would if not felled, lopped, cut or removed would obstruct or interfere with the installation and operation of the cables;
- (I) remove and discharge water from the land, and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or cleanse drainage schemes on the land or reinstate any existing drainage scheme on the land;
- (m) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus, public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers);
- (n) remove fences and structures within the land during any period in which construction, maintenance, repair or renewal is being carried out (subject to erection of any temporary stockproof fencing as is reasonably required and the re-instatement or suitable replacement of the fences or structures following the exercise of the rights);
- (o) store and stockpile materials (including excavated material);



- (p) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and monitoring the use of any trenchless installation technique, to keep in place and monitor the same through construction, maintenance, repair, replacement or decommissioning and reinstatement of the land;
- (q) to excavate materials below ground level, including soils, and to store and re-use or dispose of the same, and in so excavating to undertake any works, including works of protection or removal of archaeological remains as may be required by any written scheme of investigation approved under this Order;
- (r) lay out temporary paths and bridleways for public use;
- (s) to install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, and remediation works for environmental or ecological mitigation or enhancement works, including temporary works for noise alleviation measures and the installation of temporary barriers for the protection of fauna;
- (t) carry out such works (together with associated fencing) required by a planning permission and/or consent now or to be granted over the land and/or in accordance with any necessary licences relating to protected species and/or wildlife; and
- (u) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, maintain and inspect replacement underground cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and other conducting media together with conduits or pipes for containing the same in and under the land.

2. Transition joint bay rights

Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to lay down, install, adjust, alter, construct, create, use, maintain, repair, renew, upgrade, inspect, remove and replace transition joint bays.

3. Restrictive covenants

A restrictive covenant over the land for the benefit of the remainder of the Order land to:



_	Partners in UK offshore
	(a) prevent anything being done in or upon the land or any part thereof for the purpose of the erection of any buildings or construction, erection or works of any kind (including the foundations or footings thereto);
	(b) prevent anything being done by way of hard surfacing of the land with concrete of any kind or with any other material or surface whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage to relevant part of the authorised project nor make it materially more difficult or expensive to maintain the authorised project);
	(c) to prevent anything to be done by way of excavation of any kind in the land or any activities which would alter, increase or decrease ground cover or soil levels in any manner whatsoever save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.6m for the purposes of arable farming) or are required to be carried out by any statutory undertaker in order to exercise their statutory functions or rights in relation to their apparatus (if any) within the land without the consent in writing of the undertaker;
	(d) to prevent the planting or growing within the land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised project nor make it materially more difficult or expensive to access and maintain the relevant part of the authorised project);
	(e) to prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised project;
	(f) to prevent the carrying out of operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised project; and
	(g) to prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or areas of habitat creation or enhancement including any ploughing or grazing without the prior written consent of the undertaker.
03-038; 03-046; 03- 048; 03-051; 03-053; 03-054; 03-056; 03-	Cable corridor access rights



	Partners in UK offshore
059; 03-064; 04- 066; 04-072; 04-	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
075; 04-077; 04-079; 05-081; 05-085; 05- 086; 05-089; 05- 090; 05-092; 06-107; 07-115; 07-116; 07- 128; 07-134; 08-147;	(a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of the authorised project, the inspection, testing, maintenance, renewal, upgrading, replacement and removal of the cables and connection into any adjacent cables and associated works, to take plant and equipment on to adjoining land and make such investigations in or on the Land which is ancillary for the purposes of exercise of the rights;
128, 07-134, 08-147, 08-148; 08-148; 08- 151; 08-152; 09-161; 09-170; 10-181; 10-	(b) to construct, use, maintain and improve a permanent means of access including visibility splays, and retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the land, adjoining land and the highway;
186; 10-187; 11-196; 11- 221; 11-231; 11-232; 11- 235	(c) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning is being carried out;
	(d) erect temporary bridges and supporting or protective structures for the purposes of access to adjoining land;
	(e) place and use plant, machinery, structures and temporary structures within the land, and to erect temporary signage and provide measures for the benefit of public and personnel safety;
	(f) fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be present on the land for the purpose of enabling the right to pass and re-pass to adjoining land;
	(g) repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seeding and other ecological measures together with the right to maintain, inspect and replant such trees, shrubs and landscaping;
	(h) erect and remove temporary fencing, gates, walls, barriers or other means of enclosure; and
	(i) lay out temporary paths and bridleways for public use as temporary diversions for public rights of way which are interfered with during any period in which construction, maintenance, repair or renewal decommissioning is being carried out.
03-039; 03-040; 03-	Hedgerow enhancement rights and restrictive covenants



	Partners in UK offshore
041; 03-042; 03-043; 03-044; 07-112; 07- 113; 07-114; 07-118; 07-119; 07-121; 07-	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
122; 07-123; 07-124; 07-130; 07-131; 07-	(a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of the authorised project;
132; 08-141; 08-142; 08-143; 08-144; 08- 145; 08-149; 09-155; 09-156; 11-193; 11-194;	(b) to use, maintain and improve a permanent means of access and retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the landscaping and mitigation land, adjoining land and highway;
11-195 <u>; 11-195a</u>	(c) place and use plant, machinery, structures and temporary structures within the land and to erect temporary signage and provide measures for the benefit of the public and personnel safety;
	(d) fell, lop, cut or remove or coppice wood, uproot trees or hedges or shrubs which now or hereafter may be standing on the land for the purpose of enabling the right to pass and re-pass to adjoining land;
	(e) install, execute, implement, retain, repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seeding and other hedgerow enhancement works together with the right to maintain, inspect and replant such trees, shrubs and landscaping; and
	(f) the installation of temporary barriers for the protection of fauna.
	3. Restrictive covenants
	A restrictive covenant over the land for the benefit of the remainder of the Order land to prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of the hedgerow enhancement works without the prior written consent of the undertaker or as authorised by a planning permission or consent granted over the land.
11-191; 11-192; 11-214	Landscaping and ecological mitigation work rights and restrictive covenants
	1. Landscaping and ecological mitigation works rights



Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—

- (a) install, execute, implement, retain, repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seeding and other landscaping and ecological measures together with the right to maintain, inspect and replant such trees, shrubs and landscaping;
- (b) install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or enhancement works, including temporary works and the installation of temporary barriers for the protection of fauna;
- (c) place and use plant, machinery, structures and temporary structures within the land and to erect temporary signage and provide measures for the benefit of the public and personnel safety;
- (d) fell, lop, cut or remove or coppice wood, uproot trees or hedges or shrubs which now or hereafter may be standing on the land for the purpose of enabling the right to pass and re-pass to adjoining land; and
- (e) carry out such works (together with associated fencing) required by a planning permission and/or consent now or to be granted over the land in accordance with any necessary licences relating to protected species and/or wildlife.

2. Restrictive covenants

A restrictive covenant over the land for the benefit of the remainder of the Order land to prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of landscaping or ecological mitigation measures or works or areas of habitat creation including any ploughing or grazing, during the period within which the undertaker is bound by any consent or licence to maintain that ecological mitigation measures or works or areas of habitat creation, without the prior written consent of the undertaker.



11-221; 11-231; 11-232;	Landscaping and ecological mitigation land access rights
11-235	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
	 (a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the inspection, maintenance, renewal, upgrading, replacement and removal of the landscaping and ecological mitigation works; (b) to use, maintain and improve a permanent means of access and retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the landscaping and mitigation land, adjoining land and highway; and (c) fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be present on the access area for the purpose of enabling the right to pass and re-pass to the landscaping and ecological mitigation land.
<u>11-198;</u> 11-200; 11-201; 11-202; 11-203; 11-204;	National Grid connection rights and restrictions
11-205; 11-206; 11-207; 11-226; 11-227	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to—
	(a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables (including the removal of materials including spoil) in, under, over and/or on the land, together with such telemetry and fibre-optic lines, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers, cable clamping and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables (the "cables"), and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling;



- (b) enter, be on, and break open and break up the surface of the land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables;
- (c) to benefit from continuous vertical and lateral support for the authorised project;
- (d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting the authorised project and for removing and replacing the cables;
- (e) to use, maintain, renew improve and alter existing accesses, roads, streets, tracks or ways over the land, providing that such use is not exclusive and exercise of this right must not prevent or unreasonably inhibit use by other parties;
- (f) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning is being carried out;
- (g) place and use plant, machinery, structures and temporary structures within the land, and to erect temporary signage and provide measures for the benefit of public and personnel safety;
- (h) fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be present on the land for the purpose of enabling the right to pass and re-pass to adjoining land;
- (i) remove and discharge water from the land and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, to lay down, install, adjust, alter, construct, create, use, maintain, repair, renew, upgrade, inspect, remove and replace a drainage scheme on the land; and
- (j) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers).



Restrictive covenants

A restrictive covenant over the land for the benefit of the remainder of the Order land to:

- (a) prevent anything being done in or upon the land or any part thereof for the purpose of the erection of any buildings or construction, erection or works of any kind (including the foundations or footings thereto);
- (b) prevent anything being done by way of hard surfacing of the land with concrete of any kind or with any other material or surface whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage to relevant part of the authorised project nor make it materially more difficult or expensive to maintain the authorised project);
- (c) to prevent anything to be done by way of excavation of any kind in the land or any activities which would alter, increase or decrease ground cover or soil levels in any manner whatsoever save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.6m for the purposes of arable farming) or are required to be carried out by any statutory undertaker in order to exercise their statutory functions or rights in relation to their apparatus (if any) within the land without the consent in writing of the undertaker;
- (d) to prevent the planting or growing within the land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised project nor make it materially more difficult or expensive to access and maintain the relevant part of the authorised project);
- (e) to prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised project;
- (f) to prevent the carrying out of operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised project; and



	(g) to prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or areas of habitat creation or enhancement including any ploughing or grazing without the prior written consent of the undertaker.
11-198; 11-221; 11-222; 11-223; 11-224; 11-225; 11-231; 11-232; 11-235	 National Grid substation access rights Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised project to— (a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, inspecting the authorised project and for removing and replacing the cables and substation; (b) to use, maintain, renew improve and alter existing accesses, roads, streets, tracks or ways over the land, providing that such use is not exclusive and exercise of this right must not prevent or unreasonably inhibit use by other parties; (c) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning is being carried out; (d) place and use plant, machinery, structures and temporary structures within the land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; (e) fell, lop, cut, coppice wood, uproot trees or hedges or shrubs which now or hereafter may be present on the land for the purpose of enabling the right to pass and re-pass to adjoining land; (f) remove and discharge water from the land and to imstall, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts,
	to lay down, install, adjust, alter, construct, create, use, maintain, repair, renew, upgrade, inspect, remove and replace a drainage scheme on the land; and



(g) install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers).

- 11. This Book of Reference is comprised of five Parts, in accordance with Regulation 7(1) of the Regulations, as follows:
 - a. Part 1 contains the names and addresses of those who own, lease, occupy or have another interest in the land that will be affected by the authorised project and the rights contained in the dDCO. A person is within Category 1 if the Applicant after making diligent inquiry, knows that they are an owner, lessee, tenant, or occupier of the land. A person is within Category 2 if they are interested in the land or have the power to sell and convey or release the land. Parties recorded in Part 3 of this Book of Reference have also been included in Part 1.
 - b. Part 2 lists persons who may be entitled to make a relevant claim, also called "Category 3" persons. A person is within Category 3 if the Applicant believes that, if the Order were to be made and fully implemented, they would or might be entitled to make a relevant claim as defined in section 57(6) of the 2008 Act. A relevant claim is a claim under section 10 of the Compulsory Acquisition Act 1965, a claim under part 1 of the Land Compensation Act 1973, or a claim under section 152(3) of the 2008 Act. This part therefore contains the names and addresses of those who have an interest in the Order land and may be entitled to claim compensation for loss resulting from the implementation of the Order and use of the authorised project. This part also contains the names and addresses of those without an interest in the Order land, who may be entitled to claim compensation for loss resulting from the Order and use of the authorised project.
 - c. Part 3 contains the names and addresses of those entitled to enjoy easements or other private rights which it is proposed may be extinguished, suspended or interfered with in connection with the authorised project, pursuant to the Order;
 - d. Part 4 identifies plots in which there is a Crown interest that will be affected by the authorised project and the rights contained in the Order; and



- e. Part 5 identifies plots which constitute "special category land" for the purposes of section 132 of the Planning Act 2008 that will be affected by the authorised project and the rights contained in the Order. This is land
 - i. the acquisition of which is subject to special parliamentary procedure.
 - ii. which is special category land
 - iii. which is replacement land

Given the nature of the works to be undertaken, the Applicant believes that the Order land listed within Part 5, when burdened with the rights contained in the Order, will be no less advantageous than it was before to the following persons-

- i. the persons in whom it is vested
- ii. other persons, if any, entitled to rights of common or other rights, and
- iii. the public

The Applicant is therefore of the opinion that while it has identified open space within the Order land, there is no requirement for the provision of replacement or exchange land.

Book of Reference notes

- 12. Part 1 of this Book of Reference provides the area in square metres of all land included in the dDCO. Areas have not been repeated in Parts 2 to 5 of this Book of Reference.
- 13. The term "approximately" is used before all plot area measurements as these measurements are given in square metres and each measurement is rounded up to the nearest whole square metre.
- 14. The Tracked Version of the Clean Book of Reference highlights the updates to land interest information since the previous submission of the Book of Reference (AS-015);
 - a. Those entries struck through and highlighted in red have been removed
 - b. Those entries highlighted in green have been added; and
 - c. Those entries highlighted in orange have been updated, name, address, or qualifier.

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire						
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
01-001	Temporary Possession	Mean High Water (north of the A55, Abergele) (excluding all	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of the right to lay and maintain sea outfall pipes granted by the Deed dated 20 February 1955 and in respect of the rights to place and maintain groynes as stated in the Deed dated 1 May 1964 and in respect of the rights contained in the Deed dated 4 April 1973) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of the Deed dated 01 April 1997)	
01-002	Temporary Possession	Mean High Water (north of the	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	NONE	
01-003	Temporary Possession		The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access)	

	Land v	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-003 cont'd			Unknown			Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground sewerage apparatus) Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) Unknown
01-004	Temporary Possession	167 square metres of foreshore landward of the Mean High Water (north of the A55, Abergele)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as reputed owner) Unknown	NONE	Unknown	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water and sewerage apparatus) Unknown
01-005	Temporary Possession	2946 square metres of public highway (Sea Road, Abergele)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture)

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to only and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-005 cont'd			Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of subsoil beneath half width of public highway) Unknown		Unknown	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water and sewerage apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown
01-006	Temporary Possession	395 square metres of private road (north of the A55, Abergele)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as reputed owner) Unknown	NONE	Unknown	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 OLT (in respect of underground water and sewerage apparatus)

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	uisition or use Description of land	Qualifying persons under Regulation 7(:	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-006 cont'd						Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown
01-007	Temporary Possession	101 square metres of private road (north of the A55, Abergele) (excluding all interests of the crown)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water and sewerage apparatus)

	Land w	hich is proposed to be subject to	BOOK OF REI c: (i) powers of compulsory acquisition, (ii) r	ed Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-007 cont'd						Julie Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012) Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) Michael Williams 162 Ffordd Y Foryd Bae Cinmel Y RHYL LL18 5LS (in respect of rights of access to maintain services as contained in a Lease dated 2 June 2006) Nicholas Stephen Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012)

	Land wł	nich is proposed to be subject t	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order :RENCE - PART 1 ;ht to use the land, and/or (iii) rights to o y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	nt of acquisition or use Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Fians			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-007 cont'd						Pjan Limited Unit 32 Llys Edmund Prys St. Asaph Business Park ST. ASAPH Flintshire LL17 0JA (in respect of rights of access as contained in a Lease dated 3 November 2010) Sian Williams 5 Terfyn Court Terfyn Bodelwyddan RHYL Denbigshire LL18 5SW (in respect of rights contained in a Deed dated 21 September 2016) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964) Unknown (in respect of rights contained in a Conveyance dated 29 July 1935)

	Land v	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (i	nited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) right: onwy and Denbighshire	to carry out protective works (Regulation 7(1	.)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Ap Regulations 2009	olications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-008	Temporary Possession	325 square metres of private road (north of the A55, Abergele)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU Unknown	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground sewerage apparatus) Julie Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access) Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) Michael Williams 162 Ffordd Y Foryd Bae Cinmel Y RHYL LL18 5LS (in respect of rights of access)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-008 cont'd						Nicholas Stephen Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access) Pjan Limited Unit 32 Llys Edmund Prys St. Asaph Business Park ST. ASAPH Flintshire LL17 OJA (in respect of rights of access) Unknown
01-009	Temporary Possession	road and verges (north of the	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU Unknown (in respect of mines and minerals)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Julie Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012)

	Land wi	nich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ght to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1)(a	Category 1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-009 cont'd						Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) Michael Williams 162 Ffordd Y Foryd Bae Cinmel Y RHYL LL18 5LS (in respect of right of access to maintain services as contained in a Lease dated 2 June 2006) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights contained in a Conveyance dated 29 July 1935) Nicholas Stephen Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012)

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-009 cont'd						Pjan Limited Unit 32 Llys Edmund Prys St. Asaph Business Park ST. ASAPH Flintshire LL17 0JA (in respect of right of access to maintain services as contained in a Lease dated 3 November 2010) Sian Williams 5 Terfyn Court Terfyn Bodelwyddan RHYL Denbigshire LL18 5SW (in respect of rights contained in a Deed dated 21 September 2016) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964)
01-010	Temporary Possession	386 square metres of private road and verges (north of the A55, Abergele) (excluding all interests of the crown)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access)

	Land wł	ich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to o ry and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1)(a	Category 1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-010 cont'd			Unknown (in respect of mines and minerals)			Julie Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012) Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) Michael Williams 162 Ffordd Y Foryd Bae Cinmel Y RHYL LL18 5LS (in respect of right access to maintain services as contained in a Lease dated 2 June 2006) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights contained in a Conveyance dated 29 July 1935)

	Land w	hich is proposed to be subject to	BOOK OF REF c: (i) powers of compulsory acquisition, (ii) r	ed Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights to c wy and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Fians			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-010 cont'd						Nicholas Stephen Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012) Pjan Limited Unit 32 Llys Edmund Prys St. Asaph Business Park ST. ASAPH Flintshire LL17 0JA (in respect of rights of access to maintain services as contained in a Lease dated 3 November 2010) Sian Williams 5 Terfyn Court Terfyn Bodelwyddan RHYL Denbigshire LL18 5SW (in respect of rights contained in a Deed dated 21 September 2016) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964)

	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-011	Temporary Possession	5 (Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (as reputed owner) Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as reputed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner) Unknown	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU		Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground sewerage apparatus) Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) Unknown
02-012	Acquisition of Rights	156831 square metres of foreshore land seaward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	5	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	NONE

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-013	Temporary Possession	167 square metres of private road, hardstanding and grassland (Beach House Road)	Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (as reputed owner) Unknown	NONE	Princes Drive Colwyn Bay CONWY	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown
02-013a	Temporary Possession	58 square metres of private road, hardstanding and grassland (Beach House Road) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner) Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (as reputed owner) Unknown	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF Unknown	Unknown

	Land w	/hich is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition, (nited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights onwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-014	Temporary Possession	road and railway (Beach House Road)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as reputed owner) Unknown	NONE	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as reputed owner) Unknown	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (in respect of rights of access) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (in respect of rights of access) Unknown

	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
02-015	Temporary Possession	1 square metres of grassland (north of Abergele Road, A547)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (as reputed owner) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as reputed owner) Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (as reputed owner)	NONE	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (as reputed owner) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as reputed owner) Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (in respect of rights of access)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (in respect of rights of access) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (in respect of rights of access)	
			Unknown		Unknown	Unknown	
02-016	Temporary Possession	15 square metres of hardstanding and grassland (north of the A55, Abergele) (excluding all interests of the crown)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF	Unknown	

	Land v	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-016 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner) Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (as reputed owner) Unknown		Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay Colwyr LL29 8HT Unknown	
02-017	Temporary Possession	8927 square metres of grassland and track (north of Abergele Road, A547) (excluding all interests of the crown)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (as reputed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF Unknown	Unknown

	Land v	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-017 cont'd			Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (as reputed owner) Unknown			
02-018	Temporary Possession	5529 square metres of hardstanding, grassland and track (north of Abergele Road, A547) (excluding all interests of the crown)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals) Unknown	NONE	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of agreement to grant an easement stated in Deed dated 8 July 1994) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown

	Land v	which is proposed to be subject to	BOOK OF I c: (i) powers of compulsory acquisition, (i	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t onwy and Denbighshire	to carry out protective works (Regulation 7(1	.)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (App Regulations 2009	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-019	Temporary Possession	(north of Abergele Road, A547)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (as reputed owner) Unknown	NONE	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (as reputed owner) Unknown	Unknown
02-020	Temporary Possession	Road, A547)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 &BF (as reputed owner) Unknown	NONE	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (as reputed owner) Unknown	Unknown
02-021	Acquisition of Rights	foreshore land landward of the Mean High Water and cycle way (north of the A55, Abergele) (excluding all	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (as reputed owner)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground sewerage apparatus)

	Land v	which is proposed to be subject to	BOOK OF Ri c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-021 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner) Unknown		Unknown	Unknown
02-021a	Acquisition of Rights	3287 square metres of foreshore land landward of the Mean High Water and cycle way (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner) Unknown	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF Unknown	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground sewerage apparatus) Unknown
02-022	Acquisition of Rights	4484 square metres of grassland (lying to the north of the A55, Abergele)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (as reputed owner) Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (as reputed owner)	NONE	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (as reputed owner) Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (as reputed owner)	Unknown

				ited Development Consent Order		
	Land w	hich is proposed to be subject to	: (i) powers of compulsory acquisition, (ii)	EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-022 cont'd			Unknown		Unknown	
02-023	Acquisition of Rights	665 square metres of grassland (north of the A55, Abergele)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY COLWYM LL29 8BF (as reputed owner) Unknown	NONE	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (as reputed owner) Unknown	Unknown
02-024	Acquisition of Rights	2517 square metres of grassland (north of the A55, Abergele)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF The King's Most Excellent Majesty In- Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals) Unknown	NONE	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of agreement to grant an easement stated in Deed dated 8 July 1994) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown
02-025	Acquisition of Rights	19299 square metres of grassland and railway (north of the A55, Abergele)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW	NONE	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus)

	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
02-025 cont'd			Unknown		Unknown	Unknown	
02-026	Acquisition of Rights	1011 square metres of grassland (north of the A55, Abergele) (excluding all interests of the crown)	The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ	NONE	The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ	NONE	
02-027	Acquisition of Rights	19178 square metres of public highway, verges and hedgerow (A55, Abergele) (excluding all interests of the crown)		NONE	The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ (as highways authority)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Zayo Group UK Limited The Relay Building 114 Whitechapel High Street London E1 7PT (in respect of underground telecommunication apparatus)	

	Land w	which is proposed to be subject to	BOOK OF RE	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1)(a))	
Number on	Extent of acquisition or use	Description of land		Counties of Conwy and Denbighshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-028	Acquisition of Rights	1882 square metres of agricultural land (north of Abergele Road, A457)	Robert William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY Unknown	NONE	Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY Robert William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY Unknown	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
02-029	Acquisition of Rights	3879 square metres of public highway and verges (Abergele road, A547) (excluding all interests of the crown)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)	

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-029 cont'd			Matthew David Jones Nursery Cottage Llanddulas Road Abergele LL22 8EU (in respect of subsoil beneath half width of public highway) The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ (in respect of subsoil beneath half width of public highway) Unknown			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Unknown 114 Whitechapel High Street London E1 7PT (in respect of underground telecommunication apparatus)
02-030	Temporary Possession	509 square metres of public highway and verges (Abergele road, A547) (excluding all interests of the crown)	Charles Martin McCoubrey Hen Wrych Farm Llanddulas Road ABERGELE Conwy LL22 8EU (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-030 cont'd			Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Matthew David Jones Nursery Cottage Llanddulas Road Abergele LL22 8EU (in respect of subsoil beneath half width of public highway) The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ (in respect of subsoil beneath half width of public highway) Unknown		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Unknown Zayo Group UK Limited The Relay Building 114 Whitechapel High Street London E1 7PT (in respect of underground telecommunication apparatus)
02-031	Acquisition of Rights	108 square metres of access splay (south of Abergele Road, A547)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	Abergele Golf Club Limited Tan Y Gopa ABERGELE Clwyd LL22 8DS (in respect of the right of access, sewerage and water pipes and drainage granted by the conveyance dated 1st June 1959)

	Land v	vhich is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) right onwy and Denbighshire	s to carry out protective works (Regulation 7(2	L)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-031 cont'd			Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY		Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as the mortgagee for Eifion William Roberts) Matthew David Jones Nursery Cottage Llanddulas Road Abergele LL22 8EU (in respect of rights of access) Robert William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (in respect of rights of access)
02-032	Acquisition of Rights	63763 square metres of agricultural land, access track and hedgerow (south of Abergele Road, A547)	Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY	NONE	Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY	Abergele Golf Club Limited Tan Y Gopa ABERGELE Clwyd LL22 8DS (in respect of the right of access, sewerage pipes, water pipes and drainage granted by the conveyance dated 1 June 1959)

	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-032 cont'd						Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as the mortgagee for Eifion William Roberts) Matthew David Jones Nursery Cottage Llanddulas Road Abergele LL22 8EU (in respect of rights of access) Robert William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (in respect of rights of access) Unknown (in respect of the right of way granted by the conveyance dated 25 May 1960)
02-033	Acquisition of Rights	69448 square metres of agricultural land and access track (south of Abergele Road, A547)	Robert William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY	NONE	Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY	Unknown

	Land v	vhich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applic: Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-033 cont'd			Unknown		Robert William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY Unknown	
02-034	Acquisition of Rights	1323 square metres of woodland (south of Abergele Road, A547) (excluding all interests of the crown)	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET	The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ	Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET The Welsh Ministers Crown Building Cathays Park CARDIFF	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of rights granted by conveyance dated 16th May 1968) Richard Broyd 22 Bedford Gardens LONDON W8 7EH (as mortgagee for Gwrych Castle Preservation Trust Limited) The Natural Resources Body For Wales
						Cambria House 29 Newport Road CARDIFF CF24 0TP (in respect of the rights of access)

				ted Development Consent Order FERENCE - PART 1		
	Land w	hich is proposed to be subject to		right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-034 cont'd						The Trustees of the National Heritage Memorial Fund Historic England Cannon Bridge House 25 Dowgate Hill LONDON EC4R 2YA (as mortgagee for Gwrych Castle Preservation Trust Limited) Unknown (in respect of the rights granted by the Conveyance dated 13 December 1946) Unknown (in respect of personal covenants contained in the Transfer dated 12 December 2006) Unknown (in respect of rights granted by conveyance dated 17th October 1946)
02-035	Acquisition of Rights	807 square metres of private access track (south of Abergele Road, A547)	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET	NONE	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET	Richard Broyd 22 Bedford Gardens LONDON W8 7EH (as mortgagee for Gwrych Castle Preservation Trust Limited) The Natural Resources Body For Wales Cambria House 29 Newport Road CARDIFF CF24 0TP (in respect of rights of access)

	Land w	which is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-035 cont'd						The Trustees of the National Heritage Memorial Fund Historic England Cannon Bridge House 25 Dowgate Hill LONDON EC4R 2YA (as mortgagee for Gwrych Castle Preservation Trust Limited) Unknown (in respect of the rights granted by the Conveyance dated 13 December 1946) Unknown (in respect of personal covenants contained in the Transfer dated 12 December 2006)
02-036	Acquisition of Rights	33953 square metres of woodland and access track (south of Abergele Road, A547) (excluding all interests of the crown)	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET	The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of rights granted by conveyance dated 16th May 1968 and leases dated 7th August 1951 and 19 September 1953) Richard Broyd 22 Bedford Gardens LONDON W8 7EH (as mortgagee for Gwrych Castle Preservation Trust Limited)

	Land wi	nich is proposed to be subject t	BOOK OF REFf to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to o ry and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-036 cont'd						The Natural Resources Body For Wales Cambria House 29 Newport Road CARDIFF CF24 OTP (in respect of rights of access) The Trustees of the National Heritage Memorial Fund Historic England Cannon Bridge House 25 Dowgate Hill LONDON EC4R 2YA (as mortgagee for Gwrych Castle Preservation Trust Limited) Unknown (in respect of the rights granted by the Conveyance dated 13 December 1946) Unknown (in respect of personal covenants contained in the Transfer dated 12 December 2006) Unknown (in respect of rights granted by conveyance dated 17 October 1946)

				ted Development Consent Order EFERENCE - PART 1		
	Land w	hich is proposed to be subject to	: (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-037	Acquisition of Rights	44797 square metres of agricultural land and hedgerow (south of Abergele Road, A547)	0	NONE	Geraint Williams Bron Y Pistyll Betws Yn Rhos ABERGELE LL22 8DA Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel
03-038	Acquisition of Rights	1224 square metres of agricultural land (north of Tan-	Helen Elmira Cato Nant Y Bella Lodge	NONE	ABERGELE Conwy LL22 8DS Geraint Williams Bron Y Pistyll	Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Glenys Pierce Williams Llys Awel
		Y-Gopa Road)	Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS		Betws Yn Rhos ABERGELE LL22 8DA	Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)

	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Counties of Conwy and Denbighshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and
03-038 cont'd			Owners or Reputed OwnersLessees or TenantsOccupiersPhilip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DSHelen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DSHelen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy	Procedures) Regulations 2009 Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)	
					Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)
	Acquisition of Rights (Hedgerow)	agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	NONE	Geraint Williams Bron Y Pistyll Betws Yn Rhos ABERGELE LL22 8DA	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)

	Land v	vhich is proposed to be subject to	BOOK OF I c: (i) powers of compulsory acquisition, (i	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to onwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-039 cont'd			Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS		Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)
03-040	Acquisition of Rights (Hedgerow)	agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT	NONE	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT Unknown	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002) Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)

	Land w	which is proposed to be subject to	BOOK OF R (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-041	Acquisition of Rights (Hedgerow)	162 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	NONE	Geraint Williams Bron Y Pistyll Betws Yn Rhos ABERGELE LL22 8DA	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)
			Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS		Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)
					Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)
03-042	Acquisition of Rights (Hedgerow)	180 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT	NONE	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (A Regulations 2009	pplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Fland			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
03-042 cont'd					Unknown	Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)			
	Acquisition of Rights (Hedgerow)	agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	NONE	Geraint Williams Bron Y Pistyll Betws Yn Rhos ABERGELE LL22 8DA Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE			
					LL22 8DS	LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)			

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 7(1)(a) of the Infrastructure Planning (Ap Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
03-044	Acquisition of Rights (Hedgerow)	2284 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	NONE	Geraint Williams Bron Y Pistyll Betws Yn Rhos ABERGELE LL22 8DA	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)			
			Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS		Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)			
					Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)			
03-045	Acquisition of Rights	•	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT	NONE	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)			

	Land w	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (i	nited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights t onwy and Denbighshire	o carry out protective works (Regulation 7(1)(ə))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-045 cont'd					Unknown	Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)
03-046	Acquisition of Rights	private road and access track (north of Tan-Y-Gopa Road)	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT	NONE	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT Unknown	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)

				ted Development Consent Order EFERENCE - PART 1			
	Land w	which is proposed to be subject to		right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
03-047	Acquisition of Rights	815 square metres of agricultural land and access track (north of Tan-Y-Gopa Road)	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	NONE	Geraint Williams Bron Y Pistyll Betws Yn Rhos ABERGELE LL22 8DA Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed	
03-048	Acquisition of Rights	656 square metres of access track (north of Tan-Y-Gopa Road)	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	NONE	Geraint Williams Bron Y Pistyll Betws Yn Rhos ABERGELE LL22 8DA	dated 4 April 1997) Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	

	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Lana i lano			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
03-048 cont'd			Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS		Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
03-049	Acquisition of Rights	agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	NONE	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)	

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a)	Category 1 of the Infrastructure Planning (Apj Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
		Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT		Norma Morris 53 Morfa Road Penrhyn Bay LLANDUDNO LL30 3PT (in respect of a grazing licence)	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)			
		Unknown (in respect of mines and minerals)		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)			
				Unknown	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown (in respect of the rights reserved contained in the Conveyance dated 17			
			Extent of acquisition or use Description of land Description of land The Executor of the Estate of the Latee Ivor Williams Lys Awel Tan-y-Gopa Road ABERGELE LL22 8DT Duknown	BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights Counties of Comuties of Comuties of Comuties of Counties of Contracting States Extent of acquisition or use Description of land Category 1 Description of land Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Approximation 2009) Regulations 2009 Regulations 2009 Owners or Reputed Owners Lessees or Tenants Image: State of the Late Information Image: State of the Late Image: State of the Late Image: State of the Late Image: State of the Late	DOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (i) right to use the land, and/or (iii) rights to use the land, and/or (iii) rights to use the land, and/or (iii) rights to use the land, and/or (iii) right to use the land, and/or (iiii) right to use the land, and/or (iii) right to			

				ited Development Consent Order EFERENCE - PART 1			
	Land w	which is proposed to be subject to	: (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
03-050	Acquisition of Rights	45 square metres of agricultural land (north of Tan- Y-Gopa Road)	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS	NONE	Geraint Williams Bron Y Pistyll Betws Yn Rhos ABERGELE LL22 8DA Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel	
03-051	Acquisition of Rights	657 square metres of access track (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE	NONE	ABERGELE Conwy LL22 8DS Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE	Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE	
			LL22 8DT		LL22 8DT	Clwyd LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)	

ent of acquisition or use			0 1 1	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
	Description of land	Qualifying persons under Regulation 7(1)(a)	Category 1) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning							
		Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009							
		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT		Norma Morris 53 Morfa Road Penrhyn Bay LLANDUDNO LL30 3PT (in respect of a grazing licence)	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)							
		Unknown (in respect of mines and minerals)		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)							
				Unknown	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown (in respect of the rights reserved							
			Tan-y-Gopa Road ABERGELE LL22 8DT Unknown	Tan-y-Gopa Road ABERGELE LL22 8DT Unknown	Tan-y-Gopa Road LLANDUDNO ABERGELE LL30 3PT LL22 8DT (in respect of a grazing licence) Unknown The Executor of the Estate of the Late (in respect of mines and minerals) Lucy Awel Tan-y-Gopa Road ABERGELE Luz2 8DT Unknown (in respect of mines and minerals) The Executor of the Estate of the Late Luz2 8DT Luz2 8DT Unknown Unknown (in respect of mines and minerals) Unknown Unknown Unknown Unknown Unknown Unknown Unknown Unknown Unknown							

	Land w	hich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-052	Temporary Possession	1867 square metres of agricultural land and access track (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	NONE	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)
			The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT		Norma Morris 53 Morfa Road Penrhyn Bay LLANDUDNO LL30 3PT (in respect of a grazing licence)	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)
			Unknown (in respect of mines and minerals)		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)
					Unknown	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)

	Land w	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-052 cont'd						Unknown (in respect of the rights reserved contained in the Conveyance dated 17 October 1946)
03-053	Acquisition of Rights	814 square metres of agricultural land and access track (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	NONE	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	Bryn Pierce Williams Plas Y Corddyn Tan Y Gopa Road ABERGELE Conwy LL22 8DT (in respect of rights of access)
			The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT		Norma Morris 53 Morfa Road Penrhyn Bay LLANDUDNO LL30 3PT (in respect of a grazing licence)	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)
			Unknown (in respect of mines and minerals)		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)

			BOOK OF RI	ted Development Consent Order EFERENCE - PART 1			
	Land v	vhich is proposed to be subject to		right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
03-053 cont'd					Unknown	Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown (in respect of the rights reserved contained in the Conveyance dated 17 October 1946)	
03-054	Acquisition of Rights	25 square metres of access track (north of Tan-Y-Gopa Road)	Bryn Pierce Williams Plas Y Corddyn Tan Y Gopa Road ABERGELE Conwy LL22 8DT (as reputed owner) Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (as reputed owner)	NONE	Bryn Pierce Williams Plas Y Corddyn Tan Y Gopa Road ABERGELE Conwy LL22 8DT (as reputed owner) Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (as reputed owner)	Unknown	

	Land w	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-054 cont'd			The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (as reputed owner) Unknown		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (as reputed owner) Unknown	
03-055	Temporary Possession	2 square metres of access track (north of Tan-Y-Gopa Road)	Bryn Pierce Williams Plas Y Corddyn Tan Y Gopa Road ABERGELE Conwy LL22 8DT (as reputed owner) Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (as reputed owner) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (as reputed owner) Unknown	NONE	Bryn Pierce Williams Plas Y Corddyn Tan Y Gopa Road ABERGELE Conwy LL22 8DT (as reputed owner) Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (as reputed owner) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (as reputed owner) Unknown	Unknown

	Land v	which is proposed to be subject t	BOOK OF Ri o: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 I)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-056	Acquisition of Rights	121 square metres of public highway (Tan-Y-Gopa Road)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of subsoil beneath half width of public highway) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of subsoil beneath half width of public highway) Unknown		Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Unknown (in respect of the rights granted by a Deed dated 15 November 1984) Unknown
03-057	Temporary Possession	54 square metres of public highway (Tan-Y-Gopa Road)	Bryn Pierce Williams Plas Y Corddyn Tan Y Gopa Road ABERGELE Conwy LL22 8DT (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)

	Land v	which is proposed to be subject	BOOK OF RI to: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-057 cont'd			Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (in respect of subsoil beneath half width of public highway) Unknown		Unknown	Unknown (in respect of the rights granted by a Deed dated 15 November 1984) Unknown
03-058	Temporary Possession	58 square metres of access track (south of Tan-Y-Gopa Road)	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT (as reputed owner) Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (as reputed owner)	NONE	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT (as reputed owner) Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (as reputed owner)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown (in respect of the rights granted by a Deed dated 15 November 1984)

	Land v	which is proposed to be subject to	BOOK OF I c: (i) powers of compulsory acquisition, (i	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights onwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (App Regulations 2009	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-058 cont'd			Unknown		Unknown	Unknown
03-059	Acquisition of Rights	3 square metres of hedgerow (south of Tan-Y-Gopa Road)	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT (as reputed owner) Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (as reputed owner) Unknown	NONE	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT (as reputed owner) Eifion William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (as reputed owner) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown
03-060	Acquisition of Rights	5211 square metres of agricultural land (north of Tan- Y-Gopa Road) (excluding all interests of the crown)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	NONE	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)

	Land whi	ich is proposed to be subject	BOOK OF REFEI to: (i) powers of compulsory acquisition, (ii) right	l Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-060 cont'd			The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT		Norma Morris 53 Morfa Road Penrhyn Bay LLANDUDNO LL30 3PT (in respect of a grazing licence)	Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)
			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)
					Unknown	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)
						Unknown (in respect of the rights reserved contained in the Conveyance dated 17 October 1946)

	Land v	which is proposed to be subject t	BOOK OF Ri to: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) right nwy and Denbighshire	s to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Ag Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-061	Acquisition of Rights	823 square metres of public highway (Tan-Y-Gopa Road)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT (in respect of subsoil beneath half width of public highway) Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of subsoil beneath half width of public highway) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of subsoil beneath half width of public highway) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of subsoil beneath half width of public highway) Unknown		Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-062	Acquisition of Rights	3 square metres of agricultural land (south of Tan-Y-Gopa Road)	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT Unknown	NONE	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown
03-063	Acquisition of Rights	45009 square metres of agricultural land (south of Tan- Y-Gopa Road) and public footpath (FP 16/14)	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of public footpath 16/14) Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights to access, use, maintain and repair water pipe contained in the Deed dated 15 November 1984) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Dyfed Roberts)

	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to onwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
20110 1 10110			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-063 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown (in respect of the rights granted by a Deed dated 15 November 1984)
03-064	Acquisition of Rights	585 square metres of hardstanding and access track (south of Tan-Y-Gopa Road)	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT	NONE	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights to access, use, maintain and repair water pipe contained in the Deed dated 15 November 1984) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Dyfed Roberts)

	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ted Development Consent Order :FERENCE - PART 1 right to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-064 cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground and overhead electricity apparatus)
04-065	Acquisition of Rights	highway (Tan-Y-Gopa Road)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT (in respect of subsoil beneath half width of public highway) Unknown	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Unknown
04-066	Acquisition of Rights	agricultural land (south of Tan- Y-Gopa Road)	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT	NONE	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Dyfed Roberts)

	Land w	hich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
04-066 cont'd						Roger Pickles Brenton House Drakewalls GUNNISLAKE PL18 9EL (in respect of the rights granted by the Conveyance dated 22 October 1946)	
04-067	Acquisition of Rights	35792 square metres of agricultural land and hedgerow (south of Tan-Y-Gopa Road)	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT	NONE	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Dyfed Roberts) Roger Pickles Brenton House Drakewalls GUNNISLAKE PL18 9EL (in respect of the rights granted by the Conveyance dated 22 October 1946)	
04-068	Temporary Possession	281 square metres of agricultural land (south of Tan- Y-Gopa Road)	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT	NONE	Dyfed Roberts Tyddyn Uchaf Farm Tan Y Gopa Road ABERGELE Conwy LL22 8DT	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Dyfed Roberts)	

	Land w	hich is proposed to be subject to	BOOK OF RI : (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-068 cont'd						Roger Pickles Brenton House Drakewalls GUNNISLAKE PL18 9EL (in respect of the rights granted by the Conveyance dated 22 October 1946)
04-069	Temporary Possession		Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE The Executor of the Estate of the Late Islwyn Williams Caeau Bedw Y Nentydd Rhyd-Y-Foel ABERGELE Clwyd LL22 8EE Unknown (in respect of mines and minerals)	NONE	Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE	NONE
04-070	Acquisition of Rights	33124 square metres of agricultural land, hedgerow and watercourse (south of Tan- Y-Gopa Road)	Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE	NONE	Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE	NONE

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-070 cont'd			The Executor of the Estate of the Late Islwyn Williams Caeau Bedw Y Nentydd Rhyd-Y-Foel ABERGELE Clwyd LL22 8EE Unknown (in respect of mines and minerals)			
04-071	Temporary Possession	3690 square metres of access track and watercourse (south of Tan-Y-Gopa Road)	Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE The Executor of the Estate of the Late Islwyn Williams Caeau Bedw Y Nentydd Rhyd-Y-Foel ABERGELE Clwyd LL22 8EE Unknown (in respect of mines and minerals)	NONE	Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE	NONE

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land		Counties of Conwy and Denbighshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
04-072	Acquisition of Rights	agricultural land (south of Tan- Y-Gopa Road)	Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE The Executor of the Estate of the Late Islwyn Williams Caeau Bedw Y Nentydd Rhyd-Y-Foel ABERGELE Clwyd LL22 8EE Unknown (in respect of mines and minerals)	NONE	Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE	NONE	
04-073	Acquisition of Rights		Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	Unknown	

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-073 cont'd			Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE (in respect of subsoil beneath half width of public highway) The Executor of the Estate of the Late Islwyn Williams Caeau Bedw Y Nentydd Rhyd-Y-Foel ABERGELE Clwyd LL22 8EE (in respect of subsoil beneath half width of public highway) Unknown		Unknown	
04-074	Acquisition of Rights		Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE	NONE	Gwynedd LL32 8DU	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)

	Land wh	ich is proposed to be subject	Mona Offshore Wind Limited BOOK OF REFER to: (i) powers of compulsory acquisition, (ii) righ Counties of Conwy	RENCE - PART 1 ht to use the land, and/or (iii) rights	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a)	Category 1 of the Infrastructure Planning (App Regulations 2009	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-074 cont'd			The Executor of the Estate of the Late Islwyn Williams Caeau Bedw Y Nentydd Rhyd-Y-Foel ABERGELE Clwyd LL22 8EE Unknown (in respect of mines and minerals)		Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE	Category 2) Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and

	Land v	which is proposed to be subject t	BOOK OF R	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1	(a))
Number on	Extent of acquisition or use	Description of land	Counties of Co Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Regulations 2009 Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-074 cont'd						Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of rights stated in a deed dated the 20th January 1955 and a deed dated the 1st August 1969)
04-075	Acquisition of Rights	33 square metres of public highway (east of Pant Idda)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE (in respect of subsoil beneath half width of public highway) The Executor of the Estate of the Late Islwyn Williams Caeau Bedw Y Nentydd Rhyd-Y-Foel ABERGELE Clwyd LL22 8EE (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Unknown

	Land w	/hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
04-075 cont'd			Unknown				
04-076	Temporary Possession	Idda)	Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE The Executor of the Estate of the Late Islwyn Williams Caeau Bedw Y Nentydd Rhyd-Y-Foel ABERGELE Clwyd LL22 8EE Unknown (in respect of mines and minerals)	NONE		Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Michael Taylor Leach Nant Ganol Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EF (in respect of right of access to relay and maintain pipes and supply of waterways contained in a Transfer dated 7 June 1996) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)	

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-076 cont'd						Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of rights stated in a deed dated the 20th January 1955 and a deed dated the 1st August 1969)
04-077	Acquisition of Rights	ldda)	Hefin Lewis Williams Caeau Bedw Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EE The Executor of the Estate of the Late Islwyn Williams Caeau Bedw Y Nentydd Rhyd-Y-Foel ABERGELE Clwyd LL22 8EE		ABERGELE Conwy LL22 8EE	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Michael Taylor Leach Nant Ganol Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EF (in respect of right of access to relay and maintain pipes and supply of waterways contained in a Transfer dated 7 June 1996)

	Land v	vhich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (mited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) right: Conwy and Denbighshire	s to carry out protective works (Regulation 7(1	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 7(1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-077 cont'd			Unknown (in respect of mines and minerals)			SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of rights stated in a deed dated the 20th January 1955 and a deed dated the 1st August 1969)
04-078	Acquisition of Rights	24399 square metres of agricultural land (west of the A548)	Edward Lloyd Griffiths Ty Mawr Ucha Farm Betws Yn Rhos ABERGELE Conwy LL22 8AA	NONE	Edward Lloyd Griffiths Ty Mawr Ucha Farm Betws Yn Rhos ABERGELE Conwy LL22 8AA	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Edward Lloyd Griffiths and Edward Wyn Griffiths)

	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
04-078 cont'd			Griffith Wyn Griffiths Ty Mawr Ucha Farm Betws Yn Rhos ABERGELE Conwy LL22 8AA Unknown (in respect of mines and minerals)		Griffith Wyn Griffiths Ty Mawr Ucha Farm Betws Yn Rhos ABERGELE Conwy LL22 8AA	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)
04-079	Acquisition of Rights	485 square metres of agricultural land (west of the A548)	Edward Lloyd Griffiths Ty Mawr Ucha Farm Betws Yn Rhos ABERGELE Conwy LL22 8AA Griffith Wyn Griffiths Ty Mawr Ucha Farm Betws Yn Rhos ABERGELE Conwy LL22 8AA Unknown (in respect of mines and minerals)	NONE	Edward Lloyd Griffiths Ty Mawr Ucha Farm Betws Yn Rhos ABERGELE Conwy LL22 8AA Griffith Wyn Griffiths Ty Mawr Ucha Farm Betws Yn Rhos ABERGELE Conwy LL22 8AA	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Edward Lloyd Griffiths and Edward Wyn Griffiths)

	Land v	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (mited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) right conwy and Denbighshire	s to carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
05-080	Acquisition of Rights	10390 square metres of agricultural land (west of the A548)	Gwenllian Humphreys Owen Nant Fawr Y Nentydd Rhyd-y-Foel ABERGELE LL22 8EF John Tudur Owen Nant Fawr Y Nentydd Rhyd-y-Foel ABERGELE LL22 8EF	NONE	Gwenllian Humphreys Owen Nant Fawr Y Nentydd Rhyd-y-Foel ABERGELE LL22 8EF John Tudur Owen Nant Fawr Y Nentydd Rhyd-y-Foel ABERGELE LL22 8EF	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)	
05-081	Acquisition of Rights	632 square metres of hedgerow and verge (west of the A548)	Gwenllian Humphreys Owen Nant Fawr Y Nentydd Rhyd-y-Foel ABERGELE LL22 8EF John Tudur Owen Nant Fawr Y Nentydd Rhyd-y-Foel ABERGELE LL22 8EF	NONE	Gwenllian Humphreys Owen Nant Fawr Y Nentydd Rhyd-y-Foel ABERGELE LL22 8EF John Tudur Owen Nant Fawr Y Nentydd Rhyd-y-Foel ABERGELE LL22 8EF	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)	

	Land w	hich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (App Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-082	Acquisition of Rights	760 square metres of public highway and verge (west of the A548)	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP (in respect of subsoil beneath half width of public highway) Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Gwenllian Humphreys Owen Nant Fawr Y Nentydd Rhyd-y-Foel ABERGELE LL22 8EF (in respect of subsoil beneath half width of public highway) Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Unknown

	landu		BOOK OF RI	ted Development Consent Order EFERENCE - PART 1	anna antartina marka (Dan Jatian 7/1	
	Land w	vnich is proposed to be subject to	o: (i) powers of compulsory acquisition, (ii) Counties of Cou	nwy and Denbighshire	carry out protective works (Regulation 7(1))(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-082 cont'd			John Tudur Owen Nant Fawr Y Nentydd Rhyd-y-Foel ABERGELE LL22 8EF (in respect of subsoil beneath half width of public highway) Karen Patricia Dakin Springhill Betws Yn Rhos ABERGELE Conwy LL22 8AB (in respect of subsoil beneath half width of public highway) Nigel Glyn Williams Springhill Betws Yn Rhos ABERGELE Conwy LL22 8AB (in respect of subsoil beneath half width of public highway) LL22 8AB (in respect of subsoil beneath half width of public highway) Unknown			
05-083	Acquisition of Rights	3496 square metres of agricultural land (west of the A548)	Karen Patricia Dakin Springhill Betws Yn Rhos ABERGELE Conwy LL22 8AB	NONE	Karen Patricia Dakin Springhill Betws Yn Rhos ABERGELE Conwy LL22 8AB	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)

	Land w	/hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order IEFERENCE - PART 1) right to use the land, and/or (iii) rights t nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-083 cont'd			Nigel Glyn Williams Springhill Betws Yn Rhos ABERGELE Conwy LL22 8AB		Nigel Glyn Williams Springhill Betws Yn Rhos ABERGELE Conwy LL22 8AB	Nationwide Building Society Nationwide House Pipers Way SWINDON Wiltshire SN38 1NW (as mortgagee for Karen Patricia Dakin and Nigel Glyn Williams)
05-084	Acquisition of Rights	37069 square metres of agricultural land and hedgerow (west of the A548) and public footpath (FP 04/48)	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	NONE	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of public footpath 04/48)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Ifor Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988)
					Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	Menna Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988)

	Land v	vhich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (mited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights Conwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Luna Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
05-084 cont'd						Unknown (in respect of the rights reserved as contained in the Conveyance dated 28 October 1988)	
05-085	Acquisition of Rights	289 square metres of agricultural land (west of the A548)	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	NONE	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	Ifor Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988)	
			Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP		Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	Menna Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988) Unknown (in respect of the rights reserved as contained in the Conveyance dated 28 October 1988)	

	Land w	hich is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
05-086	Acquisition of Rights	101 square metres of agricultural land (west of the A548)	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	NONE	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	Ifor Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988) Menna Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988) Unknown (in respect of the rights reserved as contained in the Conveyance dated 28 October 1988)	
05-087	Acquisition of Rights	13 square metres of agricultural land (west of the A548)	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	NONE	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	Unknown	

	Land w	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-087 cont'd			Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Unknown		Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Unknown	
05-088	Acquisition of Rights	15 square metres of agricultural land (west of the A548)	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Unknown	NONE	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Unknown	Unknown
05-089	Acquisition of Rights	2 square metres of agricultural land (west of the A548)	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	NONE	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	Unknown

	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-089 cont'd			Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Unknown		Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Unknown	
05-090	Acquisition of Rights	93 square metres of agricultural land (west of the A548)	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Unknown (in respect of mines and minerals)	NONE	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed dated 30 June 1992) Unknown (in respect of the rights contained in the Conveyance dated 29 September 1909) Unknown (in respect of the rights granted by a Conveyance 23 July 1986)

	Land w	which is proposed to be subject to	BOOK OF R	ited Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1)(a))	
Number on		Description of land		Counties of Conwy and Denbighshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
Land Plans	Extent of acquisition or use		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-091	Acquisition of Rights	34044 square metres of agricultural land and hedgrow (west of the A548)	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	NONE	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed dated 30 June 1992) Unknown (in respect of the rights contained in the Conveyance dated 29 September 1909)	
			(in respect of mines and minerals)			Unknown (in respect of the rights granted by a Conveyance 23 July 1986) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)	
05-092	Acquisition of Rights	1496 square metres of access track (west of the A548)	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	NONE	Ceri Llwyd Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed dated 30 June 1992)	

	Land v	vhich is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition,	mited Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) righ Conwy and Denbighshire	ts to carry out protective works (Regulation 7(1	.)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
05-092 cont'd			Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP Unknown (in respect of mines and minerals)		Hugh Watkin Lloyd Plas Newydd Farm Llanfairtalahairn ABERGELE Conwy LL22 8YP	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Unknown (in respect of the rights contained in the Conveyance dated 29 September 1909) Unknown (in respect of the rights granted by a Conveyance 23 July 1986) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas	
05-093	Acquisition of Rights	53660 square metres of agricultural land, pond and hedgerow (west of the A548)	Huw David Kirkham Grove House BETWS-Y-COED LL24 OAN	Morgan Kirkham Grove House BETWS-Y-COED LL24 OAN	Morgan Kirkham Grove House BETWS-Y-COED LL24 OAN	apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted as contained in a Deed dated 26 May 1993)	

	Land v	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
24110 1 14110			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-093 cont'd			Mafona Anne Gilligan 6530 Waterway Dr COLLEGE STATION TX 77845-3605 UNITED STATES Mafona Kirkham Grove House BETWS-Y-COED LL24 0AN		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed dated 26th May 1993)
			T he Executor of the Estate of the Late John Silas Kirkham Sirior Hir Llanfair Road ABERGELE LL22 8PB			Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)
05-094	Temporary Possession	7185 square metres of agricultural land (west of the A548)	Huw David Kirkham Grove House BETWS-Y-COED LL24 0AN	Morgan Kirkham Grove House BETWS-Y-COED LL24 0AN	Morgan Kirkham Grove House BETWS-Y-COED LL24 0AN	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted as contained in a Deed dated 26 May 1993)

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009		
Land Fland			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-094 cont'd			Mafona Anne Gilligan 6530 Waterway Dr COLLEGE STATION TX 77845-3605 UNITED STATES Mafona Kirkham Grove House BETWS-Y-COED LL24 OAN The Executor of the Estate of the Late- John Silas Kirkham Sirior Hir		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed dated 26th May 1993) Wales and West Utilities Limited Wales & West House Spooner Close
			Llanfair Road A BERGELE LL22 8PB			Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)
05-095	Temporary Possession	5875 square metres of public highway and verge (A548 and B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture)

	Land wh	nich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rigi	l Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights t y and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Fland			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-095 cont'd			Huw David Kirkham Grove House BETWS-Y-COED LL24 OAN (in respect of subsoil beneath half width of public highway) James Moloney Journeys End Mylers Town Robertstown County Kildare IRELAND (in respect of subsoil beneath half width of public highway) Mafona Anne Gilligan 6530 Waterway Dr COLLEGE STATION TX 77845-3605 UNITED STATES (in respect of subsoil beneath half width of public highway) Mafona Kirkham Grove House BETWS-Y-COED LL24 OAN (in respect of subsoil beneath half width of public highway)		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)

	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o wy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
Lanu Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
05-095 cont'd			The Executor of the Estate of the Late J ohn Silas Kirkham Sirior Hir Llanfair Road ABERGELE LL22 8PB (in respect of subsoil beneath half width of public highway) Unknown			Zayo Group UK Limited The Relay Building 114 Whitechapel High Street London E1 7PT (in respect of underground telecommunication apparatus)	
06-096	Acquisition of Rights	1224 square metres of public highway and verge (north of B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Huw David Kirkham Grove House BETWS-Y-COED LL24 0AN (in respect of subsoil beneath half width of public highway) Mafona Anne Gilligan 6530 Waterway Dr COLLEGE STATION TX 77845-3605 UNITED STATES (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
06-096 cont'd			Mafona Kirkham Grove House BETWS-Y-COED LL24 0AN (in respect of subsoil beneath half width of public highway) Stuart Duncan Neil 5 Meadow Lane ST. HELENS WA9 3RF (in respect of subsoil beneath half width of public highway) The Executor of the Estate of the Late- John Silas Kirkham Sirior Hir Llanfair Road ABERGELE LL22 8PB (in respect of subsoil beneath half width of public highway) Unknown						
06-097	Acquisition of Rights	9710 square metres of agricultural land and access track (north of B5381)	Stuart Duncan Neil 5 Meadow Lane ST. HELENS WA9 3RF Unknown (in respect of mines and minerals)	NONE	Stuart Duncan Neil 5 Meadow Lane ST. HELENS WA9 3RF	NONE			

	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land		Counties of Conwy and Denbighshire Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-098	Acquisition of Rights	2988 square metres of public highway and verge (B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Huw David Kirkham Grove House BETWS-Y-COED LL24 0AN (in respect of subsoil beneath half width of public highway) Mafona Anne Gilligan 6530 Waterway Dr COLLEGE STATION TX 77845-3605 UNITED STATES (in respect of subsoil beneath half width of public highway) Mafona Kirkham Grove House BETWS-Y-COED LL24 0AN (in respect of subsoil beneath half width of public highway) Stuart Duncan Neil 5 Meadow Lane ST. HELENS WA9 3RF (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown	

	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-098 cont'd			The Executor of the Estate of the Late John Silas Kirkham Sirior Hir Llanfair Road ABERGELE LL22 8PB (in respect of subsoil beneath half width of public highway) Unknown			
06-099	Temporary Possession	231 square metres of agricultural land (west of the A548)	Andrew William Thompson c/o Pannell Kerr Forster 52 Mount Pleasant LIVERPOOL L3 5SD (as joint Law of Property Act Receiver under the Law of Property Act 1925 and joint Agricultural Credits Act receiver under the Agricultural Credits Act 1928) Lloyd Roberts Waterloo Service Station Penrefail Crossroads Moelfre ABERGELE LL22 8PN		Lloyd Roberts Waterloo Service Station Penrefail Crossroads Moelfre ABERGELE LL22 8PN	HSBC Bank PLC 8 Canada Square LONDON E14 5HQ (as beneficiary of an agricultural charge with effect from 14 March 2000) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)

	Land v	which is proposed to be subject f	BOOK OF Ri to: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-099 cont'd			Philip James Long c/o Pannell Kerr Forster 52 Mount Pleasant LIVERPOOL L3 55D (as joint Law of Property Act Receiver under the Law of Property Act 1925 and joint Agricultural Credits Act 1928) The Executor of The Estate of the Late Georgina Roberts Waterloo Service Station Penrefail Crossroads Moelfre ABERGELE LL22 8PN Unknown			SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown
06-100	Acquisition of Rights	45284 square metres of agricultural land and access track (west of the A548)	Andrew William Thompson c/o Pannell Kerr Forster 52 Mount Pleasant LIVERPOOL L3 5SD (as joint Law of Property Act Receiver under the Law of Property Act 1925 and joint Agricultural Credits Act receiver under the Agricultural Credits Act 1928)	NONE	Lloyd Roberts Waterloo Service Station Penrefail Crossroads Moelfre ABERGELE LL22 8PN	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 OLT (in respect of underground water apparatus)

	Land v	vhich is proposed to be subject to		ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1)(a))
Number on	Extent of acquisition or use	Description of land	Counties of Cou	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Regulations 2009 Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-100 cont'd			Lloyd Roberts Waterloo Service Station Penrefail Crossroads Moelfre ABERGELE LL22 8PN Philip James Long c/o Pannell Kerr Forster 52 Mount Pleasant LIVERPOOL L3 5SD (as joint Law of Property Act Receiver under the Law of Property Act 1925 and joint Agricultural Credits Act receiver under the Agricultural Credits Act 1928) The Executor of The Estate of the Late Georgina Roberts Waterloo Service Station Penrefail Crossroads Moelfre		Unknown	HSBC Bank PLC 8 Canada Square LONDON E14 5HQ (as beneficiary of an agricultural charge with effect from 14 March 2000) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside
			ABERGELE LL22 8PN Unknown			CH43 3ET (in respect of overhead electricity apparatus) Unknown
06-101	Acquisition of Rights	2175 square metres of public highway and verge (A548)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture)

	Land wh	ich is proposed to be subject	Mona Offshore Wind Limited D BOOK OF REFERE to: (i) powers of compulsory acquisition, (ii) right Counties of Conwy a	ENCE - PART 1 to use the land, and/or (iii) rights	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a) o	Category 1 of the Infrastructure Planning (App Regulations 2009	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-101 cont'd			Elizabeth Wynne Wade 39 Brookfield Gardens West Kirby WIRRAL CH48 4EL (as reputed owner) (in respect of subsoil beneath half width of public highway) Griffith Wynne Parry 14 Grove Park LIVERPOOL L8 OTL (as reputed owner) (in respect of subsoil beneath half width of public highway) Harriet Mary Parry Pen Yr Allt Farm Tan Y Fron Road ABERGELE LL22 9BB (in respect of subsoil beneath half width of public highway) Robert Wynne Parry Pen Yr Allt Farm Tan Y Fron Road ABERGELE LL22 9BB (in respect of subsoil beneath half width of public highway) Robert Wynne Parry Pen Yr Allt Farm Tan Y Fron Road ABERGELE LL22 9BB (as reputed owner) (in respect of subsoil beneath half width of public highway) Unknown		Unknown	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown

	Land w	hich is proposed to be subject t	BOOK OF Ri o: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights t nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
Lana Hans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-102	Temporary Possession	464 square metres of public highway and verge (A548)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Elizabeth Wynne Wade 39 Brookfield Gardens West Kirby WIRRAL CH48 4EL (as reputed owner) (in respect of subsoil beneath half width of public highway) Griffith Wynne Parry 14 Grove Park LIVERPOOL L8 0TL (as reputed owner) (in respect of subsoil beneath half width of public highway) Harriet Mary Parry Pen Yr Allt Farm Tan Y Fron Road ABERGELE LL22 9BB (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown	

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-102 cont'd			Robert Wynne Parry Pen Yr Allt Farm Tan Y Fron Road ABERGELE LL22 9BB (as reputed owner) (in respect of subsoil beneath half width of public highway) Unknown			
06-103	Acquisition of Rights	40796 square metres of agricultural land and hedgerow (east of A548)	Elizabeth Wynne Wade 39 Brookfield Gardens West Kirby WIRRAL CH48 4EL (as reputed owner) Griffith Wynne Parry 14 Grove Park LIVERPOOL L8 0TL (as reputed owner) Harriet Mary Parry Pen Yr Allt Farm Tan Y Fron Road ABERGELE LL22 9BB	Iwan Roberts Tyn Y Ddol Llanfairtalhaiarn ABERGELE LL22 9RA	Tyn Y Ddol Llanfairtalhaiarn ABERGELE LL22 9RA	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 OLT (in respect of underground water apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-103 cont'd			Robert Wynne Parry Pen Yr Allt Farm Tan Y Fron Road ABERGELE LL22 9BB (as reputed owner) Unknown (in respect of mines and minerals)			
06-104	Temporary Possession	14643 square metres of agricultural land (east of A548) and electricity pylon	Elizabeth Wynne Wade 39 Brookfield Gardens West Kirby WIRRAL CH48 4EL (as reputed owner) Griffith Wynne Parry 14 Grove Park LIVERPOOL L8 0TL (as reputed owner) Harriet Mary Parry Pen Yr Allt Farm Tan Y Fron Road ABERGELE LL22 9BB	Iwan Roberts Tyn Y Ddol Llanfairtalhaiarn ABERGELE LL22 9RA	Iwan Roberts Tyn Y Ddol Llanfairtalhaiarn ABERGELE LL22 9RA	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Fians			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-104 cont'd			Robert Wynne Parry Pen Yr Allt Farm Tan Y Fron Road ABERGELE LL22 9BB (as reputed owner) Unknown (in respect of mines and minerals)			
06-105	Acquisition of Rights	2380 square metres of access track (south of the B5381)	Elizabeth Wynne Wade 39 Brookfield Gardens West Kirby WIRRAL CH48 4EL Griffith Wynne Parry 14 Grove Park LIVERPOOL L8 0TL Harriet Mary Parry Pen Yr Allt Farm Tan Y Fron Road ABERGELE	Iwan Roberts Tyn Y Ddol Llanfairtalhaiarn ABERGELE LL22 9RA		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead and underground electricity apparatus) Unknown

	Land v	vhich is proposed to be subject to	BOOK OF R	ited Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to	carry out protective works (Regulation 7(1)(a))
Number on	Extent of acquisition or use	Description of land	Counties of Co Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-105 cont'd			Robert Wynne Parry Pen Yr Allt Farm Tan Y Fron Road ABERGELE LL22 9BB Unknown			
06-106	Acquisition of Rights	67348 square metres of agricultural land and hedgerow (south of the B5381) and public footpath (FP 19/12)		NONE	Arwyn Davies Bryntwydd Moelfre ABERGELE LL22 9RD	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights of access to use, maintain and repair a pipeline granted by a Deed of Grant dated 26th April 1978 and later varied in Deed of Grant dated 10 July 1992)
			David Peter Davies Bryntwydd Moelfre ABERGELE LL22 9RD		Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of public footpath 19/12)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)
					David Peter Davies Bryntwydd Moelfre ABERGELE LL22 9RD	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus)

	Land wh	nich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ght to use the land, and/or (iii) rights to o y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-106 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect of rights to erect, use, maintain, repair and remove electric lines granted by a Deed of Grant dated 29 October 1970) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground and overhead electricity apparatus) Unknown (in respect of rights of access reserved by a Conveyance dated 24 April 1952)

	Land w	hich is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition, (mited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) right Conwy and Denbighshire	s to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 7(1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-106a	Acquisition of Rights	74279 square metres of agricultural land and hedgerow (south of the B5381) and public footpath (FP 19/12)		NONE	Arwyn Davies Bryntwydd Moelfre ABERGELE LL22 9RD Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of public footpath 19/12) David Peter Davies Bryntwydd Moelfre	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights to erect, use, maintain, repair and remove electric lines granted by a Deed of Grant dated 29 October 1970) National Grid Electricity Transmission PLC 1 - 3 Strand
					ABERGELE LL22 9RD Michael Williams Bodrochwyn Fawr Llanfair Road ABERGELE LL22 8PD (in respect of a grazing licence)	London WC2N 5EH (in respect of overhead electricity apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground and overhead electricity apparatus)

	Land v	which is proposed to be subject to	BOOK OF c (i) powers of compulsory acquisition,	imited Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) righ Conwy and Denbighshire	ts to carry out protective works (Regulation 7(1	L)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (A Regulations 2009	pplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-107	Acquisition of Rights	764 square metres of agricultural land and access track (south of the B5381) and public footpath (FP 19/12)	Arwyn Davies Bryntwydd Moelfre ABERGELE LL22 9RD David Peter Davies Bryntwydd Moelfre ABERGELE LL22 9RD	NONE	Arwyn Davies Bryntwydd Moelfre ABERGELE LL22 9RD Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of public footpath 19/12) David Peter Davies Bryntwydd Moelfre ABERGELE LL22 9RD	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus)
06-108	Acquisition of Rights	14113 square metres of agricultural land and hedgerow (south of the B5381) and public footpath (FP 19/15) (excluding all interests of the crown)	Llanfair Road	NONE	Bryn Rheinallt Williams Bodrochwyn Fawr Llanfair Road ABERGELE LL22 8PD	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)

	Land wh	nich is proposed to be subject	BOOK OF REFI to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to c vy and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-108 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)			John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights to erect, maintain and repair electric lines as granted in a Deed of Grant dated 7 January 1971)

	Land v	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii	ited Development Consent Order IEFERENCE - PART 1) right to use the land, and/or (iii) rights t onwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-108 cont'd						Unknown (in respect of right of service media stated in Conveyance dated 26 July 1985)
07-109	Acquisition of Rights	agricultural land and hedgerow (south of the B5381) (excluding all interests of the crown)		NONE	Huw Bedwyr Wyn Davies Bryn Gwynt Moelfre ABERGELE LL22 9RH	Accord Mortgages Limited Yorkshire House Yorkshire Drive BRADFORD West Yorkshire BD5 8LJ (as mortgagee for Huw Bedwyr Wyn Davies and Margaret Bethan Davies)
			Margaret Bethan Davies Bryn Gwynt Moelfre ABERGELE LL22 9RH The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)		Margaret Bethan Davies Bryn Gwynt Moelfre ABERGELE LL22 9RH	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT
						(in respect of rights granted by Transfer dated 17 January 1997)

	Land w	hich is proposed to be subject t	BOOK OF REF c: (i) powers of compulsory acquisition, (ii) r	ed Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(ə))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Lana Hano			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-109 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights in relation to electricity lines as stated in Deed dated 7 January 1971) Unknown (in respect of the right to access, use, lay, repair and maintain water, soil, gas, electricity and cables as stated in the conveyance dated 26 July 1985)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire									
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009						
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
07-110	Acquisition of Rights	1391 square metres of public highway and verges (south of the B5381) (excluding all interests of the crown)	Bryn Rheinallt Williams Bodrochwyn Fawr Llanfair Road ABERGELE LL22 8PD (in respect of subsoil beneath half width of public highway) Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Huw Bedwyr Wyn Davies Bryn Gwynt Moelfre ABERGELE LL22 9RH (in respect of subsoil beneath half width of public highway) Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985)				

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
07-110 cont'd			Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Margaret Bethan Davies Bryn Gwynt Moelfre ABERGELE LL22 9RH (in respect of subsoil beneath half width of public highway) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals) Unknown			Unknown			
07-111	Acquisition of Rights	19599 square metres of agricultural land and pond (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)			

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-111 cont'd			Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN			National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
07-112	Acquisition of Rights (Hedgerow)	734 square metres of public highway and hedgerow (south of the B5381) (excluding all interests of the crown)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-112 cont'd			Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals) Unknown			Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985) Unknown	
07-113	Acquisition of Rights (Hedgerow)	284 square metres of hedgerow (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Land w	hich is proposed to be subject to	BOOK OF c (i) powers of compulsory acquisition, (mited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) right Conwy and Denbighshire	s to carry out protective works (Regulation 7(1	L)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 7(1)(a) of the Infrastructure Planning (Ag Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
	Acquisition of Rights (Hedgerow)	66 square metres of hedgerow (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
07-115	Acquisition of Rights	3 square metres of hedgerow (south of the B5381)	Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH (as reputed owner) Unknown	NONE	Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH (as reputed owner) Unknown	Unknown

	Land w	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition,	mited Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) righ Conwy and Denbighshire	s to carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-116	Acquisition of Rights	56 square metres of private road and verges (south of the B5381) and public bridleway (BR 19/27)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of public bridleway 19/27) William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Adrian Farey Ty Newydd Moelfre ABERGELE LL22 9RL (in respect of rights of access) Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH (in respect of rights of access) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Menai Williams Bryn Goleu Moelfre ABERGELE Convy LL22 9RH (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)	

	Land w	which is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-116 cont'd						Patricia Stanley Ffynnon Dyfyr Moelfre ABERGELE Clwyd LL22 9RH (in respect of rights of access) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
07-117	Acquisition of Rights	road and verges (south of the B5381) and public bridleway (BR 19/27)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of public bridleway 19/27) William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Adrian Farey Ty Newydd Moelfre ABERGELE LL22 9RL (in respect of rights of access) Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH (in respect of rights of access) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)

	Land wh	nich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	I Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to c y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-117 cont'd						Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Patricia Stanley Ffynnon Dyfyr Moelfre ABERGELE Clwyd LL22 9RH (in respect of rights of access) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)

	Land w	which is proposed to be subject to	BOOK OI c: (i) powers of compulsory acquisition,	imited Development Consent Order FREFERENCE - PART 1 (ii) right to use the land, and/or (iii) righ Conwy and Denbighshire	ts to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (A Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-118	Acquisition of Rights (Hedgerow)	road (south of the B5381) and public bridleway (BR 19/27 and BR 19/19)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of public bridleways 19/19 and 19/27) William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Adrian Farey Ty Newydd Moelfre ABERGELE LL22 9RL (in respect of rights of access) Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH (in respect of rights of access) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH (in respect of rights of access)

				ted Development Consent Order EFERENCE - PART 1		
	Land w	hich is proposed to be subject to	: (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-118 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Patricia Stanley Ffynnon Dyfyr Moelfre ABERGELE Clwyd LL22 9RH (in respect of rights of access) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
07-119	Acquisition of Rights (Hedgerow)	1638 square metres of hedgerow (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)

	Land v	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (i	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to onwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-119 cont'd			Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN			National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
07-120	Acquisition of Rights		Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH	NONE	Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights to install, use, maintain and repair electrical lines granted by a Deed of Grant dated 10 February 2000)

	Land w	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii	ited Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) right: onwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-120 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)			
	Acquisition of Rights (Hedgerow)	924 square metres of hedgerow (south of the B5381) (excluding all interests of the crown)	Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH	NONE	Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights to install, use, maintain and repair electrical lines granted by a Deed of Grant dated 10 February 2000)
			Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)		Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)

	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-122	Acquisition of Rights (Hedgerow)	586 square metres of hedgerow (south of the B5381) (excluding all interests of the crown)	Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	NONE	Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)
07-123	Acquisition of Rights (Hedgerow)	0 / 0 (Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)

	Land wh	nich is proposed to be subject	Mona Offshore Wind Limited D BOOK OF REFERE to: (i) powers of compulsory acquisition, (ii) right Counties of Conwy a	ENCE - PART 1 t to use the land, and/or (iii) rights	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a) o	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-123 cont'd			Glyn WilliamsBryn GoleuMoelfreABERGELELL22 9RH(in respect of subsoil beneath half widthof public highway)Kinmel (H) No.1 Company LimitedSt. Johns ChambersLove StreetCHESTERCheshireCH1 1QN(in respect of subsoil beneath half widthof public highway)Kinmel (H) No.2 Company LimitedSt. Johns ChambersLove StreetCH1 1QN(in respect of subsoil beneath half widthof public highway)Kinmel (H) No.2 Company LimitedSt. Johns ChambersLove StreetCHESTERCheshireCH1 1QN(in respect of subsoil beneath half widthof public highway)Menai WilliamsBryn GoleuMoelfreABERGELEConwyLL22 9RH(in respect of subsoil beneath half widthof public highway)		Unknown	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985)

	Land w	/hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-123 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)			SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown
						Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)
07-124	Acquisition of Rights (Hedgerow)	hedgerow (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
07-124 cont'd			Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN			Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted contained in a Deed dated 27 October 1992) Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)			

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
07-124 cont'd						John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Philip James Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights granted to access water as contained in a Transfer dated 7 April 1995) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)			

	Land v	vhich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-124 cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Stephanie Elizabeth Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights granted to access water as contained in a Transfer dated 7 April 1995) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)
07-125	Acquisition of Rights	89922 square metres of agricultural land, hedgerow (south of the B5381), watercourse (drain) and public bridleway (BR 19/19)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of public bridleway 19/19)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997)

	Land wh	ich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to ry and Denbighshire	carry out protective works (Regulation 7(1)(ə))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Appli Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-125 cont'd			Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN		William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997) Nicholas Peter Cooke Pant Y Glyd Moelfre ABERGELE Conwy LL22 9RG (in respect of the rights granted by the Conveyance dated 28 March 1958) Philip James Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights granted to access water as contained in a Transfer dated 7 April 1995)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
07-125 cont'd						Robert Maldwyn Davies 1 Lon Ffawydd ABERGELE LL22 7DU (in respect of rights granted by Transfer dated 24 November 1995) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Stephanie Elizabeth Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights granted to access water as contained in a Transfer dated 7 April 1995) Susan Emma Hancox Pant Y Glyd Moelfre ABERGELE Conwy LL22 9RG (in respect of the rights granted by the Conveyance dated 28 March 1958)			

	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-125 cont'd						William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
07-126	Acquisition of Rights	1863 square metres of public highway and verge (south of the B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) (in respect of subsoil beneath half width of public highway) Unknown	NONE	Bodlondeb	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Lana Fiano			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
07-127	Acquisition of Rights	agricultural land and hedgerow (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus)			
07-128	Acquisition of Rights	750 square metres of agricultural land (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted as contained in a Deed dated 27 October 1992) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)			

				ted Development Consent Order EFERENCE - PART 1			
	Land w	hich is proposed to be subject to		right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-129	Acquisition of Rights	5773 square metres of agricultural land and hedgerow (south of the B5381)	Gwilym Williams Talgrwn Bach Moelfre ABERGELE Conwy LL22 9RW Iona Eleri Williams Talgrwn Bach Moelfre ABERGELE Conwy LL22 9RW	NONE	Gwilym Williams Talgrwn Bach Moelfre ABERGELE Conwy LL22 9RW Iona Eleri Williams Talgrwn Bach Moelfre ABERGELE Conwy LL22 9RW	Philip James Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights of access relating to service media and a water pipe in a Transfer dated 22 December 2004) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Stephanie Elizabeth Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights of access relating to service media and a water pipe in a Transfer dated 22 December 2004)	
07-130	Acquisition of Rights (Hedgerow)	214 square metres of agricultural land and hedgerow (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	William Bryn Davies Glandyfr Moelfre ABERGELE LL22 9RG	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted as contained in a Deed dated 27 October 1992)	

	Land v	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order IEFERENCE - PART 1) right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-130 cont'd			Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN			Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)
07-131	Acquisition of Rights (Hedgerow)	2518 square metres of hedgerow (south of B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	Philip Roberts Sarn Rug St George LL22 9PR	Philip Roberts Sarn Rug St George LL22 9PR	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)

	Land wi	nich is proposed to be subject t	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 tht to use the land, and/or (iii) rights to a y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-131 cont'd						Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)

	Land w	hich is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-131 cont'd						William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
07-132	Acquisition of Rights (Hedgerow)		Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Unknown	NONE	Gwynedd LL32 8DU (as highway authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown

	Land w	hich is proposed to be subject t	BOOK OF co: (i) powers of compulsory acquisition,	mited Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) right Conwy and Denbighshire	s to carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-133	Acquisition of Rights	24389 square metres of agricultuiral land and hedgerow (south of B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	Philip Roberts Sarn Rug St George LL22 9PR	Philip Roberts Sarn Rug St George LL22 9PR	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)	

	Land v	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-133 cont'd						Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
07-134	Acquisition of Rights	B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	Philip Roberts Sarn Rug St George LL22 9PR	Philip Roberts Sarn Rug St George LL22 9PR	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)

	Land w	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-134 cont'd						William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
08-135	Acquisition of Rights	442 square metres of public highway (south of B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Unknown	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Unknown

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
08-136	Acquisition of Rights	62213 square metres of agricultual land, access track, waterway and hedgerow (south of B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	Philip Roberts Sarn Rug St George LL22 9PR	Philip Roberts Sarn Rug St George LL22 9PR	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)			

	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-136 cont'd						William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
08-137	Temporary Possession	agricultural land, access track and hedgerow (south of B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	Philip Roberts Sarn Rug St George LL22 9PR	Philip Roberts Sarn Rug St George LL22 9PR	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Lund Huns			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-137 cont'd						William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
08-138	Temporary Possession	1936 square metres of public highway and verge (B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Unknown Unknown Zayo Group UK Limited The Relay Building 114 Whitechapel High Street London E1 7PT (in respect of underground telecommunication apparatus)

	Land v	vhich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-138 cont'd			Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Unknown			
08-139	Temporary Possession	736 square metres of public highway and verge (B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus)

	Land w	which is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ted Development Consent Order :FERENCE - PART 1 right to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applici Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-140	Acquisition of Rights	highway and verges (south of the B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Unknown	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown
	Acquisition of Rights (Hedgerow)	of the B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture)

	Land wh	ich is proposed to be subject	Mona Offshore Wind Limited D BOOK OF REFER to: (i) powers of compulsory acquisition, (ii) right Counties of Conwy a	ENCE - PART 1 t to use the land, and/or (iii) rights t	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a) o	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Lana Fland			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-141 cont'd			Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Unknown		Unknown	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect to underground gas apparatus)

	Land w	hich is proposed to be subject to	BOOK OF c (i) powers of compulsory acquisition, (mited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights Conwy and Denbighshire	to carry out protective works (Regulation 7(1	.)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
08-142	Acquisition of Rights (Hedgerow)	1098 square metres of hedgerow (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)	

	Land wi	nich is proposed to be subject t	BOOK OF REFE co: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to vy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Fiding			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-142 cont'd						Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)

	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-143	Acquisition of Rights (Hedgerow)	25 square metres of hedgerow (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (as reputed owner) Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (as reputed owner) Unknown	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	Unknown William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus) Unknown
08-144	Acquisition of Rights (Hedgerow)	1043 square metres of hedgerow (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
08-144 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-145	Acquisition of Rights (Hedgerow)	50 square metres of public highway and hedgerow (south of the B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway)	NONE	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	

	Land w	which is proposed to be subject to	BOOK OF RE : (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1))(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-145 cont'd			Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Unknown			
08-146	Acquisition of Rights	74666 square metres of agricultural land and hedgerow (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)

	Land w	hich is proposed to be subject to	BOOK OF REF c: (i) powers of compulsory acquisition, (ii) ri	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to vy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures				Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-146 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)

	Land v	which is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition, (i	nited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights onwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (App Regulations 2009	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-146 cont'd						William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
08-147	Acquisition of Rights	109 square metres of agricultural land (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
08-148	Acquisition of Rights	392 square metres of agricultural land (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992)

	Land wł	nich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	ed Development Consent Order iERENCE - PART 1 ight to use the land, and/or (iii) rights to o wy and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-148 cont'd			Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN			Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
08-149	Acquisition of Rights (Hedgerow)	4191 square metres of hedgerow (south of the B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER CHESTER CHESTER CH1 1QN	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	William Lloyd Evans Dinorben Farm St. George ABERGELE Conwy LL22 9RT	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)			

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Editor Flatis			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-149 cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
08-150	Acquisition of Rights	B5381)	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT	NONE	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT	Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)

	Land w	nich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to c ry and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-150 cont'd			Unknown (in respect of mines and minerals)			Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Malcolm Egerton Baron Caer Clawdd Groesffordd Marli ABERGELE Conwy LL22 9DU (in respect of right to lay down and maintain a water pipe as stated in Conveyance dated 11 April 1972) Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights) National Westminster Bank PLC 250 Bishopsgate London EC2M 4AA (as mortgagee for Steven Lloyd Jones)

	Land w	hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-150 cont'd						Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)
08-151	Acquisition of Rights	B5381)	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT Unknown (in respect of mines and minerals)	NONE	Groesffordd Marli ABERGELE LL22 9DT	Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Malcolm Egerton Baron Caer Clawdd Groesffordd Marli ABERGELE Conwy LL22 9DU (in respect of right to lay down and maintain a water pipe as stated in Conveyance dated 11 April 1972)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-151 cont'd						Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights) National Westminster Bank PLC 250 Bishopsgate London EC2M 4AA (as mortgagee for Steven Lloyd Jones) Richard William Kenrick Price Rhiwlas Estate Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)
08-152	Acquisition of Rights	19 square metres of public highway and access splay (south of the B5381)	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT (in respect of subsoil beneath half width of public highway)	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Unknown	Unknown

	Land v	vhich is proposed to be subject to	BOOK OF RI c (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-152 cont'd			Unknown			
08-153	Acquisition of Rights	480 square metres of public highway and hedgerow (south of the B5381)	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT (in respect of subsoil beneath half width of public highway) Unknown	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Unknown	Unknown
08-154	Acquisition of Rights	20598 square metres of agricultural land (south of the B5381)	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT	NONE		Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993)

	Land wh	nich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	I Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to o y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-154 cont'd			Unknown (in respect of mines and minerals)			Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)

	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii) i	ed Development Consent Order FERENCE - PART 1 'ight to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-154 cont'd						National Westminster Bank PLC 250 Bishopsgate London EC2M 4AA (as mortgagee for Steven Lloyd Jones) Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011) Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974)

	Land w	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (i	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) right onwy and Denbighshire	s to carry out protective works (Regulation 7(1	.)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-155	Acquisition of Rights (Hedgerow)	electricity pylon (south of the B5381)	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT Unknown (in respect of mines and minerals)	NONE	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993) Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)

	Land w	hich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ght to use the land, and/or (iii) rights to o y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-155 cont'd						Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012) National Westminster Bank PLC 250 Bishopsgate London EC2M 4AA (as mortgagee for Steven Lloyd Jones) Richard William Kenrick Price Rhiwlas Estate Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
09-155 cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)			

	Land w	hich is proposed to be subject to	BOOK OF F : (i) powers of compulsory acquisition, (ii	ited Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights t onwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-156	Acquisition of Rights (Hedgerow)	39 square metres of hedgerow (south of the B5381)	A Owen Cyf Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	NONE	A Owen Cyf Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Ian Alexander Thomas Johnson Ddol Farmhouse Groesffordd Marli Abergele LL29 9EB (in respect of the rights granted by a Transfer dated 19 January 2001)
09-157	Acquisition of Rights	B5381)	A Owen Cyf Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	NONE	A Owen Cyf Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Ian Alexander Thomas Johnson Ddol Farmhouse Groesffordd Marli Abergele LL29 9EB (in respect of the rights granted by a Transfer dated 19 January 2001)

	Land w	hich is proposed to be subject to	BOOK OF RE (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-157 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus)
09-158	Acquisition of Rights		Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT Unknown (in respect of mines and minerals)	NONE		Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993) Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)

	Land w	hich is proposed to be subject to	BOOK OF REF o: (i) powers of compulsory acquisition, (ii) ri	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to vy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-158 cont'd						Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012) National Westminster Bank PLC 250 Bishopsgate London EC2M 4AA (as mortgagee for Steven Lloyd Jones)

				ted Development Consent Order EFERENCE - PART 1		
	Land w	which is proposed to be subject to	: (i) powers of compulsory acquisition, (ii)		carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
24110 1 14110			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-158 cont'd						Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011) Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974)
09-159	Acquisition of Rights	15703 square metres of agricultural land (south of the B5381)	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT	NONE	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993)

	Land wi	nich is proposed to be subject t	BOOK OF REFE co: (i) powers of compulsory acquisition, (ii) rig	l Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to a y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-159 cont'd			Unknown (in respect of mines and minerals)			Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)

	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ed Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-159 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) National Westminster Bank PLC 250 Bishopsgate London EC2M 4AA (as mortgagee for Steven Lloyd Jones) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)

	Land v	vhich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
09-159 cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011) Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974)	
09-160	Temporary Possession	B5381)	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT	NONE	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993)	

	Land wi	hich is proposed to be subject t	BOOK OF REFE co: (i) powers of compulsory acquisition, (ii) rig	I Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1))(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-160 cont'd			Unknown (in respect of mines and minerals)			Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)

	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ed Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Fidins			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-160 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) National Westminster Bank PLC 250 Bishopsgate London EC2M 4AA (as mortgagee for Steven Lloyd Jones) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
09-160 cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)			

	Land w	which is proposed to be subject to	BOOK OF I c: (i) powers of compulsory acquisition, (i	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to onwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-161	Acquisition of Rights	2683 square metres of access track (south of B5381)	Unknown	NONE	Unknown	Gareth Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of rights of access) Gwen Vaughan Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of rights of access) Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 0HF (in respect of rights of access) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus)

	Land w	hich is proposed to be subject to	BOOK OF REF c: (i) powers of compulsory acquisition, (ii) ri	ed Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to vy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-161 cont'd						Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED (in respect of rights of access) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT (in respect of rights of access) Unknown Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)

	Land w	/hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-162	Temporary Possession	571 square metres of verge (south of B5381)	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT (as reputed owner) Unknown	NONE	Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT (as reputed owner) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of overhead and underground telecommunication apparatus) Unknown
09-163	Temporary Possession	811 square metres of public highway and verge (B5381, Roman Road)	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN	NONE
09-164	Temporary Possession	810 square metres of public highway and verge (B5381, Roman Road)	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway)	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Unknown

	Land v	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-164 cont'd			Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) (in respect of subsoil beneath half width of public highway) Unknown			
09-165	Temporary Possession	234 square metres of hedgerow, verge and access splay (south of B5381)	Gareth Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (as reputed owner)	NONE	Tan Dderwen Glascoed ABERGELE	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of overhead telecommunication apparatus)

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-165 cont'd			Gwen Vaughan Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (as reputed owner) Unknown		Gwen Vaughan Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (as reputed owner) Unknown	Unknown
09-166	Temporary Possession	1199 square metres of public highway and verge (B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority)	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights reserved as contained in a Conveyance dated 6 February 1981) Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights reserved as contained in a Conveyance dated 6 February 1981)

	Land v	vhich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-166 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Zayo Group UK Limited The Relay Building 114 Whitechapel High Street London E1 7PT (in respect of underground telecommunication apparatus)
09-167	Temporary Possession	and Cae Onnen Road)	Caryl Ann Pugh Celston Hall Farm Llanasa TREFFYNNON CH8 9NE (in respect of subsoil beneath half width of public highway) Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of subsoil beneath half width of public highway)	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Unknown	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (in respect of street furniture) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)

	Land wh	ich is proposed to be subject	Mona Offshore Wind Limited BOOK OF REFEF to: (i) powers of compulsory acquisition, (ii) righ Counties of Conwy	RENCE - PART 1 ht to use the land, and/or (iii) rights to (carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a)	Category 1) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-167 cont'd			Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Gareth Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of subsoil beneath half width of public highway) Gary Stephen Johnston Glascoed Lodge Glascoed ABERGELE LL22 9DE (in respect of subsoil beneath half width of public highway) Gwen Vaughan Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of subsoil beneath half width of public highway) Gwen Vaughan Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of subsoil beneath half width of public highway)			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead and underground electricity apparatus) Unknown Zayo Group UK Limited The Relay Building 114 Whitechapel High Street London E1 7PT (in respect of underground telecommunication apparatus)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a)	Category 1 of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
09-167 cont'd			Kevin Arthur Bainbridge Waterloo Glascoed ABERGELE LL22 9DE (in respect of subsoil beneath half width of public highway) Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of subsoil beneath half width of public highway) Kirsty Bainbridge Waterloo Glascoed ABERGELE LL22 9DE (in respect of subsoil beneath half width of public highway)						

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-167 cont'd			Osian Daniel Pugh Celston Hall Farm Llanasa TREFFYNNON CH8 9NE (in respect of subsoil beneath half width of public highway) Suzanne Johnston Glascoed Lodge Glascoed ABERGELE LL22 9DE (in respect of subsoil beneath half width of public highway) Unknown			
09-168	Acquisition of Rights	585 square metres of access track (off Cae Onnen)	Unknown	NONE		Gareth Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of rights of access) Gwen Vaughan Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of rights of access)

	Land w	hich is proposed to be subject t	BOOK OF REF o: (i) powers of compulsory acquisition, (ii) ri	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to c vy and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-168 cont'd						Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF (in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED (in respect of rights of access) Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT (in respect of rights of access) Unknown

	Land w	vhich is proposed to be subject to	BOOK OF I c: (i) powers of compulsory acquisition, (i	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights onwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-169	Acquisition of Rights	agricultural land (north of Cae Onnen Road)	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	NONE	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	Unknown
09-170	Acquisition of Rights	agricultural land (north of Cae Onnen Road)	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	NONE	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	Unknown

	Land w	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights onwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (App Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-171	Acquisition of Rights	3859 square metres of agricultural land, hedgerow and access track (north of Cae Onnen Road)	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF	NONE	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)
			Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown		Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	Unknown
09-172	Acquisition of Rights		Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF	NONE	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus)
			Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown		Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	Unknown

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights t nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-173	Acquisition of Rights	1121 square metres of public highway (Cae Onnen Road)	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF (in respect of subsoil beneath half width of public highway) Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED (in respect of subsoil beneath half width of public highway) Unknown	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Unknown	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown
09-173a	Acquisition of Rights	13 square metres of public highway (Cae Onnen Road)	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 0HF	NONE	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 0HF	Unknown

	Land v	which is proposed to be subject t	BOOK OF to: (i) powers of compulsory acquisition, (mited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) right Conwy and Denbighshire	s to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 7(1)(a) of the Infrastructure Planning (Ap Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-173a cont'd			Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown		Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	
09-173b	Acquisition of Rights	31 square metres of public highway (Cae Onnen Road)	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 0HF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	NONE	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown
09-174	Acquisition of Rights	4474 square metres of agricultural land and access track (south of Cae Onnen Road)	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF	NONE	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)

	Land w	/hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-174 cont'd			Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown		Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	Unknown
09-175	Acquisition of Rights	agricultural land, access track and an air shaft (south of Cae Onnen Road) and public	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (in respect of public footpath 105/5) Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 0HF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 OLT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of overhead telecommunication apparatus) Unknown

	Land w	hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-176	Acquisition of Rights	track (south of Cae Onnen Road) and public footpath (FP 105/5)	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	NONE	County Hall Wynnstay Road	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown
09-177	Acquisition of Rights	woodland (Coed Carreg- Dafydd)	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF	NONE	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF	Unknown

	Land w	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	ited Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to onwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-177 cont'd			Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown		Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	
09-178	Acquisition of Rights	· ·	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (in respect of public footpath 105/5) Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 0HF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED Unknown	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Aj Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
10-179	Acquisition of Rights	0 / 0	Gillian Ann Parry Plas Hafod Groesffordd Marli ABERGELE LL22 9DS	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (in respect of public footpath 105/5)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)			
			Hugh Morris Parry Plas Hafod Groesffordd Marli ABERGELE LL22 9DS	Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED	Gillian Ann Parry Plas Hafod Groesffordd Marli ABERGELE LL22 9DS	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights in relation to electricity lines as stated in Deed dated 22 September 2021)			
					Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead and underground electricity apparatus)			
					Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED	Unknown (in respect of rights stated in Lease dated 10 February 1932)			

	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)	i(a))
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-180	Acquisition of Rights	11123 square metres of agricultural land, hedgerow and a pond (Maes Cefn)	John Woolliams Maes Cefn LLANELWY LL17 OHE Unknown (in respect of mines and minerals) Winifred Stephane Woolliams Maes Cefn LLANELWY LL17 OHE	NONE	Maes Cefn	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)
10-181	Acquisition of Rights	178 square metres of agricultural land (Maes Cefn)	John Woolliams Maes Cefn LLANELWY LL17 OHE Unknown (in respect of mines and minerals)	NONE	John Woolliams Maes Cefn LLANELWY LL17 OHE Winifred Stephane Woolliams Maes Cefn LLANELWY LL17 OHE	NONE

	Land w	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-181 cont'd			Winifred Stephane Woolliams Maes Cefn LLANELWY LL17 0HE			
10-182	Acquisition of Rights	696 square metres of public highway (east of Plas Hafod)	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Gillian Ann Parry Plas Hafod Groesffordd Marli ABERGELE LL22 9DS (in respect of subsoil beneath half width of public highway) Hugh Morris Parry Plas Hafod Groesffordd Marli ABERGELE LL22 9DS (in respect of subsoil beneath half width of public highway)	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown

	Land w	nich is proposed to be subject	Mona Offshore Wind Limited BOOK OF REFER to: (i) powers of compulsory acquisition, (ii) righ Counties of Conwy	RENCE - PART 1 ht to use the land, and/or (iii) rights to a	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a)	Category 1 of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-182 cont'd			Huw Lloyd EvansTan Y GraigCefnST. ASAPHLL17 OHF(in respect of subsoil beneath half widthof public highway)John WoolliamsMaesCefnLLANELWYLL17 OHE(in respect of subsoil beneath half widthof public highway)Robert John Lloyd EvansBryn HenGroesffordd MarliABERGELEClwydLL22 9ED(in respect of subsoil beneath half widthof public highway)UnknownWinifred Stephane WoolliamsMaesCefnLLANELWYLL17 OHE(in respect of subsoil beneath half widthof public highway)			

	Land v	vhich is proposed to be subject to	BOOK OF I c (i) powers of compulsory acquisition, (i	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to onwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-183		. ,	John Woolliams Maes Cefn LLANELWY LL17 OHE Unknown (in respect of mines and minerals)	NONE	John Woolliams Maes Cefn LLANELWY LL17 OHE Winifred Stephane Woolliams Maes Cefn LLANELWY LL17 OHE	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)
			Winifred Stephane Woolliams Maes Cefn LLANELWY LL17 OHE			
10-184	Acquisition of Rights	5284 square metres of agricultural land and hedgerow (east of Plas Hafod)	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE	NONE	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Robert John Lloyd Evans and Huw Lloyd Evans) Edward Durand Hotham Ashcroft Furlong West End Chadlington
			Clwyd LL22 9ED		Clwyd LL22 9ED	CHIPPING NORTON OX7 3NJ (in respect of rights granted by Conveyance dated 2 September 1966)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
10-184 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Robin Hugh Barlow Carreg Dafydd Groesffordd Marli ABERGELE Conwy LL22 9ED (in respect of rights reserved by Conveyance dated 29 March 1978) Susan Elizabeth Hotham Plas Newydd Groesffordd Marli ABERGELE Conwy LL22 9DS (in respect of rights granted by Conveyance dated 2 September 1966) Unknown (in respect of rights reserved by Conveyance dated 29 March 1978)			
10-185	Acquisition of Rights	49204 square metres of agricultural land and hedgerow (south of Glascoed Road)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)			

	Land wi	nich is proposed to be subject	BOOK OF REFf to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to ca ry and Denbighshire	arry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-185 cont'd						Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	Land v	which is proposed to be subject	BOOK OF F to: (i) powers of compulsory acquisition, (ii	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) righ onwy and Denbighshire	ts to carry out protective works (Regulation 7(1	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (A Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-186	Acquisition of Rights	945 square metres of agricultural land (south of Glascoed Road)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus)

	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1))(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-186 cont'd						Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)
10-187	Acquisition of Rights	track (south of Glascoed Road, B5831)	Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF (as reputed owner) loan Arfon Pierce Ysgubor Eos Groesffordd Marli ABERGELE Clwyd LL22 9DR (as reputed owner) Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED (as reputed owner)	NONE	Unknown	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Unknown

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire							
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
10-187 cont'd			Sadie Louise Pierce Ysgubor Eos Groesffordd Marli ABERGELE Clwyd LL22 9DR (as reputed owner) The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (as reputed owner) Unknown					
10-188	Acquisition of Rights	29211 square metres of agricultural land and hedgerow (south of Glascoed Road, B5831) and and public footpath (FP 105/6)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014)		

	Land wi	nich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to ca y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-188 cont'd						Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	landw	which is proposed to be subject to		ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to (carry out protective works (Regulation 7/1)(a))	
				nwy and Denbighshire	carry out protective works (Regulation 7/1	(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
10-188 cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-189	Acquisition of Rights	highway (south of the National Grid Bodelwyddan substation)	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect of subsoil beneath half width of public highway) Unknown	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown	

	Land w	/hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights nwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-190	Freehold Acquisition	3730 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Lund Fluits			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-190 cont'd						Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)			

	Land w	which is proposed to be subject to	BOOK OF R (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights t nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-190 cont'd						Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)
11-191	Acquisition of Rights	(south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)

	Land wh	ich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 tht to use the land, and/or (iii) rights to ca y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-191 cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LLG0 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	Land v	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-191 cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)
11-192	Acquisition of Rights	12141 square metres of agricultural land and hedgerow (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014)

	Land wh	ich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order :RENCE - PART 1 ;ht to use the land, and/or (iii) rights to ca y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-192 cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	Land v	vhich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights t nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (App Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-192 cont'd						Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)
11-193	Acquisition of Rights (Hedgerow)	1144 square metres of hedgerow (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)

	Land w	hich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-193 cont'd						Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-193 cont'd						Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)			
11-194	Acquisition of Rights (Hedgerow)	National Grid Bodelwyddan substation)	Derek James Greaves Isfryn Cefn ST. ASAPH LL17 OHG Olive Greaves Isfryn Cefn ST. ASAPH LL17 OHG Unknown	NONE	Derek James Greaves Isfryn Cefn ST. ASAPH LL17 OHG Olive Greaves Isfryn Cefn ST. ASAPH LL17 OHG Unknown	Unknown			
11-195	Acquisition of Rights (Hedgerow)	road and hedgerow (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)			

	Land wi	nich is proposed to be subject	BOOK OF REFI to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to o ry and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-195 cont'd						Craig Dakin Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Katherine Edwards Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access)

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-195 cont'd						Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground and overhead electricity apparatus) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
	Acquisition of Rights (Hedgerow)	13 square metres of private road and hedgerow (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	NONE	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)

	Land wi	nich is proposed to be subject i	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	ed Development Consent Order ERENCE - PART 1 Ight to use the land, and/or (iii) rights to o wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-195a cont'd						Craig Dakin Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Katherine Edwards Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access)

	Land v	vhich is proposed to be subject to	BOOK OF Ri c (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-195a cont'd						Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground and overhead electricity apparatus) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-196	Acquisition of Rights	238 square metres of private road and verges (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights granted by deed dated 4th March 2017)

	Land w	hich is proposed to be subject to	BOOK OF REF c: (i) powers of compulsory acquisition, (ii) ri	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(-	Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Lana Hano			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-196 cont'd						Craig Dakin Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Katherine Edwards Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Land v	vhich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights nwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (App Regulations 2009	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-196 cont'd						Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-197	Freehold Acquisition	pond, access track and	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)

	Land w	hich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-197 cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus and underground telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)

	Land wh	nich is proposed to be subject t	BOOK OF REFE co: (i) powers of compulsory acquisition, (ii) rig	l Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to o y and Denbighshire	carry out protective works (Regulation 7(1))(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-197 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	

	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-197 cont'd						Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)
11-197a	Temporary Possession	National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	NONE	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)

	Land w	hich is proposed to be subject to	BOOK OF REFE c: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
Land Fiding			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-197a cont'd						Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus and underground telecommunication apparatus) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	

	Land v	vhich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-197a cont'd						Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)
11-197b	Temporary Possession	road (south of the National	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	NONE	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus and underground telecommunication apparatus) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)

	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order :FERENCE - PART 1 right to use the land, and/or (iii) rights wy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-197c	Freehold Acquisition	6354 square metres of agricultural land, hedgerow, pond, access track and electricity pylon (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	NONE	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)

	Land v	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-197c cont'd						Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)
11-198	Acquisition of Rights	track (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014)

	Land w	hich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-198 cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus and underground telecommunication apparatus)	

	Land wh	ich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to car ry and Denbighshire	rry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Application Regulations 2009	ons: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-198 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Fiding			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-199	Freehold Acquisition	5408 square metres of agricultural land and treeline (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)			

	Land wi	nich is proposed to be subject	BOOK OF REFf to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to can ry and Denbighshire	rry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Application Regulations 2009	ons: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-199 cont'd						Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	Land v	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	nited Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) right onwy and Denbighshire	s to carry out protective works (Regulation 7(1	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Ap Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-200	Acquisition of Rights	pylon (south of the National	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (in respect of underground electricity apparatus) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-200 cont'd						Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)			

	Land wh	ich is proposed to be subject	BOOK OF REFI to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to car ry and Denbighshire	rry out protective works (Regulation 7(1	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applicatio Regulations 2009	ons: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-200 cont'd						Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)

	Land w	which is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition, (nited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) right: onwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Hans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-201	Acquisition of Rights	agricultural land and hedgerow (west of the National Grid Bodelwyddan substation)	Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	NONE	Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed of Grant dated 12 November 1992) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Mona Elizabeth Davies and Robert Bryn Davies) Island Green Power UK Limited Unit 25.7 Coda Studios 189 Munster Road London SW6 6AW (in respect of an option agreement) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of underground electricity apparatus and underground

	Land wł	nich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to o y and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
20110 1 10110			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-201 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of restrictive covenants and the rights granted by the Deed dated 06 November 2009) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by a Deed of Grant dated 12 November 1992) The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect of the rights as stated in the Deed dated 06 November 2009) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)	

	Land w	hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-201 cont'd						Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)
11-202	Acquisition of Rights	(west of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014)

	Land wł	nich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to a y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-202 cont'd						Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)

	Land wh	ich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order :RENCE - PART 1 ;ht to use the land, and/or (iii) rights to ca y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-202 cont'd						Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)

	Land w	which is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (mited Development Consent Order REFERENCE - PART 1 (ii) right to use the land, and/or (iii) right Conwy and Denbighshire	s to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Aj Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Hans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-203	Acquisition of Rights	9570 square metres of agricultural land (west of the National Grid Bodelwyddan substation)	Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	NONE	Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed of Grant dated 12 November 1992) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Mona Elizabeth Davies and Robert Bryn Davies) Island Green Power UK Limited Unit 25.7 Coda Studios 189 Munster Road London SW6 6AW (in respect of an option agreement) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of restrictive covenants and the rights granted by the Deed dated 06

	Land wi	hich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	I Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-203 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of underground electricity apparatus and underground telecommunication apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by a Deed of Grant dated 12 November 1992) The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect of the rights as stated in the Deed dated 06 November 2009) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	ited Development Consent Order IEFERENCE - PART 1) right to use the land, and/or (iii) rights t nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-203 cont'd						Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of a Deed dated 10 June 2011) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)
11-204	Acquisition of Rights	3021 square metres of agricultural land (west of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993)

	Land w	hich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-204 cont'd						Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (in respect of underground electricity apparatus) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	

	Land wi	nich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to o y and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-204 cont'd						Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of proposed underground telecommunication apparatus)	

	Land wh	nich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to o y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-204 cont'd						Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)

	Land w	/hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-204 cont'd						Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)
11-205	Acquisition of Rights	12 square metres of agricultural land (east of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)

	Land wl	nich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	l Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to o y and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-205 cont'd						Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	

	Land w	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights nwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-205 cont'd						Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-206	Acquisition of Rights	252 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (in respect of underground electricity apparatus) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014)

	Land w	hich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	I Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-206 cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)	

	Land v	vhich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-206 cont'd						Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-207	Acquisition of Rights	1184 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014)

	Land wh	nich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to ca y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-207 cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect of underground telecommunication apparatus)

	Land wh	nich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to car y and Denbighshire	rry out protective works (Regulation 7(1	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applicatio Regulations 2009	ons: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-207 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)

	Land w	which is proposed to be subject to	BOOK OF : (i) powers of compulsory acquisition, (nited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) right onwy and Denbighshire	ts to carry out protective works (Regulation 7(2	L)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-208	Freehold Acquisition	Bodelwyddan substation)	Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY	NONE	Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights to have and use cables contained in a Deed dated 4 May 2017) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights contained in a Deed dated 10 June 1977)
11-209	Freehold Acquisition	Bodelwyddan substation)	Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY	NONE	Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights to have and use cables contained in a Deed dated 4 May 2017) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights contained in a Deed dated 10 June 1977)

	Land w	hich is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	ited Development Consent Order IEFERENCE - PART 1) right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-210	Freehold Acquisition	1498 square metres of woodland (south of Glascoed Road)	Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY	NONE	Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights to have and use cables contained in a Deed dated 4 May 2017) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of underground telecommunication apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights contained in a Deed dated 10 June 1977) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)

	Land w	/hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights nwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-211	Freehold Acquisition	2013 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)	

	Land wh	ich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	I Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to ca y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-211 cont'd						Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus and underground telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to onwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-211 cont'd						Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)	
11-212	Freehold Acquisition	2656 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	NONE	Emma Smith Tan Y Bryn Cefn ST. ASAPH LL17 OHG	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary in respect of an Option Agreement dated 3rd April 2014)	

	Land	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (ii	ited Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights onwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (App Regulations 2009	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Earla Fians			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-212 cont'd						Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-213	Freehold Acquisition	15147 square metres of agricultural land and pond (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)

	Land wh	ich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to ca y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
201011010			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-213 cont'd						Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)		o carry out protective works (Regulation 7(1)(a))
Number on	Extent of acquisition or use	Description of land	Counties of Co Qualifying persons under Regulation 7(cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-214	Acquisition of Rights	67861 square metres of agricultural land, pond, hedgrow, access track and electricity pylon (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Craig Dakin Tan Y Bryn Uchaf Cefn LLANELWY LL17 0HG (in respect of rights of access) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Katherine Edwards Tan Y Bryn Uchaf Cefn LLANELWY LL17 0HG (in respect of rights of access)

	Land v	vhich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	ited Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights onwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (App Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-214 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-215	Temporary Possession	38142 square metres of agricultural land and pond (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-215 cont'd						Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)			

	Land v	vhich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1))(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-215 cont'd						Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 OEY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-216	Freehold Acquisition	(south of the National Grid	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Bodysgaw Isaf Llannefydd DENBIGH Denbighshire	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)
						Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)

	Land v	vhich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-216 cont'd						Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-217	Freehold Acquisition	822 square metres of agricultural land (south of the National Grid Bodelwyddan substation	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014)

	Land v	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-217 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-218	Temporary Possession	16202 square metres of agricultural land, hedgerow and pond (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	NONE	Dewi Roberts Fron Hill Cefn Meiriadog LLANELWY LL17 OHP	Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of right of drainage and access for maintenance)

	Land v	vhich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights onwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (App Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-218 cont'd						Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of right of drainage and access for maintenance) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-218a	Temporary Possession	14130 square metres of agricultural land, hedgerow and pond (south of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	NONE	Dewi Roberts Fron Hill Cefn Meiriadog LLANELWY LL17 OHP	Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of right of drainage and access for maintenance)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-218a cont'd						Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of right of drainage and access for maintenance) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)			
11-219	Freehold Acquisition	5663 square metres of agricultural land and hedgerow (south of the Gwynt y Mor Offshore Wind Farm Substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU	Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)			

	Land wi	hich is proposed to be subject t	BOOK OF to: (i) powers of compulsory acquisition, (mited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to Conwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-219 cont'd				Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR		Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (in respect of underground electricity apparatus) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	

	Land wl	hich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	I Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-219 cont'd						Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of underground telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	ited Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to onwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-219 cont'd						Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)
11-220	Freehold Acquisition	3075 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH (as reputed owner)	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH (as reputed owner)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)

	Land wh	ich is proposed to be subject	BOOK OF REFI to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to can ry and Denbighshire	rry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Application Regulations 2009	ons: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-220 cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)

	Land v	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-220 cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)
11-221	Acquisition of Rights	2146 square metres of private road and verges (east of the Gwynt y Mor Offshore Wind Farm Substation)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH (as reputed owner)	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH (as reputed owner)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus and underground abandoned water apparatus)

	Land wh	nich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to vy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-221 cont'd						ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the lease dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-221 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus)
11-221a	Temporary Possession	127 Square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH		Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014)

	Land w	hich is proposed to be subject t	BOOK OF RE co: (i) powers of compulsory acquisition, (ii) r	ed Development Consent Order FERENCE - PART 1 'ight to use the land, and/or (iii) rights to a wy and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-221a cont'd			Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH (as reputed owner)			ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the lease dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect to the right to access as stated in the deed dated 6 November 2009)

	Land w	/hich is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land		ations: Prescribed Forms and Procedures)	7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-221a cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)
11-222	Acquisition of Rights	174 square metres of private road (south of the Gwynt y Mor Offshore Wind Farm Substation)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH (as reputed owner)	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH (as reputed owner)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the lease dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009)

	Land w	hich is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))	
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-222 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)	
11-223	Acquisition of Rights	979 square metres of private road and verges (south of the Gwynt y Mor Offshore Wind Farm Substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU		Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS (in respect of rights of access) Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	

	Land w	hich is proposed to be subject to	BOOK OF REF o: (i) powers of compulsory acquisition, (ii) r	ed Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-223 cont'd						Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus)

	Land wi	nich is proposed to be subject	BOOK OF REFE co: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to ca ry and Denbighshire	arry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-223 cont'd						Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002) Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights granted by lease dated 16 September 2011) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)

	Land w	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) right nwy and Denbighshire	s to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Ap Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-224	Acquisition of Rights	29 square metres of private road and verges (south of the Gwynt y Mor Offshore Wind Farm Substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET	Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS (in respect of rights of access) Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-224 cont'd						Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)			

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Editor Flatis			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-224 cont'd						Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights granted by lease dated 16 September 2011) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)			
11-225	Acquisition of Rights		The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)			

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-225 cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus)			

	Land wł	nich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to ca ry and Denbighshire	arry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applicat Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-225 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002) Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)

	Land w	hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Lana i lano			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-225 cont'd						Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)
11-226		National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)

	Land w	hich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-226 cont'd						Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of underground telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-226 cont'd						Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)
11-227	Acquisition of Rights	360 square metres of agricultural land (east of the National Grid Bodelwyddan substation)	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)

	Land wh	ich is proposed to be subject	BOOK OF to: (i) powers of compulsory acquisition, (mited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to Conwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 7(1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-227 cont'd				National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993)
				Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR	Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)
						Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)
						Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)

	Land wi	nich is proposed to be subject t	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	l Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to o y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-227 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)

	Land wh	ich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to ca y and Denbighshire	arry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applicati Regulations 2009	ions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
2010 1 10115			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-227 cont'd						Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)

	Land v	vhich is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (nited Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) righ conwy and Denbighshire	s to carry out protective works (Regulation 7(1	.)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (A Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-228	Temporary Possession	1337 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the lease dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus)

	Land w	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-228 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus)
11-229	Temporary Possession	6518 square metres of agricultural land (south of Glascoed road, B5381)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH		Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-229 cont'd			Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH			Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus and underground abandoned water apparatus) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009)			

	Land w	hich is proposed to be subject to	BOOK OF REF c: (i) powers of compulsory acquisition, (ii) r	ed Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
24114 1 14115			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-229 cont'd						Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground and overhead electricity apparatus)

	Land w	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to onwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-229 cont'd						Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)
11-229a	Temporary Possession	313 square metres of agricultural land (south of Glascoed road, B5381)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus and underground abandoned water apparatus) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)

	Land wi	nich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	l Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to c y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-229a cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)

	Land w	hich is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-229a cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground and overhead electricity apparatus)
11-230	Temporary Possession	Glascoed road, B5381) and public bridleway (BR 208/32)	Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (as reputed owner)	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (in respect of public bridleway 208/32)	Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights to install and use electric lines as stated in the Deed of Grant dated 6 November 2009)
			The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (as reputed owner) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (as reputed owner)		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (as reputed owner) The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (as reputed owner)	Unknown

	Land w	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights t nwy and Denbighshire	o carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-230 cont'd			Unknown		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (as reputed owner) Unknown	
11-230a	Temporary Possession	38 square metres of agricultural land (south of Glascoed road, B5381) and public bridleway (BR 208/32)	Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (as reputed owner)	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (in respect of public bridleway 208/32)	Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights to install and use electric lines as stated in the Deed of Grant dated 6 November 2009)
			The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (as reputed owner) Timothy Michael Bell Cefn Estate Office		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR The Executor of the Estate of the Late David Watkin Williams-Warn BT	Unknown
			Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (as reputed owner)		David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	

	Land v	which is proposed to be subject to	BOOK OF R : (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights nwy and Denbighshire	to carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Lana i lano			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-230a cont'd			Unknown		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY Unknown	
11-231	Acquisition of Rights	33 square metres of private road (off Glascoed road, B5381) and public bridleway (BR 208/32)	Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (as reputed owner) The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (as reputed owner) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY	NONE	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (in respect of public bridleway 208/32) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (as reputed owner) The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights contained within a Lease dated 30 October 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights in relation to the installation and use of electric lines within a Deed of Grant dated 6 November 2009) Unknown

	Land w	which is proposed to be subject to	BOOK OF R	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iji) rights to	carry out protective works (Regulation 7(1)(a))
Number on	Extent of acquisition or use	Description of land	Counties of Co	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Regulations 2009 Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-231 cont'd			Unknown		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (as reputed owner) Unknown	
11-232	Acquisition of Rights	1147 square metres of private road (south of Glascoed road, B5381)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated in the lease dated 30 October 2014) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground abandoned water apparatus) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)

	Land wł	nich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	I Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to c y and Denbighshire	arry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applicat Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Fland			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-232 cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 6 January 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground proposed telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus)

	Land w	hich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to many and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Lund Huns			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-232 cont'd						The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)
11-233	Temporary Possession	1439 square metres of agricultural land and hardstanding (south of Glascoed road, B5381)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH		Glascoed Road ST. ASAPH Clwyd LL17 OLH	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)

	Land wl	hich is proposed to be subject t	BOOK OF REFE co: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to o y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-233 cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 6 January 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)

	Land v	vhich is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-233 cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)
11-233a	Temporary Possession	783 square metres of agricultural land and hardstanding (south of Glascoed road, B5381)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)

	Land w	hich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-233a cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 6 January 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)

	Land v	which is proposed to be subject t	BOOK OF F co: (i) powers of compulsory acquisition, (ii	nited Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to onwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-233a cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)
11-233b	Temporary Possession	2013 square metres of agricultural land (south of Glascoed road, B5381)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)

	Land wi	hich is proposed to be subject t	BOOK OF REFE o: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights to y and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-233b cont'd						Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 6 January 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)

	Land v	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	ited Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to nwy and Denbighshire	carry out protective works (Regulation 7(1)(ə))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-233b cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)
11-234	Temporary Possession	523 square metres of agricultural land and access splay (south of Glascoed road, B5381)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of right of way stated in Transfer dated 10 June 2016) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus)

	Land w	which is proposed to be subject t	BOOK OF R co: (i) powers of compulsory acquisition, (ii	ited Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to onwy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-234 cont'd						SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)
11-234a	Temporary Possession	79 square metres of agricultural land (south of Glascoed road, B5381)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of right of way stated in Transfer dated 10 June 2016) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire					
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 7(1)(a) of the Infrastructure Planning (Ag Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-234b	Temporary Possession	338 square metres of agricultural land (south of Glascoed road, B5381)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of right of way stated in Transfer dated 10 June 2016) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)
11-235	Acquisition of Rights	194 square metres of private road (south of Glascoed road, B5381)	Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	NONE	Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 OLH	ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of right of way stated in Transfer dated 10 June 2016)

	Land w	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ted Development Consent Order FFERENCE - PART 1 right to use the land, and/or (iii) rights to wy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Flans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-235 cont'd			Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH			Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)
11-236	Temporary Possession	highway and verges (Glascoed road, B5381)	Anthony Stephenson AJ Bell Investcentre 4 Exchange Quay Salford Quays Manchester M5 3EE (in respect of subsoil beneath half width of public highway) Betty May Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH (in respect of subsoil beneath half width of public highway)	NONE		Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (in respect of street furniture) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (in respect of underground electricity apparatus)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) Counties of Conwy and Denbighshire					
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-236 cont'd			Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (as highway authority) Eryl James Gomer Davies Derwen Deg Glascoed Road ST. ASAPH LL17 0LH (in respect of subsoil beneath half width of public highway) Glyn Jones Plas yr Esgob Glascoed Road ST. ASAPH Clwyd LL17 0LH (in respect of subsoil beneath half width of public highway) Linda Louise Griffiths Cottage Villa Glascoed Road LLANELWY LL17 0LH (in respect of subsoil beneath half width of public highway)			Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus and underground abandoned water apparatus) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground and overhead electricity apparatus)

	Land w	hich is proposed to be subject	BOOK OF REFE to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to vy and Denbighshire	carry out protective works (Regulation 7(1)(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-236 cont'd			Neil Watkins AJ Bell Investcentre 4 Exchange Quay Salford Quays Manchester M5 3EE (in respect of subsoil beneath half width of public highway) Nesta Wyn Davies Derwen Deg Glascoed Road ST. ASAPH LL17 0LH (in respect of subsoil beneath half width of public highway) Sippdeal Trustees Limited 4 Exchange Quay SALFORD Greater Manchester M5 3EE (in respect of subsoil beneath half width of public highway) Unknown			Unknown Zayo Group UK Limited The Relay Building 114 Whitechapel High Street London E1 7PT (in respect of underground telecommunication apparatus)

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire					
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
01-001	177297 square metres of foreshore land seaward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of the right to lay and maintain sea outfall pipes granted by the Deed dated 20 February 1955 and in respect of the rights to place and maintain groynes as stated in the Deed dated 1 May 1964 and in respect of the rights contained in the Deed dated 4 April 1973) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of the Deed dated 01 April 1997)			
01-003	10945 square metres of foreshore landward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) Unknown			

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
	167 square metres of foreshore landward of the Mean High Water (north of the A55, Abergele)	Unknown	
01-005	2946 square metres of public highway (Sea Road, Abergele)	Unknown	
01-006	395 square metres of private road (north of the A55, Abergele)	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) Unknown	
	101 square metres of private road (north of the A55, Abergele) (excluding all interests of the crown)	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access)	

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-007 cont'd		Julie Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012)
		Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access)
		Michael Williams 162 Ffordd Y Foryd Bae Cinmel Y RHYL LL18 5LS (in respect of rights of access to maintain services as contained in a Lease dated 2 June 2006)
		Nicholas Stephen Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-007 cont'd		Pjan Limited Unit 32 Llys Edmund Prys St. Asaph Business Park ST. ASAPH Flintshire LL17 OJA (in respect of rights of access as contained in a Lease dated 3 November 2010)	
		Sian Williams 5 Terfyn Court Terfyn Bodelwyddan RHYL Denbigshire LL18 5SW (in respect of rights contained in a Deed dated 21 September 2016)	
		The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964)	
		Unknown (in respect of rights contained in a Conveyance dated 29 July 1935)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-008	325 square metres of private road (north of the A55, Abergele)	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access)	
		Julie Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access)	
		Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access)	
		Michael Williams 162 Ffordd Y Foryd Bae Cinmel Y RHYL LL18 5LS (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-008 cont'd		Nicholas Stephen Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access)	
		Pjan Limited Unit 32 Llys Edmund Prys St. Asaph Business Park ST. ASAPH Flintshire LL17 0JA (in respect of rights of access)	
		Unknown	
	1931 square metres of private road and verges (north of the A55, Abergele) (excluding all interests of the crown)	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Julie Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-009 cont'd		Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access)	
		Michael Williams 162 Ffordd Y Foryd Bae Cinmel Y RHYL LL18 5LS (in respect of right of access to maintain services as contained in a Lease dated 2 June 2006)	
		Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights contained in a Conveyance dated 29 July 1935)	
		Nicholas Stephen Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012)	

		nore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-009 cont'd		Pjan Limited Unit 32 Llys Edmund Prys St. Asaph Business Park ST. ASAPH Flintshire LL17 OJA (in respect of right of access to maintain services as contained in a Lease dated 3 November 2010)
		Sian Williams 5 Terfyn Court Terfyn Bodelwyddan RHYL Denbigshire LL18 5SW (in respect of rights contained in a Deed dated 21 September 2016)
		The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964)
01-010	386 square metres of private road and verges (north of the A55, Abergele) (excluding all interests of the crown)	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access)

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-010 cont'd		Julie Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012)
		Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access)
		Michael Williams 162 Ffordd Y Foryd Bae Cinmel Y RHYL LL18 5LS (in respect of right access to maintain services as contained in a Lease dated 2 June 2006)
		Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights contained in a Conveyance dated 29 July 1935)
		Nicholas Stephen Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012)

		ore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-010 cont'd		Pjan Limited Unit 32 Llys Edmund Prys St. Asaph Business Park ST. ASAPH Flintshire LL17 OJA (in respect of rights of access to maintain services as contained in a Lease dated 3 November 2010)
		Sian Williams 5 Terfyn Court Terfyn Bodelwyddan RHYL Denbigshire LL18 5SW (in respect of rights contained in a Deed dated 21 September 2016)
		The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964)
01-011	11530 square metres of private road and verges (north of the A55, Abergele) (excluding all interests of the crown)	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-011 cont'd		Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access)	
		Unknown	
02-013	167 square metres of private road, hardstanding and grassland (Beach House Road)	Unknown	
02-013a	58 square metres of private road, hardstanding and grassland (Beach House Road) (excluding all interests of the crown)	Unknown	
02-014	116 square metres of private road and railway (Beach House Road)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (in respect of rights of access) Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-014 cont'd		Unknown	
02-015	1 square metres of grassland (north of Abergele Road, A547)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (in respect of rights of access) Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (in respect of rights of access) Unknown	
02-016	15 square metres of hardstanding and grassland (north of the A55, Abergele) (excluding all interests of the crown)	Unknown	
02-017	8927 square metres of grassland and track (north of Abergele Road, A547) (excluding all interests of the crown)	Unknown	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-018	5529 square metres of hardstanding, grassland and track (north of Abergele Road, A547) (excluding all interests of the crown)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of agreement to grant an easement stated in Deed dated 8 July 1994)	
		Unknown	
02-019	92 square metres of grassland (north of Abergele Road, A547)	Unknown	
02-020	1140 square metres of grassland (north of Abergele Road, A547)	Unknown	
02-021	13245 square metres of foreshore land landward of the Mean High Water and cycle way (north of the A55, Abergele) (excluding all interests of the crown)	Unknown	
02-021a	3287 square metres of foreshore land landward of the Mean High Water and cycle way (north of the A55, Abergele) (excluding all interests of the crown)	Unknown	
02-022	4484 square metres of grassland (lying to the north of the A55, Abergele)	Unknown	
02-023	665 square metres of grassland (north of the A55, Abergele)	Unknown	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-024	2517 square metres of grassland (north of the A55, Abergele)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of agreement to grant an easement stated in Deed dated 8 July 1994)	
		Unknown	
02-025	19299 square metres of grassland and railway (north of the A55, Abergele)	Unknown	
02-028	1882 square metres of agricultural land (north of Abergele Road, A457)	Unknown	
02-029	3879 square metres of public highway and verges (Abergele road, A547) (excluding all interests of the crown)	Unknown	
02-030	509 square metres of public highway and verges (Abergele road, A547) (excluding all interests of the crown)	Unknown	
02-031	108 square metres of access splay (south of Abergele Road, A547)	Abergele Golf Club Limited Tan Y Gopa ABERGELE Clwyd LL22 8DS (in respect of the right of access, sewerage and water pipes and drainage granted by the conveyance dated 1st June 1959)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-031 cont'd		Matthew David Jones Nursery Cottage Llanddulas Road Abergele LL22 8EU (in respect of rights of access)	
		Robert William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (in respect of rights of access)	
	63763 square metres of agricultural land, access track and hedgerow (south of Abergele Road, A547)	Abergele Golf Club Limited Tan Y Gopa ABERGELE Clwyd LL22 8DS (in respect of the right of access, sewerage pipes, water pipes and drainage granted by the conveyance dated 1 June 1959) Matthew David Jones Nursery Cottage Llanddulas Road Abergele LL22 8EU (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-032 cont'd		Robert William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (in respect of rights of access) Unknown (in respect of the right of way granted by the conveyance dated 25 May 1960)	
02-033	69448 square metres of agricultural land and access track (south of Abergele Road, A547)	Unknown	
02-034	1323 square metres of woodland (south of Abergele Road, A547) (excluding all interests of the crown)	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of rights granted by conveyance dated 16th May 1968) The Natural Resources Body For Wales Cambria House 29 Newport Road CARDIFF CF24 0TP (in respect of the rights of access)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-034 cont'd		Unknown (in respect of the rights granted by the Conveyance dated 13 December 1946)	
		Unknown (in respect of personal covenants contained in the Transfer dated 12 December 2006)	
		Unknown (in respect of rights granted by conveyance dated 17th October 1946)	
02-035	807 square metres of private access track (south of Abergele Road, A547)	The Natural Resources Body For Wales Cambria House 29 Newport Road CARDIFF CF24 OTP (in respect of rights of access)	
		Unknown (in respect of the rights granted by the Conveyance dated 13 December 1946)	
		Unknown (in respect of personal covenants contained in the Transfer dated 12 December 2006)	
02-036	33953 square metres of woodland and access track (south of Abergele Road, A547) (excluding all interests of the crown)	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of rights granted by conveyance dated 16th May 1968 and leases dated 7th August 1951 and 19 September 1953)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-036 cont'd		The Natural Resources Body For Wales Cambria House 29 Newport Road CARDIFF CF24 0TP (in respect of rights of access)	
		Unknown (in respect of the rights granted by the Conveyance dated 13 December 1946)	
		Unknown (in respect of personal covenants contained in the Transfer dated 12 December 2006)	
		Unknown (in respect of rights granted by conveyance dated 17 October 1946)	
03-037	44797 square metres of agricultural land and hedgerow (south of Abergele Road, A547)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
		Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-037 cont'd		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
03-038	1224 square metres of agricultural land (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
		Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)	
		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-039	165 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
		Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)	
		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
03-040	186 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)	
		Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-041	162 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
		Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)	
		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
03-042	180 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)	
		Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)	

Mona Offshore Wind Limited Development Consent Order		
BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on		
Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
03-043	17 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)
		Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)
		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)
03-044	2284 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
03-044 cont'd		Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)
		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)
03-045	19034 square metres of agricultural land, hedgerow and access track (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)
		Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)
	1494 square metres of hardstanding (Henblas Farm), private road and access track (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-046 cont'd		Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)	
03-047	815 square metres of agricultural land and access track (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel	
03-048	656 square metres of access track (north of Tan-Y-Gopa Road)	Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT	
		(in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-048 cont'd		Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)	
		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
03-049	24349 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002) Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-049 cont'd		Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)	
		Unknown (in respect of the rights reserved contained in the Conveyance dated 17 October 1946)	
03-050	45 square metres of agricultural land (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
		Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)	
		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-051	657 square metres of access track (north of Tan-Y-Gopa Road)	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)	
		Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)	
		Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)	
		Unknown (in respect of the rights reserved contained in the Conveyance dated 17 October 1946)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-052	1867 square metres of agricultural land and access track (north of Tan-Y-Gopa Road)	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)	
		Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)	
		Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)	
		Unknown (in respect of the rights reserved contained in the Conveyance dated 17 October 1946)	

		hore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
03-053	814 square metres of agricultural land and access track (north of Tan-Y-Gopa Road)	Bryn Pierce Williams Plas Y Corddyn Tan Y Gopa Road ABERGELE Conwy LL22 8DT (in respect of rights of access)
		Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)
		Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)
		Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-053 cont'd		Unknown (in respect of the rights reserved contained in the Conveyance dated 17 October 1946)	
03-054	25 square metres of access track (north of Tan-Y-Gopa Road)	Unknown	
03-055	2 square metres of access track (north of Tan-Y-Gopa Road)	Unknown	
03-056	121 square metres of public highway (Tan-Y-Gopa Road)	Unknown (in respect of the rights granted by a Deed dated 15 November 1984) Unknown	
03-057	54 square metres of public highway (Tan-Y-Gopa Road)	Unknown (in respect of the rights granted by a Deed dated 15 November 1984) Unknown	
03-058	58 square metres of access track (south of Tan-Y-Gopa Road)	Unknown (in respect of the rights granted by a Deed dated 15 November 1984)	

		ore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 jounties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
03-058 cont'd		Unknown
03-059	3 square metres of hedgerow (south of Tan-Y-Gopa Road)	Unknown
03-060	5211 square metres of agricultural land (north of Tan-Y-Gopa Road) (excluding all interests of the crown)	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT (In respect of the rights granted by the Transfer dated 30 September 2002) Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004) Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-060 cont'd		Unknown (in respect of the rights reserved contained in the Conveyance dated 17 October 1946)	
03-061	823 square metres of public highway (Tan-Y-Gopa Road)	Unknown	
03-062	3 square metres of agricultural land (south of Tan-Y-Gopa Road)	Unknown	
	45009 square metres of agricultural land (south of Tan-Y-Gopa Road) and public footpath (FP 16/14)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights to access, use, maintain and repair water pipe contained in the Deed dated 15 November 1984) Unknown (in respect of the rights granted by a Deed dated 15 November 1984)	
03-064	585 square metres of hardstanding and access track (south of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights to access, use, maintain and repair water pipe contained in the Deed dated 15 November 1984)	
04-065	481 square metres of public highway (Tan-Y-Gopa Road)	Unknown	

	Mona Offsl	hore Wind Limited Development Consent Order	
	BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-066	130 square metres of agricultural land (south of Tan-Y-Gopa Road)	Roger Pickles Brenton House Drakewalls GUNNISLAKE PL18 9EL (in respect of the rights granted by the Conveyance dated 22 October 1946)	
04-067	35792 square metres of agricultural land and hedgerow (south of Tan-Y-Gopa Road)	Roger Pickles Brenton House Drakewalls GUNNISLAKE PL18 9EL (in respect of the rights granted by the Conveyance dated 22 October 1946)	
04-068	281 square metres of agricultural land (south of Tan-Y-Gopa Road)	Roger Pickles Brenton House Drakewalls GUNNISLAKE PL18 9EL (in respect of the rights granted by the Conveyance dated 22 October 1946)	
04-073	685 square metres of public highway (east of Pant Idda)	Unknown	
	44216 square metres of agricultural land and hedgerow (east of Pant Idda) and public footpath (FP 04/43 and FP 04/44)	Michael Taylor Leach Nant Ganol Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EF (in respect of right of access to relay and maintain pipes and supply of waterways contained in a Transfer dated 7 June 1996)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-074 cont'd		Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of rights stated in a deed dated the 20th January 1955 and a deed dated the 1st August 1969)	
04-075	33 square metres of public highway (east of Pant Idda)	Unknown	
04-076	2638 square metres of agricultural land (east of Pant Idda)	Michael Taylor Leach Nant Ganol Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EF (in respect of right of access to relay and maintain pipes and supply of waterways contained in a Transfer dated 7 June 1996) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of rights stated in a deed dated the 20th January 1955 and a deed dated the 1st August 1969)	

	Description of Land Potential claims under \$10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
Number on Land Plans			
04-077	1667 square metres of agricultural land, hedgerow and access track (east of Pant Idda)	Michael Taylor Leach Nant Ganol Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8FF (in respect of right of access to relay and maintain pipes and supply of waterways contained in a Transfer dated 7 June 1996) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of rights stated in a deed dated the 20th January 1955 and a deed dated the 1st August 1969)	
05-082	760 square metres of public highway and verge (west of the A548)	Unknown	
05-084	37069 square metres of agricultural land and hedgerow (west of the A548) and public footpath (FP 04/48)	Ifor Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-084 cont'd		Menna Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988) Unknown (in respect of the rights reserved as contained in the Conveyance dated 28 October 1988)	
05-085	289 square metres of agricultural land (west of the A548)	Ifor Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988)	
		Menna Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988)	
		Unknown (in respect of the rights reserved as contained in the Conveyance dated 28 October 1988)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-086	101 square metres of agricultural land (west of the A548)	Ifor Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988)	
		Menna Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988)	
		Unknown (in respect of the rights reserved as contained in the Conveyance dated 28 October 1988)	
05-087	13 square metres of agricultural land (west of the A548)	Unknown	
05-088	15 square metres of agricultural land (west of the A548)	Unknown	
05-089	2 square metres of agricultural land (west of the A548)	Unknown	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-090	93 square metres of agricultural land (west of the A548)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed dated 30 June 1992)	
		Unknown (in respect of the rights contained in the Conveyance dated 29 September 1909)	
		Unknown (in respect of the rights granted by a Conveyance 23 July 1986)	
05-091	34044 square metres of agricultural land and hedgrow (west of the A548)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed dated 30 June 1992) Unknown (in respect of the rights contained in the Conveyance dated 29 September 1909) Unknown (in respect of the rights granted by a Conveyance 23 July 1986)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-092	1496 square metres of access track (west of the A548)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed dated 30 June 1992)	
		Unknown (in respect of the rights contained in the Conveyance dated 29 September 1909)	
		Unknown (in respect of the rights granted by a Conveyance 23 July 1986)	
05-093	53660 square metres of agricultural land, pond and hedgerow (west of the A548)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted as contained in a Deed dated 26 May 1993)	
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed dated 26th May 1993)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire			
Number on Land Plans	Number on Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2			
05-094	7185 square metres of agricultural land (west of the A548)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted as contained in a Deed dated 26 May 1993)		
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed dated 26th May 1993)		
05-095	5875 square metres of public highway and verge (A548 and B5381)	Unknown		
06-096	1224 square metres of public highway and verge (north of B5381)	Unknown		
06-098	2988 square metres of public highway and verge (B5381)	Unknown		
06-099	231 square metres of agricultural land (west of the A548)	Unknown		
06-100	45284 square metres of agricultural land and access track (west of the A548)	Unknown		

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
06-101	2175 square metres of public highway and verge (A548)	Unknown	
06-102	464 square metres of public highway and verge (A548)	Unknown	
06-105	2380 square metres of access track (south of the B5381)	Unknown	
06-106	67348 square metres of agricultural land and hedgerow (south of the B5381) and public footpath (FP 19/12)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights of access to use, maintain and repair a pipeline granted by a Deed of Grant dated 26th April 1978 and later varied in Deed of Grant dated 10 July 1992) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights to erect, use, maintain, repair and remove electric lines granted by a Deed of Grant dated 29 October 1970) Unknown (in respect of rights of access reserved by a Conveyance dated 24 April 1952)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
06-106a	74279 square metres of agricultural land and hedgerow (south of the B5381) and public footpath (FP 19/12)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights to erect, use, maintain, repair and remove electric lines granted by a Deed of Grant dated 29 October 1970)	
06-108	14113 square metres of agricultural land and hedgerow (south of the B5381) and public footpath (FP 19/15) (excluding all interests of the crown)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
06-108 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights to erect, maintain and repair electric lines as granted in a Deed of Grant dated 7 January 1971) Unknown (in respect of right of service media stated in Conveyance dated 26 July 1985)	
07-109	407 square metres of agricultural land and hedgerow (south of the B5381) (excluding all interests of the crown)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-109 cont'd		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights in relation to electricity lines as stated in Deed dated 7 January 1971) Unknown (in respect of the right to access, use, lay, repair and maintain water, soil, gas, electricity and cables as stated in the conveyance dated 26 July 1985)	
07-110	1391 square metres of public highway and verges (south of the B5381) (excluding all interests of the crown)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-110 cont'd		John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)	
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985)	
		Unknown	
07-111	19599 square metres of agricultural land and pond (south of the B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
	734 square metres of public highway and hedgerow (south of the B5381) (excluding all interests of the crown)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)	
		John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)	
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985)	
		Unknown	
07-113	284 square metres of hedgerow (south of the B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-113 cont'd		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
07-114	66 square metres of hedgerow (south of the B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
07-115	3 square metres of hedgerow (south of the B5381)	Unknown	
	56 square metres of private road and verges (south of the B5381) and public bridleway (BR 19/27)	Adrian Farey Ty Newydd Moelfre ABERGELE LL22 9RL (in respect of rights of access)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
07-116 cont'd		Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH (in respect of rights of access)
		Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
		Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH (in respect of rights of access)
		Patricia Stanley Ffynnon Dyfyr Moelfre ABERGELE Clwyd LL22 9RH (in respect of rights of access)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-116 cont'd		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
07-117	1043 square metres of private road and verges (south of the B5381) and public bridleway (BR 19/27)	Adrian Farey Ty Newydd Moelfre ABERGELE LL22 9RL (in respect of rights of access) Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH (in respect of rights of access) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

		nore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
07-117 cont'd		Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH (in respect of rights of access)
		Patricia Stanley Ffynnon Dyfyr Moelfre ABERGELE Clwyd LL22 9RH (in respect of rights of access)
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
07-118	3198 square metres of private road (south of the B5381) and public bridleway (BR 19/27 and BR 19/19)	Adrian Farey Ty Newydd Moelfre ABERGELE LL22 9RL (in respect of rights of access)

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
07-118 cont'd		Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH (in respect of rights of access)
		Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
		Menai Williams Bryn Goleu Moelfre ABERGELE Conwy LL22 9RH (in respect of rights of access)
		Patricia Stanley Ffynnon Dyfyr Moelfre ABERGELE Clwyd LL22 9RH (in respect of rights of access)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-118 cont'd		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
07-119	1638 square metres of hedgerow (south of the B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP	
		(in respect of the rights granted by the Conveyance dated 12 January 1979)	
	13556 square metres of agricultural land (south of the B5381) (excluding all interests of the crown)	 SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights to install, use, maintain and repair electrical lines granted by a Deed of Grant dated 10 February 2000) 	

		ore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
07-121	924 square metres of hedgerow (south of the B5381) (excluding all interests of the crown)	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights to install, use, maintain and repair electrical lines granted by a Deed of Grant dated 10 February 2000)
	586 square metres of hedgerow (south of the B5381) (excluding all interests of the crown)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)
07-123	3859 square metres of public highway and hedgerow (south of the B5381) (excluding all interests of the crown)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2	
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
07-123 cont'd		John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985) Unknown
07-124	2532 square metres of hedgerow (south of the B5381)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997)

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
07-124 cont'd		Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted contained in a Deed dated 27 October 1992)
		Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)
		John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997)
		Philip James Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights granted to access water as contained in a Transfer dated 7 April 1995)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-124 cont'd		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)	
		Stephanie Elizabeth Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights granted to access water as contained in a Transfer dated 7 April 1995)	
07-125	89922 square metres of agricultural land, hedgerow (south of the B5381), watercourse (drain) and public bridleway (BR 19/19)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-125 cont'd		John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997)	
		Nicholas Peter Cooke Pant Y Glyd Moelfre ABERGELE Conwy LL22 9RG (in respect of the rights granted by the Conveyance dated 28 March 1958)	
		Philip James Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights granted to access water as contained in a Transfer dated 7 April 1995)	
		Robert Maldwyn Davies 1 Lon Ffawydd ABERGELE LL22 7DU (in respect of rights granted by Transfer dated 24 November 1995)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 2	
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-125 cont'd		Stephanie Elizabeth Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights granted to access water as contained in a Transfer dated 7 April 1995) Susan Emma Hancox	
		Pant Y Glyd Moelfre ABERGELE Conwy LL22 9RG (in respect of the rights granted by the Conveyance dated 28 March 1958)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
07-126	1863 square metres of public highway and verge (south of the B5381)	Unknown	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-128	750 square metres of agricultural land (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted as contained in a Deed dated 27 October 1992)	
	5773 square metres of agricultural land and hedgerow (south of the B5381)	Philip James Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights of access relating to service media and a water pipe in a Transfer dated 22 December 2004) Stephanie Elizabeth Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights of access relating to service media and a water pipe in a Transfer dated 22 December 2004)	
07-130	214 square metres of agricultural land and hedgerow (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted as contained in a Deed dated 27 October 1992)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-131	2518 square metres of hedgerow (south of B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992)	
		Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
07-132	40 square metres of public highway and hedgerow (B5381)	Unknown	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-133	24389 square metres of agricultuiral land and hedgerow (south of B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992)	
		Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-134	73 square metres of agricultural land (south of the B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-135	442 square metres of public highway (south of B5381)	Unknown	
	62213 square metres of agricultual land, access track, waterway and hedgerow (south of B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-136 cont'd		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-137	23867 square metres of agricultural land, access track and hedgerow (south of B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-138	1936 square metres of public highway and verge (B5381)	Unknown	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-140	1158 square metres of public highway and verges (south of the B5381)	Unknown	
08-141	881 square metres of public highway and hedgerow (south of the B5381)	Unknown	
08-142	1098 square metres of hedgerow (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992)	
		Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	Counties of Conwy and Denbighshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-142 cont'd		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-143	25 square metres of hedgerow (south of the B5381)	Unknown	
08-144	1043 square metres of hedgerow (south of the B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-145	50 square metres of public highway and hedgerow (south of the B5381)	Unknown	

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2			
Counties of Conwy and Denbighshire			
Number on Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
Land Plans 74666 square metres of agricultural land and hedgerow (south of the B5381) 08-146 74666 square metres of agricultural land and hedgerow (south of the B5381) 1 1	Centrica PLC Milistream Maidenhead Road WiNDSOR Berkshire SL4 SGD (in respect of the rights granted by the Deed dated 27 October 1992) Janice Marie Davies Pistyli Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Shell U.K. Limited Shell Centre York Road LONDON SEL 7NA (in respect of the rights granted by the Deed dated 27 October 1992) William Goronwy Davies Pistyli Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Deed dated 27 October 1992) William Goronwy Davies Pistyli Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)		

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-147	109 square metres of agricultural land (south of the B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-148	392 square metres of agricultural land (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 2	
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-148 cont'd		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-149	4191 square metres of hedgerow (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-149 cont'd		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-150	8599 square metres of agricultural land (south of the B5381)	Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)	
		Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-150 cont'd		Malcolm Egerton Baron Caer Clawdd Groesffordd Marli ABERGELE Conwy LL22 9DU (in respect of right to lay down and maintain a water pipe as stated in Conveyance dated 11 April 1972)	
		Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights)	
		Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)	
08-151 310	0 square metres of agricultural land (south of the B5381)	Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
08-151 cont'd		Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)
		Malcolm Egerton Baron Caer Clawdd Groesffordd Marli ABERGELE Conwy LL22 9DU (in respect of right to lay down and maintain a water pipe as stated in Conveyance dated 11 April 1972)
		Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights)
		Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-152	19 square metres of public highway and access splay (south of the B5381)	Unknown	
08-153	480 square metres of public highway and hedgerow (south of the B5381)	Unknown	
08-154	20598 square metres of agricultural land (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993) Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) (in respect of rights granted by the conveyance dated 4 October 1989) (in respect of rights granted by the conveyance dated 4 October 1989) (in respect of rights granted by the conveyance dated 4 October 1989)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
08-154 cont'd		Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)
		Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-154 cont'd		Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974)	
09-155	2642 square metres of hedgerow, access track and electricity pylon (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993) Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Kinmel No.4 Company Limited St. Johns Chambers Love Street CH5TER Cheshire CH11QN (in respect of rights granted by the conveyance dated 4 October 1989) Kinmel No.4 Company Limited	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
09-155 cont'd		Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)
		Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-155 cont'd		Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974)	
09-156	39 square metres of hedgerow (south of the B5381)	lan Alexander Thomas Johnson Ddol Farmhouse Groesffordd Marli Abergele LL29 9EB (in respect of the rights granted by a Transfer dated 19 January 2001)	
09-157	22005 square metres of agricultural land (south of the B5381)	Ian Alexander Thomas Johnson Ddol Farmhouse Groesffordd Marli Abergele LL29 9EB (in respect of the rights granted by a Transfer dated 19 January 2001)	
09-158	356 square metres of hedgerow (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-158 cont'd		Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)	
		Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)	
		Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-158 cont'd		Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011) Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974)	
09-159	15703 square metres of agricultural land (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-159 cont'd		Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)	
		Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)	
		Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-159 cont'd		Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)	
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011)	
		Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974)	
09-160	51355 square metres of agricultural land (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-160 cont'd		Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)	
		Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)	
		Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-160 cont'd		Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)	
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011)	
		Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974)	
09-161	2683 square metres of access track (south of B5381)	Gareth Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of rights of access)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
09-161 cont'd		Gwen Vaughan Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of rights of access)
		Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF (in respect of rights of access)
		Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED (in respect of rights of access)
		Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT (in respect of rights of access)
		Unknown

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-162	571 square metres of verge (south of B5381)	Unknown	
09-164	810 square metres of public highway and verge (B5381, Roman Road)	Unknown	
09-165	234 square metres of hedgerow, verge and access splay (south of B5381)	Unknown	
09-166	1199 square metres of public highway and verge (B5381)	Kinmel (H) No.1 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights reserved as contained in a Conveyance dated 6 February 1981)	
		Kinmel (H) No.2 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights reserved as contained in a Conveyance dated 6 February 1981)	
	5273 square metres of public highway and verge (B5381, Glascoed Road, Roman Road and Cae Onnen Road)	Unknown	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-168	585 square metres of access track (off Cae Onnen)	Gareth Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of rights of access)	
		Gwen Vaughan Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of rights of access)	
		Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 OHF (in respect of rights of access)	
		Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd LL22 9ED (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-168 cont'd		Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT (in respect of rights of access)	
		Unknown	
09-169	34835 square metres of agricultural land (north of Cae Onnen Road)	Unknown	
09-170	30 square metres of agricultural land (north of Cae Onnen Road)	Unknown	
09-171	3859 square metres of agricultural land, hedgerow and access track (north of Cae Onnen Road)	Unknown	
09-172	1284 square metres of agricultural land (north of Cae Onnen Road)	Unknown	
09-173	1121 square metres of public highway (Cae Onnen Road)	Unknown	
09-173a	13 square metres of public highway (Cae Onnen Road)	Unknown	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-173b	31 square metres of public highway (Cae Onnen Road)	Unknown	
09-174	4474 square metres of agricultural land and access track (south of Cae Onnen Road)	Unknown	
	38922 square metres of agricultural land, access track and an air shaft (south of Cae Onnen Road) and public footpath (FP 105/5)	Unknown	
09-176	1109 square metres of access track (south of Cae Onnen Road) and public footpath (FP 105/5)	Unknown	
09-177	734 square metres of woodland (Coed Carreg-Dafydd)	Unknown	
	24186 square metres of agricultural land (south of Cae Onnen Road) and public footpath (FP 105/5)	Unknown	
	64148 square metres of agricultural land, hedgerow and an electricity pylon (south of Plas Hafod) and public footpath (FP 105/5)	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights in relation to electricity lines as stated in Deed dated 22 September 2021)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-179 cont'd		Unknown (in respect of rights stated in Lease dated 10 February 1932)	
10-182	696 square metres of public highway (east of Plas Hafod)	Unknown	
10-184	5284 square metres of agricultural land and hedgerow (east of Plas Hafod)	Edward Durand Hotham Ashcroft Furlong West End Chadlington CHIPPING NORTON OX7 3NJ (in respect of rights granted by Conveyance dated 2 September 1966) Robin Hugh Barlow Carreg Dafydd Groesffordd Marli ABERGELE Conwy L122 9ED (in respect of rights reserved by Conveyance dated 29 March 1978) Susan Elizabeth Hotham Plas Newydd Groesffordd Marli ABERGELE Conwy L122 9DS (in respect of rights granted by Conveyance dated 2 September 1966)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-184 cont'd		Unknown (in respect of rights reserved by Conveyance dated 29 March 1978)	
10-185	49204 square metres of agricultural land and hedgerow (south of Glascoed Road)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
10-186	945 square metres of agricultural land (south of Glascoed Road)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-186 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn Ecfn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
10-187	1525 square metres of access track (south of Glascoed Road, B5831)	Unknown	
10-188	29211 square metres of agricultural land and hedgerow (south of Glascoed Road, B5831) and and public footpath (FP 105/6)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-188 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-189	797 square metres of public highway (south of the National Grid Bodelwyddan substation)	Unknown	
11-190	3730 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-190 cont'd		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-191	31865 square metres of agricultural land, pond and well (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-191 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
	12141 square metres of agricultural land and hedgerow (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-192 cont'd		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-193	1144 square metres of hedgerow (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-193 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-194	544 square metres of hedgerow (south of the National Grid Bodelwyddan substation)	Unknown	
11-195	1242 square metres of private road and hedgerow (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Craig Dakin Tan Y Bryn Uchaf Cefn LLANELWY LL17 0HG (in respect of rights of access)	
		LL17 0HG	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-195 cont'd		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)
		Katherine Edwards Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-195a	13 square metres of private road and hedgerow (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Craig Dakin Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	
		Katherine Edwards Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-195a cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
	238 square metres of private road and verges (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights granted by deed dated 4th March 2017) Craig Dakin Tan Y Bryn Uchaf	
		Cefn LLANELWY LL17 0HG (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
Counties of Conwy and Denbighshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-196 cont'd		Katherine Edwards Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access)	
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-197	180786 square metres of agricultural land, hedgerow, pond, access track and electricity pylon (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-197 cont'd		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-197 cont'd		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-197a	1530 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-197a cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 OEY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-197c	6354 square metres of agricultural land, hedgerow, pond, access track and electricity pylon (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-197c cont'd		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-198	13315 square metres of agricultural land and access track (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)	
		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-198 cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams	
		Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-199	5408 square metres of agricultural land and treeline (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-199 cont'd		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)
		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-199 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 OEY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
	24600 square metres of agricultural land and electricity pylon (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-200 cont'd		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)
		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-200 cont'd		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
	33659 square metres of agricultural land and hedgerow (west of the National Grid Bodelwyddan substation)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed of Grant dated 12 November 1992) National Grid Electricity Transmission PLC 1 - 3 Strand	
		London WC2N 5EH (in respect of restrictive covenants and the rights granted by the Deed dated 06 November 2009)	
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by a Deed of Grant dated 12 November 1992)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-201 cont'd		The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect of the rights as stated in the Deed dated 06 November 2009)	
11-202	17087 square metres of agricultural land and hedgerow (west of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-202 cont'd		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)
		Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-202 cont'd		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-203	9570 square metres of agricultural land (west of the National Grid Bodelwyddan substation)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed of Grant dated 12 November 1992)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of restrictive covenants and the rights granted by the Deed dated 06 November 2009)	
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by a Deed of Grant dated 12 November 1992)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-203 cont'd		The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect of the rights as stated in the Deed dated 06 November 2009) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of a Deed dated 10 June 2011)	
	3021 square metres of agricultural land (west of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-204 cont'd		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)
		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)

	Mona Offs	hore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans		Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-204 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993)
		Timothy Michael Bell Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-205	12 square metres of agricultural land (east of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-205 cont'd		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)
		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-205 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
	252 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-206 cont'd		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)
		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-206 cont'd		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-207	1184 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-207 cont'd		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-208	7985 square metres of woodland (Hendy Gorse, south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights to have and use cables contained in a Deed dated 4 May 2017)	
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights contained in a Deed dated 10 June 1977)	
11-209	1717 square metres of woodland (Hendy Gorse, south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights to have and use cables contained in a Deed dated 4 May 2017)	
		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights contained in a Deed dated 10 June 1977)	
11-210	1498 square metres of woodland (south of Glascoed Road)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights to have and use cables contained in a Deed dated 4 May 2017)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-210 cont'd		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights contained in a Deed dated 10 June 1977)	
11-211	2013 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-211 cont'd		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-212	2656 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
	15147 square metres of agricultural land and pond (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-214		Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Craig Dakin Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	
		Katherine Edwards Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-214 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-215	38142 square metres of agricultural land and pond (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-215 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-216	9552 square metres of agricultural land and hedgerow (south of the National Grid Bodelwyddan substation) and public bridleway (BR 208/3)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-216 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-217	822 square metres of agricultural land (south of the National Grid Bodelwyddan substation	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-217 cont'd		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-218	16202 square metres of agricultural land, hedgerow and pond (south of the National Grid Bodelwyddan substation)	Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of right of drainage and access for maintenance) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 GLR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of right of drainage and access for maintenance)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-218 cont'd		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-218a	Bodelwyddan substation)	Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of right of drainage and access for maintenance) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of right of drainage and access for maintenance)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-218a cont'd		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
	5663 square metres of agricultural land and hedgerow (south of the Gwynt y Mor Offshore Wind Farm Substation)	Burbo Extension Ltd 5 Howick Place LONDON SWIP 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Qu14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-219 cont'd		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-220	3075 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014)	
		ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-221	2146 square metres of private road and verges (east of the Gwynt y Mor Offshore Wind Farm Substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014)	
		ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 GPB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-221a	127 Square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014)	
		ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-222	174 square metres of private road (south of the Gwynt y Mor Offshore Wind Farm Substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014)	
		ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans		Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-223	979 square metres of private road and verges (south of the Gwynt y Mor Offshore Wind Farm Substation)	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS (in respect of rights of access)	
		Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-223 cont'd		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)
		Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-223 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights granted by lease dated 16 September 2011)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)	
	29 square metres of private road and verges (south of the Gwynt y Mor Offshore Wind Farm Substation)	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS (in respect of rights of access) Burbo Extension Ltd S Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-224 cont'd		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)
		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-224 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)
		Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights granted by lease dated 16 September 2011)
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	Counties of Conwy and Denbighshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-225	581 square metres of private road (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	
		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-225 cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)
		Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans		Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-226	22 square metres of agricultural land (east of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	
		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-226 cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)
		Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		Timothy Michael Bell Cefn St. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-227	360 square metres of agricultural land (east of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009)	

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-227 cont'd		Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)
		Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-227 cont'd		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993)	
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)	
	1337 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey	
		KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)	

		shore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-228 cont'd		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)
11-229	6518 square metres of agricultural land (south of Glascoed road, B5381)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014)
		ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-229 cont'd		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)	
11-229a	313 square metres agricultural land (south of Glascoed road, B5381)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014)	
		ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-229a cont'd		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)	
11-230	44 square metres of agricultural land (south of Glascoed road, B5381) and public bridleway (BR 208/32)	Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights to install and use electric lines as stated in the Deed of Grant dated 6 November 2009) Unknown	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	counties of Conwy and Denbighshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
	38 square metres of agricultural land (south of Glascoed road, B5381) and public bridleway (BR 208/32)	Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights to install and use electric lines as stated in the Deed of Grant dated 6 November 2009) Unknown	
	33 square metres of private road (off Glascoed road, B5381) and public bridleway (BR 208/32)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights contained within a Lease dated 30 October 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights in relation to the installation and use of electric lines within a Deed of Grant dated 6 November 2009) Unknown	

	Μ	Iona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-232	1147 square metres of private road (south of Glascoed road, B5381)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated in the lease dated 30 October 2014)
		ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 6 January 2009)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-232 cont'd		The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)	
11-233	1439 square metres of agricultural land and hardstanding (south of Glascoed road, B5381)	Burbo Extension Ltd 5 Howick Place LONDON SWIP 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SNS 6PB (in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 6 January 2009)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-233 cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)	
		The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)	
11-233a	783 square metres of agricultural land and hardstanding (south of Glascoed road, B5381)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)	

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2		
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-233a cont'd		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 6 January 2009)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)
		The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)
11-233b 2013 square me	etres of agricultural land (south of Glascoed road, B5381)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated in the lease dated 30 October 2014)

		Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
11-233b cont'd		ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 6 January 2009)
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)
		The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-234		ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of right of way stated in Transfer dated 10 June 2016)	
11-234a	79 square metres of agricultural land (south of Glascoed road, B5381)	ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of right of way stated in Transfer dated 10 June 2016)	
11-234b		ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of right of way stated in Transfer dated 10 June 2016)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-235	194 square metres of private road (south of Glascoed road, B5381)	ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of right of way stated in Transfer dated 10 June 2016)	
11-236	1769 square metres of public highway and verges (Glascoed road, B5381)	Unknown	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Plas-Y-Corddyn, Tan Y Gopa Road, Abergele, LL22 8DT	Bryn Pierce Williams Plas Y Corddyn Tan Y Gopa Road ABERGELE Conwy LL22 8DT	
N/A	Plas Onn Isa, Tan γ Goppa Road, Abergele, LL22 8DT	John Parry Jones Plas Onn Isa Tan Y Gopa Road ABERGELE Conwy LL22 8DT Lyn Jones Plas Onn Isa Tan Y Gopa Road ABERGELE Conwy LL22 8DT	
N/A	Nursery Cottage, Old Gwrych Gardens, Llanddulas Road, Abergele, LL22 8ET	Matthew David Jones Nursery Cottage Llanddulas Road Abergele LL22 8EU	
N/A	Penrefail Camp, Penrefail Crossroads, Moelfre, Abergele, LL22 8PN	James Moloney Journeys End Mylers Town Robertstown County Kildare IRELAND	
N/A	Pwll-Y-Cibau Bach, Moelfre, Abergele, LL22 9RN	David Anthony Jones Pwll Y Cibau Bach Moelfre ABERGELE LL22 9RN	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Jane Margaret Jones Pwll Y Cibau Bach Moelfre ABERGELE LL22 9RN	
N/A	Tyn-Y-Mynydd, Moelfre, Abergele, LL22 9RW	Philip James Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW Stephanie Elizabeth Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW	
N/A	Waterloo, Glascoed, St.Asaph, Conwy, LL22 9DE	Kevin Arthur Bainbridge Waterloo Glascoed ABERGELE LL22 9DE Kirsty Bainbridge Waterloo Glascoed ABERGELE LL22 9DE	
N/A	Glascoed Lodge, Glascoed, Abergele, LL22 9DE	Gary Stephen Johnston Glascoed Lodge Glascoed ABERGELE LL22 9DE	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Suzanne Johnston Glascoed Lodge Glascoed ABERGELE LL22 9DE	
	Glascoed Lodge, Glascoed, Abergele, LL22 9DE	Gary Stephen Johnston Glascoed Lodge Glascoed ABERGELE LL22 9DE Suzanne Johnston Glascoed Lodge Glascoed ABERGELE LL22 9DE	
N/A	Glascoed Lodge, Glascoed, Abergele, LL22 9DE	Gary Stephen Johnston Glascoed Lodge Glascoed ABERGELE LL22 9DE Suzanne Johnston Glascoed Lodge Glascoed ABERGELE LL22 9DE	
N/A	Cwm Massey, Glascoed, Abergele, LL22 9DE	Benjamin Jon Madeley Cwm Massey Glascoed ABERGELE LL22 9DE Helen Louise Evans Cwm Massey Glascoed ABERGELE LL22 9DE	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Tan Y Bryn, Cefn, St Asaph, LL17 0HG	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	
N/A	Cae Llwyd, Cefn, St Asaph, LL17 OHG	Harry Watkin Williams-Wynn Forsters LLP 22 Baker Street LONDON W1U 3BW	
N/A	Tyddyn Meredydd, Cefn, St Asaph, LL17 OHG	Kathryn Margaret Hussey Tyddyn Meredydd Cefn ST. ASAPH Denbighshire LL17 OHG Martyn Richard Hussey Tyddyn Meredydd Cefn ST. ASAPH Denbighshire LL17 OHG	
N/A	Cottage Villa, Glascoed Road, St Asaph, LL17 0LH	Linda Louise Griffiths Cottage Villa Glascoed Road LLANELWY LL17 0LH	
N/A	Llys Y Wennol, Glascoed Road, St Asaph, LL17 0LH	Carol Crook Llys Y Wennol Glascoed Road LLANELWY LL17 0LH	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 2 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Grahame Stewart Crook Llys Y Wennol Glascoed Road LLANELWY LL17 0LH	
N/A	Ty Crwn, Llanddulas Road, Abergele, LL22 8EU	Mark Baker 21 Lisvane Street CARDIFF CF24 4LH	
N/A	Tan Y Bryn Uchaf, Cefn, St Asaph, LL17 OHG	The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
01-001	177297 square metres of foreshore land seaward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of the right to lay and maintain sea outfall pipes granted by the Deed dated 20 February 1955 and in respect of the rights to place and maintain groynes as stated in the Deed dated 1 May 1964 and in respect of the rights contained in the Deed dated 4 April 1973)	
		Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 OLT (in respect of the Deed dated 01 April 1997)	
01-003	10945 square metres of foreshore landward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground sewerage apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-003 cont'd		Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) Unknown	
	167 square metres of foreshore landward of the Mean High Water (north of the A55, Abergele)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water and sewerage apparatus) Unknown	
01-005	2946 square metres of public highway (Sea Road, Abergele)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water and sewerage apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-005 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown	
01-006	395 square metres of private road (north of the A55, Abergele)	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water and sewerage apparatus) Mark Baker 21 Lisvane Street CARDIFF CF24 LH (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-006 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown	
	101 square metres of private road (north of the A55, Abergele) (excluding all interests of the crown)	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water and sewerage apparatus) Julie Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012)	

		Mona Offshore Wind Limited Development Consent Order	
		BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-007 cont'd		Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) Michael Williams 162 Ffordd Y Foryd Bae Cinmel Y RHYL LL18 5LS (in respect of rights of access to maintain services as contained in a Lease dated 2 June 2006) Nicholas Stephen Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012) Pjan Limited Unit 32 Llys Edmund Prys St. AsapH Flintshire LL17 0JA (in respect of rights of access as contained in a Lease dated 3 November 2010)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-007		Sian Williams	
cont'd		5 Terfyn Court Terfyn Bodelwyddan RHYL Denbigshire LL18 5SW (in respect of rights contained in a Deed dated 21 September 2016) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964) Unknown (in respect of rights contained in a Conveyance dated 29 July 1935)	
01-008	325 square metres of private road (north of the A55, Abergele)	Anthony Arnold Rowson Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 OLT (in respect of underground sewerage apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
	Names of all those entitled to enjoy easements or other private rights over land		
Number on	Description of land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
01-008		Julie Williams	
cont'd		Pantri Bach	
		Promenade	
		ABERGELE	
		LL22 7PP	
		(in respect of rights of access)	
		Mark Baker	
		21 Lisvane Street	
		CARDIFF	
		CF24 4LH	
		(in respect of rights of access)	
		Michael Williams	
		162 Ffordd Y Foryd	
		Bae Cinmel	
		Y RHYL	
		LL18 5LS	
		(in respect of rights of access)	
		Nicheles Stenhen Williams	
		Nicholas Stephen Williams Pantri Bach	
		Promenade	
		ABERGELE	
		LL22 7PP	
		(in respect of rights of access)	
		Pjan Limited	
		Unit 32	
		Llys Edmund Prys	
		St. Asaph Business Park ST. ASAPH	
		Flintshire	
		LL17 0JA	
		(in respect of rights of access)	

	Mona Offsh	nore Wind Limited Development Consent Order	
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-008		Unknown	
cont'd			
01-009	1931 square metres of private road and verges (north of the A55, Abergele) (excluding all	Anthony Arnold Rowson	
	interests of the crown)	Castle Cove Caravan Park	
		Promenade	
		ABERGELE	
		LL22 7PP	
		(in respect of rights of access)	
		Julie Williams	
		Pantri Bach	
		Promenade	
		ABERGELE	
		LL22 7PP	
		(in respect of rights of access as contained in a Lease dated 5 July 2012)	
		Mark Baker	
		21 Lisvane Street	
		CARDIFF	
		CF24 4LH	
		(in respect of rights of access)	
		Michael Williams	
		162 Ffordd Y Foryd	
		Bae Cinmel	
		YRHYL	
		LL18 5LS	
		(in respect of right of access to maintain services as contained in a Lease dated 2 June 2006)	
		Network Rail Infrastructure Limited	
		Waterloo General Office	
		LONDON	
		SE1 8SW	
		(in respect of rights contained in a Conveyance dated 29 July 1935)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
01.000		(Applications: Prescribed Forms and Procedures) Regulations 2009	
01-009 cont'd		Nicholas Stephen Williams Pantri Bach	
cont u		Promenade	
		ABERGELE	
		LL22 7PP	
		(in respect of rights of access as contained in a Lease dated 5 July 2012)	
		Pjan Limited	
		Unit 32	
		Llys Edmund Prys	
		St. Asaph Business Park	
		ST. ASAPH	
		Flintshire	
		LL17 OJA (in respect of view of any to maintain convicts an approximation of the scheme dated 2 Neuromber 2010)	
		(in respect of right of access to maintain services as contained in a Lease dated 3 November 2010)	
		Sian Williams	
		5 Terfyn Court	
		Terfyn	
		Bodelwyddan	
		RHYL	
		Denbigshire	
		LL18 5SW	
		(in respect of rights contained in a Deed dated 21 September 2016)	
		The King's Most Excellent Majesty In Right Of His Crown	
		1 St. James's Market	
		LONDON	
		SW1Y 4AH	
		(in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-010	386 square metres of private road and verges (north of the A55, Abergele) (excluding all	Anthony Arnold Rowson	
01-010	interests of the crown)	Castle Cove Caravan Park Promenade ABERGELE LL22 7PP (in respect of rights of access) Julie Williams Pantri Bach Promenade ABERGELE LL22 7PP (in respect of rights of access as contained in a Lease dated 5 July 2012) Mark Baker 21 Lisvane Street CARDIFF CF24 4LH (in respect of rights of access) Michael Williams 162 Ffordd Y Foryd	
		Bae Cinmel Y RHYL LL18 5LS (in respect of right access to maintain services as contained in a Lease dated 2 June 2006) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rights contained in a Conveyance dated 29 July 1935)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number		Names of all those entitled to enjoy easements or other private rights over land	
Number on Land Plans	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
Lallu Pialis		(Applications: Prescribed Forms and Procedures) Regulations 2009	
01-010		Nicholas Stephen Williams	
cont'd		Pantri Bach	
contru		Promenade	
		ABERGELE	
		LL22 7PP	
		(in respect of rights of access as contained in a Lease dated 5 July 2012)	
		Pjan Limited	
		Unit 32	
		Llys Edmund Prys	
1 1		St. Asaph Business Park	
		ST. ASAPH	
		Flintshire	
		LL17 OJA	
		(in respect of rights of access to maintain services as contained in a Lease dated 3 November 2010)	
		Sian Williams	
		5 Terfyn Court	
		Terfyn	
		Bodelwyddan	
		RHYL	
		Denbigshire	
		LL18 5SW	
		(in respect of rights contained in a Deed dated 21 September 2016)	
		The King's Most Excellent Majesty In Right Of His Crown	
		1 St. James's Market	
		LONDON	
		SW1Y 4AH	
		(in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964)	

I	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	ſ	ounties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
Lana mans		(Applications: Prescribed Forms and Procedures) Regulations 2009	
01-011	11530 square metres of private road and verges (north of the A55, Abergele) (excluding all		
	interests of the crown)	Castle Cove Caravan Park	
		Promenade	
		ABERGELE	
		LL22 7PP	
		(in respect of rights of access)	
		Dŵr Cymru Cyfyngedig	
		Linea	
		Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 0LT	
		(in respect of underground sewerage apparatus)	
		Mark Baker	
		21 Lisvane Street	
		CARDIFF	
		CF24 4LH	
		(in respect of rights of access)	
		Unknown	
02-013	167 square metres of private road, hardstanding and grassland (Beach House Road)	SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of underground electricity apparatus)	
		Unknown	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-013a	58 square metres of private road, hardstanding and grassland (Beach House Road) (excluding all interests of the crown)	Unknown	
02-014	116 square metres of private road and railway (Beach House Road)	Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWVN BAY Clwyd LL29 8BF (in respect of rights of access) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-014 cont'd		Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (in respect of rights of access) Unknown	
02-015		Jennings Building & Civil Engineering Limited Plas Eirias Business Centre Abergele Road COLWYN BAY Clwyd LL29 8BF (in respect of rights of access) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Thornley Leisure (Sales) Limited William Sutcliffe Suite Raymond Court Princes Drive Colwyn Bay CONWY LL29 8HT (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
Number on Land Plans	C Description of Land	Counties of Conwy and Denbighshire Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-015 cont'd		Unknown	
02-016	15 square metres of hardstanding and grassland (north of the A55, Abergele) (excluding all interests of the crown)	Unknown	
02-017	8927 square metres of grassland and track (north of Abergele Road, A547) (excluding all interests of the crown)	Unknown	
02-018	5529 square metres of hardstanding, grassland and track (north of Abergele Road, A547) (excluding all interests of the crown)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of agreement to grant an easement stated in Deed dated 8 July 1994) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
Counties of Conwy and Denbighshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-019	92 square metres of grassland (north of Abergele Road, A547)	Unknown	
02-020	1140 square metres of grassland (north of Abergele Road, A547)	Unknown	
02-021	13245 square metres of foreshore land landward of the Mean High Water and cycle way (north of the A55, Abergele) (excluding all interests of the crown)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground sewerage apparatus) Unknown	
02-021a	3287 square metres of foreshore land landward of the Mean High Water and cycle way (north of the A55, Abergele) (excluding all interests of the crown)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground sewerage apparatus) Unknown	
02-022	4484 square metres of grassland (lying to the north of the A55, Abergele)	Unknown	
02-023	665 square metres of grassland (north of the A55, Abergele)	Unknown	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-024	2517 square metres of grassland (north of the A55, Abergele)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of agreement to grant an easement stated in Deed dated 8 July 1994) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown	
02-025	19299 square metres of grassland and railway (north of the A55, Abergele)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown	
	19178 square metres of public highway, verges and hedgerow (A55, Abergele) (excluding all interests of the crown)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 OLT (in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-027 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Zayo Group UK Limited The Relay Building 114 Whitechapel High Street London E1 7PT (in respect of underground telecommunication apparatus)	
02-028	1882 square metres of agricultural land (north of Abergele Road, A457)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	

	Mona Offs	hore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
02-029	3879 square metres of public highway and verges (Abergele road, A547) (excluding all	Conwy County Borough Council
l	interests of the crown)	Bodlondeb
l		Bangor Road
l		CONWY
		Gwynedd
1		LL32 8DU
		(in respect of street furniture)
		Dŵr Cymru Cyfyngedig
		Linea
		Fortran Road
1		St. Mellons
		CARDIFF
		CF3 OLT
		(in respect of underground water apparatus)
		Openreach Limited
		6 Gracechurch Street
		LONDON
		EC3V 0AT
		(in respect of underground telecommunication apparatus)
		Unknown
		Zayo Group UK Limited
		The Relay Building
		114 Whitechapel High Street
		London
		E1 7PT
		(in respect of underground telecommunication apparatus)

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
	509 square metres of public highway and verges (Abergele road, A547) (excluding all	Dŵr Cymru Cyfyngedig	
	interests of the crown)	Linea Factore Decid	
		Fortran Road St. Mellons	
		St. Mellons CARDIFF	
		CARDIFF CF3 0LT	
		(in respect of underground water apparatus)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of underground telecommunication apparatus)	
		Unknown	
		Zayo Group UK Limited	
		The Relay Building	
		114 Whitechapel High Street	
		London	
		E1 7PT	
		(in respect of underground telecommunication apparatus)	
02-031	108 square metres of access splay (south of Abergele Road, A547)	Abergele Golf Club Limited	
		Tan Y Gopa	
		ABERGELE	
		Clwyd	
		LL22 8DS	
		(in respect of the right of access, sewerage and water pipes and drainage granted by the conveyance dated 1st June 1959)	
		1	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-031 cont'd		Matthew David Jones Nursery Cottage Llanddulas Road Abergele LL22 8EU (in respect of rights of access) Robert William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (in respect of rights of access)	
02-032	63763 square metres of agricultural land, access track and hedgerow (south of Abergele Road, A547)	Abergele Golf Club Limited Tan Y Gopa ABERGELE Clwyd LL22 8DS (in respect of the right of access, sewerage pipes, water pipes and drainage granted by the conveyance dated 1 June 1959) Matthew David Jones Nursery Cottage Llanddulas Road Abergele LL22 8EU (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	C	counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-032 cont'd		Robert William Roberts Tan Yr Ogof Farm Abergele Road Llanddulas ABERGELE Conwy LL22 8EY (in respect of rights of access) Unknown (in respect of the right of way granted by the conveyance dated 25 May 1960)	
02-033	69448 square metres of agricultural land and access track (south of Abergele Road, A547)	Unknown	
	1323 square metres of woodland (south of Abergele Road, A547) (excluding all interests of the crown)	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of rights granted by conveyance dated 16th May 1968) The Natural Resources Body For Wales Cambria House 29 Newport Road CARDIFF CF24 0TP (in respect of the rights of access) Unknown (in respect of the rights granted by the Conveyance dated 13 December 1946)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-034 cont'd		Unknown (in respect of personal covenants contained in the Transfer dated 12 December 2006)	
		Unknown (in respect of rights granted by conveyance dated 17th October 1946)	
02-035	807 square metres of private access track (south of Abergele Road, A547)	The Natural Resources Body For Wales Cambria House 29 Newport Road CARDIFF CF24 OTP (in respect of rights of access) Unknown (in respect of the rights granted by the Conveyance dated 13 December 1946) Unknown (in respect of personal covenants contained in the Transfer dated 12 December 2006)	
02-036	33953 square metres of woodland and access track (south of Abergele Road, A547) (excluding all interests of the crown)	Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of rights granted by conveyance dated 16th May 1968 and leases dated 7th August 1951 and 19 September 1953) The Natural Resources Body For Wales Cambria House 29 Newport Road CARDIFF CF24 0TP (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-036 cont'd		Unknown (in respect of the rights granted by the Conveyance dated 13 December 1946)	
		Unknown (in respect of personal covenants contained in the Transfer dated 12 December 2006)	
		Unknown (in respect of rights granted by conveyance dated 17 October 1946)	
03-037	44797 square metres of agricultural land and hedgerow (south of Abergele Road, A547)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)	
		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	

1	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-038	1224 square metres of agricultural land (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
03-039	165 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-039 cont'd		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
03-040	186 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002) Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)	
03-041	162 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-041 cont'd		The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
03-042	180 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002) Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)	
03-043	17 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	C	Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-043		The Executor of the Estate of the Late Ivor Williams	
cont'd		Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
03-044	2284 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
03-045	19034 square metres of agricultural land, hedgerow and access track (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-045 cont'd		Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)	
03-046	1494 square metres of hardstanding (Henblas Farm), private road and access track (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE Ll22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown (in respect of the rights contained in a Conveyance dated 17 October 1946)	
03-047	815 square metres of agricultural land and access track (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-047 cont'd		Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	
03-048	656 square metres of access track (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-049	24349 square metres of agricultural land and hedgerow (north of Tan-Y-Gopa Road)	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd Ll22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002) Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy Ll22 8DS (in respect of rights stated in a transfer dated the 18th November 2004) Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy Ll22 8DS (in respect of rights stated in a transfer dated the 18th November 2004) Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy Ll22 8DS (in respect of rights stated in a transfer dated the 18th November 2004) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown (in respect of the rights reserved contained in the Conveyance dated 17 October 1946)	

		Mona Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-050	45 square metres of agricultural land (north of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997) Gwrych Castle Preservation Trust Limited Gwrych Castle Llanddulas Road ABERGELE Conwy LL22 8ET (in respect of a right to entry to maintain apparatus as contained in a Conveyance dated 17 October 1946) The Executor of the Estate of the Late Ivor Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of a right to a supply of water to owners farm as contained in a Deed dated 4 April 1997)
03-051	657 square metres of access track (north of Tan-Y-Gopa Road)	Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002)

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
03-051		Helen Elmira Cato	
cont'd		Nant Y Bella Lodge	
cont u		Tan-Y-Gopa Road	
		ABERGELE	
		Conwy	
		LL22 8DS	
		(in respect of rights stated in a transfer dated the 18th November 2004)	
		Philip Roy Cato	
		Nant Y Bella Lodge	
		Tan-Y-Gopa Road	
		ABERGELE	
		Conwy	
		LL22 8DS	
		(in respect of rights stated in a transfer dated the 18th November 2004)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
		Unknown	
		(in respect of the rights reserved contained in the Conveyance dated 17 October 1946)	
03-052	1867 square metres of agricultural land and access track (north of Tan-Y-Gopa Road)	Elwyn John Jacobs	
		Henblas	
		Tan-Y-Gopa Road	
		ABERGELE	
		Clwyd	
		LL22 8DT	
		(in respect of the rights granted by the Transfer dated 30 September 2002)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
03-052		Helen Elmira Cato	
cont'd		Nant Y Bella Lodge	
		Tan-Y-Gopa Road	
		ABERGELE	
		Conwy	
		LL22 8DS	
		(in respect of rights stated in a transfer dated the 18th November 2004)	
		· ····································	
		Philip Roy Cato	
		Nant Y Bella Lodge	
		Tan-Y-Gopa Road	
		ABERGELE	
		Conwy LL22 8DS	
		(in respect of rights stated in a transfer dated the 18th November 2004)	
		(in respect of rights stated in a transfer dated the 18th November 2004)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		СН43 ЗЕТ	
		(in respect of overhead electricity apparatus)	
		Unknown	
		(in respect of the rights reserved contained in the Conveyance dated 17 October 1946)	
03-053	814 square metres of agricultural land and access track (north of Tan-Y-Gopa Road)	Bryn Pierce Williams	
		Plas Y Corddyn	
		Tan Y Gopa Road	
		ABERGELE	
		Conwy	
		LL22 8DT	
		(in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
	(Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
03-053		Elwyn John Jacobs	
cont'd		Henblas Tar M Gara Dead	
		Tan-Y-Gopa Road ABERGELE	
		Clwyd	
		LL22 8DT	
		(in respect of the rights granted by the Transfer dated 30 September 2002)	
		Helen Elmira Cato	
		Nant Y Bella Lodge	
		Tan-Y-Gopa Road	
		ABERGELE	
		Conwy	
		LL22 8DS	
		(in respect of rights stated in a transfer dated the 18th November 2004)	
		Philip Roy Cato	
		Nant Y Bella Lodge	
		Tan-Y-Gopa Road	
		ABERGELE	
		Conwy	
		LL22 8DS	
		(in respect of rights stated in a transfer dated the 18th November 2004)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET (in respect of overhead electricity apparatus)	
		(in respect of overhead electricity apparatus)	
		Unknown	
		(in respect of the rights reserved contained in the Conveyance dated 17 October 1946)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-054	25 square metres of access track (north of Tan-Y-Gopa Road)	Unknown	
03-055	2 square metres of access track (north of Tan-Y-Gopa Road)	Unknown	
03-056	121 square metres of public highway (Tan-Y-Gopa Road)	Unknown (in respect of the rights granted by a Deed dated 15 November 1984)	
		Unknown	
03-057	54 square metres of public highway (Tan-Y-Gopa Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown (in respect of the rights granted by a Deed dated 15 November 1984) Unknown	
03-058	58 square metres of access track (south of Tan-Y-Gopa Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown (in respect of the rights granted by a Deed dated 15 November 1984)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	C	Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-058 cont'd		Unknown	
03-059	3 square metres of hedgerow (south of Tan-Y-Gopa Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	
03-060		Elwyn John Jacobs Henblas Tan-Y-Gopa Road ABERGELE Clwyd LL22 8DT (in respect of the rights granted by the Transfer dated 30 September 2002) Helen Elmira Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004) Philip Roy Cato Nant Y Bella Lodge Tan-Y-Gopa Road ABERGELE Conwy LL22 8DS (in respect of rights stated in a transfer dated the 18th November 2004)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-060 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown (in respect of the rights reserved contained in the Conveyance dated 17 October 1946)	
03-061	823 square metres of public highway (Tan-Y-Gopa Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown	
03-062	3 square metres of agricultural land (south of Tan-Y-Gopa Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-062		SP Manweb PLC	
cont'd		3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)	
		Unknown	
	45009 square metres of agricultural land (south of Tan-Y-Gopa Road) and public footpath (FP 16/14)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights to access, use, maintain and repair water pipe contained in the Deed dated 15 November 1984) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)	
		Unknown (in respect of the rights granted by a Deed dated 15 November 1984)	

	Mona Of	fshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
02.064	FOE causes matrice of bardstanding and access track (cauth of Tan V Cons Read)	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-064	585 square metres of hardstanding and access track (south of Tan-Y-Gopa Road)	Glenys Pierce Williams Llys Awel Tan-y-Gopa Road ABERGELE LL22 8DT (in respect of the rights to access, use, maintain and repair water pipe contained in the Deed dated 15 November 1984) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground and overhead electricity apparatus)
04-065	481 square metres of public highway (Tan-Y-Gopa Road)	Unknown
04-066	130 square metres of agricultural land (south of Tan-Y-Gopa Road)	Roger Pickles Brenton House Drakewalls GUNNISLAKE PL18 9EL (in respect of the rights granted by the Conveyance dated 22 October 1946)
04-067	35792 square metres of agricultural land and hedgerow (south of Tan-Y-Gopa Road)	Roger Pickles Brenton House Drakewalls GUNNISLAKE PL18 9EL (in respect of the rights granted by the Conveyance dated 22 October 1946)

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-068	281 square metres of agricultural land (south of Tan-Y-Gopa Road)	Roger Pickles Brenton House Drakewalls GUNNISLAKE PL18 9EL (in respect of the rights granted by the Conveyance dated 22 October 1946)	
04-073	685 square metres of public highway (east of Pant Idda)	Unknown	
04-074	44216 square metres of agricultural land and hedgerow (east of Pant Idda) and public footpath (FP 04/43 and FP 04/44)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Michael Taylor Leach Nant Ganol Y Nentydd Rhyd-y-Foel ABERGELE Conwy LL22 8EF (in respect of right of access to relay and maintain pipes and supply of waterways contained in a Transfer dated 7 June 1996) Openreach Limited 6 Grazechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)	

[Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-074		SP Manweb PLC	
cont'd		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
		Wales and West Utilities Limited	
		Wales & West House	
		Spooner Close	
		Coedkernew	
		CASNEWYDD	
		NP10 8FZ	
		(in respect of rights stated in a deed dated the 20th January 1955 and a deed dated the 1st August 1969)	
		Wales and West Utilities Limited	
		Wales & West House	
		Spooner Close	
		Coedkernew	
		CASNEWYDD	
		NP10 8FZ	
		(in respect of underground gas apparatus)	
04-075	33 square metres of public highway (east of Pant Idda)	Unknown	
04-076	2638 square metres of agricultural land (east of Pant Idda)	Dŵr Cymru Cyfyngedig	
	(Linea	
		Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 0LT	
		(in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
	Names of all those entitled to enjoy easements or other private rights over land		
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
04-076		Michael Taylor Leach	
cont'd		Nant Ganol	
		Y Nentydd	
		Rhyd-y-Foel	
		ABERGELE	
		Conwy	
		LL22 8EF	
		(in respect of right of access to relay and maintain pipes and supply of waterways contained in a Transfer dated 7 June 1996)	
		Wales and West Utilities Limited	
1		Wales & West House	
		Spooner Close	
		Coedkernew	
		CASNEWYDD	
		NP10 8FZ	
		(in respect of rights stated in a deed dated the 20th January 1955 and a deed dated the 1st August 1969)	
		Wales and West Utilities Limited	
		Wales & West House	
		Spooner Close	
		Coedkernew	
		CASNEWYDD	
		NP10 8FZ	
		(in respect of underground gas apparatus)	
04-077	1667 square metres of agricultural land, hedgerow and access track (east of Pant Idda)	Dŵr Cymru Cyfyngedig	
		Linea	
		Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 0LT	
		(in respect of underground water apparatus)	

Mona Off	shore Wind Limited Development Consent Order
	BOOK OF REFERENCE - PART 3
	Counties of Conwy and Denbighshire
Names of all those entitled to enjoy easements or other private rights over land	
Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
	(Applications: Prescribed Forms and Procedures) Regulations 2009
	Michael Taylor Leach
	Nant Ganol
	Y Nentydd
	Rhyd-y-Foel
	ABERGELE
	Сопму
	LL22 8EF
	(in respect of right of access to relay and maintain pipes and supply of waterways contained in a Transfer dated 7 June 1996)
	SP Manweb PLC
	3 Prenton Way
	North Cheshire Trading Estate
	PRENTON
	Merseyside
	CH43 3ET
	(in respect of overhead electricity apparatus)
	Wales and West Utilities Limited
	Wales & West House
	Spooner Close
	Coedkernew
	CASNEWYDD
	NP10 8FZ
	(in respect of rights stated in a deed dated the 20th January 1955 and a deed dated the 1st August 1969)
	Wales and West Utilities Limited
	Wales & West House
	Spooner Close
	Coedkernew
	CASNEWYDD
	NP10 8FZ
	(in respect of underground gas apparatus)
	Description of Land

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-078	24399 square metres of agricultural land (west of the A548)	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)	
05-080	10390 square metres of agricultural land (west of the A548)	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)	
05-081	632 square metres of hedgerow and verge (west of the A548)	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)	
05-082	760 square metres of public highway and verge (west of the A548)	Unknown	
05-083	3496 square metres of agricultural land (west of the A548)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-084	37069 square metres of agricultural land and hedgerow (west of the A548) and public	Dŵr Cymru Cyfyngedig	
05 004	footpath (FP 04/48)	Linea	
		Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 OLT	
		(in respect of underground water apparatus)	
		lfor Davies	
		Pentre Du	
		Llanfairtalhaiarn	
		ABERGELE	
		Conwy	
		LL22 8TL	
		(in respect of the rights reserved contained in the Conveyance dated 28 October 1988)	
		Menna Davies	
		Pentre Du	
		Llanfairtalhaiarn	
		ABERGELE	
		Conwy	
		LL22 8TL	
		(in respect of the rights reserved contained in the Conveyance dated 28 October 1988)	
		Unknown	
		(in respect of the rights reserved as contained in the Conveyance dated 28 October 1988)	
05-085	289 square metres of agricultural land (west of the A548)	lfor Davies	
		Pentre Du	
		Llanfairtalhaiarn	
		ABERGELE	
		Conwy	
		LL22 8TL	
		(in respect of the rights reserved contained in the Conveyance dated 28 October 1988)	
I			

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-085 cont'd		Menna Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988) Unknown (in respect of the rights reserved as contained in the Conveyance dated 28 October 1988)	
05-086	101 square metres of agricultural land (west of the A548)	Ifor Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988) Menna Davies Pentre Du Llanfairtalhaiarn ABERGELE Conwy	
05-087	13 square metres of agricultural land (west of the A548)	LL22 8TL (in respect of the rights reserved contained in the Conveyance dated 28 October 1988) Unknown (in respect of the rights reserved as contained in the Conveyance dated 28 October 1988) Unknown	

	Mona C	Offshore Wind Limited Development Consent Order	
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-088	15 square metres of agricultural land (west of the A548)	Unknown	
05-089	2 square metres of agricultural land (west of the A548)	Unknown	
05-090	93 square metres of agricultural land (west of the A548)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed dated 30 June 1992) Unknown (in respect of the rights contained in the Conveyance dated 29 September 1909) Unknown (in respect of the rights granted by a Conveyance 23 July 1986)	
05-091	34044 square metres of agricultural land and hedgrow (west of the A548)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed dated 30 June 1992) Unknown (in respect of the rights contained in the Conveyance dated 29 September 1909) Unknown (in respect of the rights granted by a Conveyance 23 July 1986)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-091 cont'd		Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)	
05-092	1496 square metres of access track (west of the A548)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of the rights granted by a Deed dated 30 June 1992) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Unknown (in respect of the rights contained in the Conveyance dated 29 September 1909) Unknown (in respect of the rights granted by a Conveyance 23 July 1986) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)	

	Mona Offshore Wind Limited Development Consent Order			
	BOOK OF REFERENCE - PART 3			
	Counties of Conwy and Denbighshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning		
05-093	53660 square metres of agricultural land, pond and hedgerow (west of the A548)	(Applications: Prescribed Forms and Procedures) Regulations 2009 Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted as contained in a Deed dated 26 May 1993) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Shell U.K. Limited Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed dated 26th May 1993) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)		
05-094	7185 square metres of agricultural land (west of the A548)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted as contained in a Deed dated 26 May 1993)		

	Mona Offshore Wind Limited Development Consent Order		
BOOK OF REFERENCE - PART 3			
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-094 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed dated 26th May 1993) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)	
05-095	5875 square metres of public highway and verge (A548 and B5381)	Conwy County Borough Council Bodlondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of overhead and underground telecommunication apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
05-095 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON	
		Merseyside CH43 3ET (in respect of overhead electricity apparatus)	
		Unknown	
		Wales and West Utilities Limited Wales & West House	
		Spooner Close Coedkernew CASNEWYDD	
		NP10 8FZ (in respect of underground gas apparatus)	
		Zayo Group UK Limited The Relay Building 114 Whitechapel High Street	
		London E1 7PT (in respect of underground telecommunication apparatus)	
06-096	1224 square metres of public highway and verge (north of B5381)	Conwy County Borough Council Bodlondeb	
		Bangor Road CONWY Gwynedd	
		LL32 8DU (in respect of street furniture)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-096 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	
06-098	2988 square metres of public highway and verge (B5381)	Conwy County Borough Council Bodlondeb	
		Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture)	
		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus)	
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)	
		Unknown	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
00.000		(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-099	231 square metres of agricultural land (west of the A548)	Openreach Limited 6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of overhead telecommunication apparatus)	
		(
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
		Unknown	
06-100	45284 square metres of agricultural land and access track (west of the A548)	Dŵr Cymru Cyfyngedig	
		Linea	
		Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 OLT	
		(in respect of underground water apparatus)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of overhead and underground telecommunication apparatus)	
		·	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-100 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown	
06-101	2175 square metres of public highway and verge (A548)	Conwy County Borough Council Bodiondeb Bangor Road CONWY Gwynedd LL32 8DU (in respect of street furniture) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	C	Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-101 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown	
06-102	464 square metres of public highway and verge (A548)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown	
06-103	40796 square metres of agricultural land and hedgerow (east of A548)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-103 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)	
06-104	14643 square metres of agricultural land (east of A548) and electricity pylon	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)	
06-105	2380 square metres of access track (south of the B5381)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect of overhead electricity apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-105 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead and underground electricity apparatus) Unknown	
	67348 square metres of agricultural land and hedgerow (south of the B5381) and public footpath (FP 19/12)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights of access to use, maintain and repair a pipeline granted by a Deed of Grant dated 26th April 1978 and later varied in Deed of Grant dated 10 July 1992) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) National Grid Electricity Transmission PLC	
		1 - 3 Strand London WC2N 5EH (in respect of rights to erect, use, maintain, repair and remove electric lines granted by a Deed of Grant dated 29 October 1970)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 3		
		BOOK OF REFERENCE - PART 3 Counties of Conwy and Denbighshire	
	Names of all those entitled to enjoy easements or other private rights over land		
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-106		National Grid Electricity Transmission PLC	
cont'd		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of overhead electricity apparatus)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of overhead and underground telecommunication apparatus)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of underground and overhead electricity apparatus)	
		/	
		Unknown	
		(in respect of rights of access reserved by a Conveyance dated 24 April 1952)	
06-106a	74279 square metres of agricultural land and hedgerow (south of the B5381) and public	Dŵr Cymru Cyfyngedig	
50 1000	footpath (FP 19/12)	Linea	
		Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 0LT	
		(in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
1	Midita Offsi	BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
06-106a		National Grid Electricity Transmission PLC	
cont'd		1 - 3 Strand	
cont u		London	
		WC2N 5EH	
		(in respect of overhead electricity apparatus)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of rights to erect, use, maintain, repair and remove electric lines granted by a Deed of Grant dated 29 October 1970)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of underground and overhead electricity apparatus)	
		(in respect of underground and overhead electricity apparatus)	
		Will Davies	
		Plas Isaf	
		Groesffordd Marli	
		ABERGELE	
		LL22 9DY	
06-107	764 square metres of agricultural land and access track (south of the B5381) and public	Centrica PLC	
	footpath (FP 19/12)	Millstream	
	, ,	Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of rights of access to use, maintain and repair a pipeline granted by a Deed of Grant dated 26th April 1978 and later varied in	
		Deed of Grant dated 10 July 1992)	

		Mona Offshore Wind Limited Development Consent Order	
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-107		Dŵr Cymru Cyfyngedig	
cont'd		Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights to erect, use, maintain, repair and remove electric lines granted by a Deed of Grant dated 29 October 1970)	
		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus)	
		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus)	
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground and overhead electricity apparatus)	

	Mona Offs	hore Wind Limited Development Consent Order	
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-107 cont'd		Unknown (in respect of rights of access reserved by a Conveyance dated 24 April 1952)	
06-108	14113 square metres of agricultural land and hedgerow (south of the B5381) and public footpath (FP 19/15) (excluding all interests of the crown)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985)	
		Shell Centre York Road LONDON SE1 7NA	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-108 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights to erect, maintain and repair electric lines as granted in a Deed of Grant dated 7 January 1971) Unknown (in respect of right of service media stated in Conveyance dated 26 July 1985)	
	407 square metres of agricultural land and hedgerow (south of the B5381) (excluding all interests of the crown)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect of overhead electricity apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-109 cont'd		Shell U.K. Limited Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights in relation to electricity lines as stated in Deed dated 7 January 1971) Unknown (in respect of the right to access, use, lay, repair and maintain water, soil, gas, electricity and cables as stated in the conveyance dated 26 July 1985)	
07-110	1391 square metres of public highway and verges (south of the B5381) (excluding all interests of the crown)	Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-110 cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights granted as contained in a Deed of Grant dated 30 July 1985) Unknown	
07-111	19599 square metres of agricultural land and pond (south of the B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-112	734 square metres of public highway and hedgerow (south of the B5381) (excluding all	Audrey Denson Morgan	
	interests of the crown)	Fardre	
		St. George	
		ABERGELE	
		Conwy	
		LL22 9RT	
		(in respect of rights granted by Transfer dated 17 January 1997)	
		John Henry Morgan	
		Fardre	
		St. George	
		ABERGELE	
		Conwy	
		(in respect of rights granted by Transfer dated 17 January 1997)	
		Shell U.K. Limited	
		Shell Centre	
		York Road	
		LONDON	
		SE1 7NA	
		(in respect of rights granted as contained in a Deed of Grant dated 30 July 1985)	
		Unknown	
07-113	284 square metres of hedgerow (south of the B5381)	Janice Marie Davies	
		Pistyll	
		Moelfre	
		ABERGELE	
		LL22 9RP	
		(in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-113 cont'd		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
07-114	66 square metres of hedgerow (south of the B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
	3 square metres of hedgerow (south of the B5381)	Unknown	
	56 square metres of private road and verges (south of the B5381) and public bridleway (BR 19/27)	Adrian Farey Ty Newydd Moelfre ABERGELE LL22 9RL (in respect of rights of access)	

	Mona Offs	hore Wind Limited Development Consent Order	
		BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire		
	Names of all those entitled to enjoy easements or other private rights over land		
Number on	Description of Lond	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-116		Glyn Williams	
cont'd		Bryn Goleu	
		Moelfre	
		ABERGELE	
		LL22 9RH	
		(in respect of rights of access)	
		Janice Marie Davies	
		Pistyll	
		Moelfre	
		ABERGELE	
		LL22 9RP	
		(in respect of the rights granted by the Conveyance dated 12 January 1979)	
		Menai Williams	
		Bryn Goleu	
		Moelfre	
		ABERGELE	
		Conwy	
		LL22 9RH	
		(in respect of rights of access)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of underground telecommunication apparatus)	
		Patricia Stanley	
		Ffynnon Dyfyr	
		Moelfre	
		ABERGELE	
		Clwyd	
		LL22 9RH	
		(in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-116 cont'd		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
07-117	1043 square metres of private road and verges (south of the B5381) and public bridleway (BR 19/27)	Adrian Farey Ty Newydd Moelfre ABERGELE LL22 9RL (in respect of rights of access) Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH (in respect of rights of access) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Menai Williams Bryn Goleu Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Menai Williams Bryn Goleu Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Menai Williams Bryn Goleu Moelfre ABERGELE (in respect of rights of access)	

	Mona Offshore Wind Limited Development Consent Order		
BOOK OF REFERENCE - PART 3			
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-117 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Patricia Stanley Ffynnon Dyfyr Moelfre ABERGELE Clwyd LL22 9RH (in respect of rights of access) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
	3198 square metres of private road (south of the B5381) and public bridleway (BR 19/27 and BR 19/19)	Adrian Farey Ty Newydd Moelfre ABERGELE LL22 9RL (in respect of rights of access) Glyn Williams Bryn Goleu Moelfre ABERGELE LL22 9RH (in respect of rights of access)	

	Mon	a Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
07-118		Janice Marie Davies
cont'd		Pistyll
		Moelfre
		ABERGELE
		LL22 9RP
		(in respect of the rights granted by the Conveyance dated 12 January 1979)
		Manai Milliana
		Menai Williams
		Bryn Goleu
		Moelfre ABERGELE
		Conwy LL22 9RH
		(in respect of rights of access)
		(in respect of rights of access)
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect of overhead electricity apparatus)
		Openreach Limited
		6 Gracechurch Street
		LONDON
		EC3V 0AT
		(in respect of underground telecommunication apparatus)
		Patricia Stanley
		Ffynnon Dyfyr
		Moelfre
		ABERGELE
		Clwyd
		LL22 9RH
		(in respect of rights of access)

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-118 cont'd		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
07-119	1638 square metres of hedgerow (south of the B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
07-120	13556 square metres of agricultural land (south of the B5381) (excluding all interests of the crown)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	C	Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-120		SP Manweb PLC	
cont'd		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of rights to install, use, maintain and repair electrical lines granted by a Deed of Grant dated 10 February 2000)	
07-121	924 square metres of hedgerow (south of the B5381) (excluding all interests of the crown)	SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of rights to install, use, maintain and repair electrical lines granted by a Deed of Grant dated 10 February 2000)	
		(
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
07-122	586 square metres of hedgerow (south of the B5381) (excluding all interests of the crown)	Audrey Denson Morgan	
0, 122		Fardre	
		St. George	
		ABERGELE	
		Conwy	
		LL22 9RT	
		(in respect of rights granted by Transfer dated 17 January 1997)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-122 cont'd	3859 square metres of public highway and hedgerow (south of the B5381) (excluding all interests of the crown)	John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) Audrey Denson Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Transfer dated 17 January 1997) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of underground water apparatus)	

		Mona Offshore Wind Limited Development Consent Order	
	BOOK OF REFERENCE - PART 3 Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation $7(1)(c)$ of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-123		Openreach Limited	
cont'd		6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of underground telecommunication apparatus)	
		Shell U.K. Limited	
		Shell Centre	
		York Road	
		LONDON	
		SE1 7NA	
		(in respect of rights granted as contained in a Deed of Grant dated 30 July 1985)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
		Unknown	
		Wales and West Utilities Limited	
		Wales & West House	
		Spooner Close	
		Coedkernew	
		CASNEWYDD	
		NP10 8FZ	
		(in respect of underground gas apparatus)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Processibad Forms and Procedures) Pagulations 2000	
07-124	2532 square metres of hedgerow (south of the B5381)	(Applications: Prescribed Forms and Procedures) Regulations 2009 Audrey Denson Morgan	
07-124	2532 square metres of hedgerow (south of the B5381)	Audrey Denson Morgan Fardre Fardre St. George ABERKGELE Conwy L122 9RT (in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997) Centrica PLC Millstream Maldenhead Road WINDSOR Berkshire SL4 SGD (in respect of the rights granted contained in a Deed dated 27 October 1992) Conwy County Borough Council Bodiondeb Bangor Road L123 B0U (in respect of street furniture) DVir Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-124		Janice Marie Davies	
cont'd		Pistyll	
		Moelfre	
		ABERGELE	
		LL22 9RP	
		(in respect of the rights granted by the Conveyance dated 12 January 1979)	
		John Henry Morgan	
		Fardre	
		St. George	
		ABERGELE	
		Conwy	
		LL22 9RT	
		(in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of underground telecommunication apparatus)	
		Philip James Banfield	
		Tyn Y Mynydd	
		Moelfre	
		ABERGELE	
		Conwy	
		LL22 9RW	
		(in respect of rights granted to access water as contained in a Transfer dated 7 April 1995)	
		Shell U.K. Limited	
		Shell Centre	
		York Road	
		LONDON	
		SE1 7NA	
		(in respect of the rights granted by the Deed dated 27 October 1992)	

	Mona Offs	hore Wind Limited Development Consent Order	
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-124		SP Manweb PLC	
cont'd		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		СН43 ЗЕТ	
		(in respect of overhead electricity apparatus)	
		Stephanie Elizabeth Banfield	
		Tyn Y Mynydd	
		Moelfre	
		ABERGELE	
		Conwy	
		LL22 9RW	
		(in respect of rights granted to access water as contained in a Transfer dated 7 April 1995)	
		Wales and West Utilities Limited	
		Wales & West House	
		Spooner Close	
		Coedkernew	
		CASNEWYDD	
		NP10 8FZ	
		(in respect of underground gas apparatus)	
07-125	89922 square metres of agricultural land, hedgerow (south of the B5381), watercourse	Audrey Denson Morgan	
	(drain) and public bridleway (BR 19/19)	Fardre	
		St. George	
		ABERGELE	
		Conwy	
		LL22 9RT	
		(in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997)	
		1	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
07-125		(Applications: Prescribed Forms and Procedures) Regulations 2009 Janice Marie Davies	
cont'd		Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
		John Henry Morgan Fardre St. George ABERGELE Conwy LL22 9RT (in respect of rights granted by Deed dated 15 January 1997 and Transfer dated 17 January 1997)	
		Nicholas Peter Cooke Pant Y Glyd Moelfre ABERGELE Conwy LL22 9RG (in respect of the rights granted by the Conveyance dated 28 March 1958)	
		Philip James Banfield Tyn Y Mynydd Moelfre ABERGELE Conwy LL22 9RW (in respect of rights granted to access water as contained in a Transfer dated 7 April 1995)	
		Robert Maldwyn Davies 1 Lon Ffawydd ABERGELE LL22 7DU (in respect of rights granted by Transfer dated 24 November 1995)	

		Mona Offshore Wind Limited Development Consent Order	
	BOOK OF REFERENCE - PART 3 Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-125		SP Manweb PLC	
cont'd		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of underground electricity apparatus)	
		Stephanie Elizabeth Banfield	
		Tyn Y Mynydd	
		Moelfre	
		ABERGELE	
		Conwy	
		LL22 9RW	
		(in respect of rights granted to access water as contained in a Transfer dated 7 April 1995)	
		Susan Emma Hancox	
		Pant Y Glyd	
		Moelfre	
		ABERGELE	
		Conwy	
		LL22 9RG	
		(in respect of the rights granted by the Conveyance dated 28 March 1958)	
		William Goronwy Davies	
		Pistyll	
		Moelfre	
		ABERGELE	
		LL22 9RP	
		(in respect of the rights granted by the Conveyance dated 12 January 1979)	
		· · · · · · · · · · · · · · · · · · ·	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-126	1863 square metres of public highway and verge (south of the B5381)	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Unknown	
07-127	13176 square metres of agricultural land and hedgerow (south of the B5381)	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus)	
07-128	750 square metres of agricultural land (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted as contained in a Deed dated 27 October 1992) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
Lanu Flans		(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-129	5773 square metres of agricultural land and hedgerow (south of the B5381)	Philip James Banfield	
		Tyn Y Mynydd	
		Moelfre	
		ABERGELE	
		Conwy	
		LL22 9RW	
		(in respect of rights of access relating to service media and a water pipe in a Transfer dated 22 December 2004)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of underground electricity apparatus)	
		(
		Stephanie Elizabeth Banfield	
		Tyn Y Mynydd	
		Moelfre	
		ABERGELE	
		Conwy	
		LL22 9RW	
		(in respect of rights of access relating to service media and a water pipe in a Transfer dated 22 December 2004)	
07-130	214 square metres of agricultural land and hedgerow (south of the B5381)	Centrica PLC	
		Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of rights granted as contained in a Deed dated 27 October 1992)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number		Names of all those entitled to enjoy easements or other private rights over land	
Number on Land Plans	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
Lanu Plans		(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-130		Dŵr Cymru Cyfyngedig	
cont'd		Linea	
		Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 0LT	
		(in respect of underground water apparatus)	
		Wales and West Utilities Limited	
		Wales & West House	
		Spooner Close	
		Coedkernew	
		CASNEWYDD	
		NP10 8FZ	
		(in respect of underground gas apparatus)	
07-131	2518 square metres of hedgerow (south of B5381)	Centrica PLC	
		Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of the rights granted by the Deed dated 27 October 1992)	
		Dŵr Cymru Cyfyngedig	
		Linea	
		Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 0LT	
		(in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-131		Janice Marie Davies	
cont'd		Pistyll Moelfre ABERGELE	
		LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
		Shell U.K. Limited Shell Centre York Road	
		LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)	
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)	
		Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-132	40 square metres of public highway and hedgerow (B5381)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	
07-133	24389 square metres of agricultuiral land and hedgerow (south of B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus)	

	٨	Nona Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
07-133		Shell U.K. Limited
cont'd		Shell Centre
		York Road
		LONDON SE1 7NA
		(in respect of the rights granted by the Deed dated 27 October 1992)
		(in respect of the rights granted by the beed dated 27 October 1992)
		Wales and West Utilities Limited
		Wales & West House
		Spooner Close
		Coedkernew CASNEWYDD
		NP10 8FZ
		(in respect of underground gas apparatus)
		William Goronwy Davies
		Pistyll
		Moelfre
		ABERGELE
		LL22 9RP
		(in respect of the rights granted by the Conveyance dated 12 January 1979)
07-134	73 square metres of agricultural land (south of the B5381)	Janice Marie Davies
		Pistyll
		Moelfre
		ABERGELE
		LL22 9RP
		(in respect of the rights granted by the Conveyance dated 12 January 1979)
		Openreach Limited
		6 Gracechurch Street
		LONDON
		EC3V OAT
		(in respect of overhead telecommunication apparatus)

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-134 cont'd		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-135	442 square metres of public highway (south of B5381)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Unknown	
	62213 square metres of agricultual land, access track, waterway and hedgerow (south of B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-136		Janice Marie Davies	
cont'd		Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Shell U.K. Limited Shell Centre York Road LONDON	
		SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992) William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-137	23867 square metres of agricultural land, access track and hedgerow (south of B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus)	

ART 3 pighshire Names of all those entitled to enjoy easements or other private rights over land
Names of all those entitled to enjoy easements or other private rights over land
cluding private rights of navigation over water) which it is proposed shall be extinguished,
suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
(Applications: Prescribed Forms and Procedures) Regulations 2009
tate
ectricity apparatus)
anted by the Conveyance dated 12 January 1979)
d underground telecommunication apparatus)
pet
d telecommunication apparatus)
n

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Neurolean an		Names of all those entitled to enjoy easements or other private rights over land	
Number on Land Plans	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-139	736 square metres of public highway and verge (B5381)	Conwy County Borough Council	
00 135	voo square metres of public mgrway and verge (boober)	Bodlondeb	
		Bangor Road	
		CONWY	
		Gwynedd	
		LL32 8DU	
		(in respect of street furniture)	
		Openreach Limited 6 Gracechurch Street	
		b Gracechurch Street LONDON	
		EC3V OAT	
		(in respect of overhead and underground telecommunication apparatus)	
08-140	1158 square metres of public highway and verges (south of the B5381)	Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of overhead telecommunication apparatus)	
		Unknown	
08.141	001 answers making of multiplicity into and had a row (as which of the DE201)	Comuni Countri Dorouch Council	
08-141	881 square metres of public highway and hedgerow (south of the B5381)	Conwy County Borough Council Bodlondeb	
		Bangor Road	
		CONWY	
		Gwynedd	
		LL32 8DU	
		(in respect of street furniture)	

		Mona Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-141		Dŵr Cymru Cyfyngedig
08-141 cont'd		Dwr Cymru Cytyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)
		Unknown Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect to underground gas apparatus)

	Mona Offs	hore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Lond	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation $7(1)(c)$ of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
08-142	1098 square metres of hedgerow (south of the B5381)	Centrica PLC
		Millstream
		Maidenhead Road
		WINDSOR
		Berkshire
		SL4 5GD
		(in respect of the rights granted by the Deed dated 27 October 1992)
		Dŵr Cymru Cyfyngedig
		Linea
		Fortran Road
		St. Mellons
		CARDIFF
		CF3 0LT
		(in respect of underground water apparatus)
		Janice Marie Davies
		Pistyll
		Moelfre
		ABERGELE
		LL22 9RP
		(in respect of the rights granted by the Conveyance dated 12 January 1979)
		Openreach Limited
		6 Gracechurch Street
		EC3V OAT
		(in respect of overhead telecommunication apparatus)
		Shell U.K. Limited
		Shell Centre
		York Road
		LONDON
		SE1 7NA
		(in respect of the rights granted by the Deed dated 27 October 1992)
		•

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
Land Flans		(Applications: Prescribed Forms and Procedures) Regulations 2009	
08-142		SP Manweb PLC	
cont'd		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
		Wales and West Utilities Limited	
		Wales & West House	
		Spooner Close	
		Coedkernew	
		CASNEWYDD	
		NP10 8FZ	
		(in respect of underground gas apparatus)	
		William Goronwy Davies	
		Pistyll Moelfre	
		ABERGELE	
		ABERGELE LL22 9RP	
		(in respect of the rights granted by the Conveyance dated 12 January 1979)	
		(in respect of the rights granted by the conveyance dated 12 January 1979)	
00.142		On any on the limit of	
08-143	25 square metres of hedgerow (south of the B5381)	Openreach Limited	
		6 Gracechurch Street LONDON	
		EC3V 0AT	
		(in respect of overhead telecommunications apparatus)	
		in respect of overhead teleconfinancations apparatus)	
		Unknown	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Lond	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
08-144	1043 square metres of hedgerow (south of the B5381)	Conwy County Borough Council	
		Bodlondeb	
		Bangor Road	
		CONWY	
		Gwynedd	
		LL32 8DU	
		(in respect of street furniture)	
		Janice Marie Davies	
		Pistyll	
		Moelfre	
		ABERGELE	
		LL22 9RP	
		(in respect of the rights granted by the Conveyance dated 12 January 1979)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of overhead telecommunication apparatus)	
		William Goronwy Davies	
		Pistyll	
		Moelfre	
		ABERGELE	
		LL22 9RP	
		(in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-145	50 square metres of public highway and hedgerow (south of the B5381)	Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of overhead telecommunication apparatus)	

	Mona	Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans 08-145	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 Unknown
cont'd 08-146	74666 cause metres of agricultural land and badgerow (couth of the B5281)	Centrica PLC
08-146	74666 square metres of agricultural land and hedgerow (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 OLT (in respect of underground water apparatus) Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of overhead telecommunication apparatus)

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
08-146 cont'd		Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA	
		(in respect of the rights granted by the Deed dated 27 October 1992)	
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate	
		PRENTON Merseyside	
		CH43 3ET (in respect of overhead electricity apparatus)	
		Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)	
		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-147	109 square metres of agricultural land (south of the B5381)	Janice Marie Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-147 cont'd		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-148	392 square metres of agricultural land (south of the B5381)	Centrica PLC Millistream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Janice Marie Davies Pistyll Moeffre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 SET (in respect of overhead electricity apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-148 cont'd		William Goronwy Davies Pistyll Moelfre ABERGELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979)	
08-149	4191 square metres of hedgerow (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by the Deed dated 27 October 1992) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Janice Marie Davies Pistyli Moelfre ABERCELE LL22 9RP (in respect of the rights granted by the Conveyance dated 12 January 1979) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of the rights granted by the Deed dated 27 October 1992)	

		Mona Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
08-149		SP Manweb PLC
cont'd		3 Prenton Way
		North Cheshire Trading Estate
		PRENTON
		Merseyside
		CH43 3ET
		(in respect of overhead electricity apparatus)
		Wales and West Utilities Limited
		Wales & West House
		Spooner Close
		Coedkernew
		CASNEWYDD
		NP10 8FZ
		(in respect of underground gas apparatus)
		William Goronwy Davies
		Pistyll
		Moelfre
		ABERGELE
		LL22 9RP
		(in respect of the rights granted by the Conveyance dated 12 January 1979)
08-150	8599 square metres of agricultural land (south of the B5381)	Kinmel No.3 Company Limited
08-150	square metres of agricultural land (south of the B5381)	St. Johns Chambers
		Love Street
		CHESTER Cheshire
		CH1 1QN
		(in respect of rights granted by the conveyance dated 4 October 1989)

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
08-150 cont'd		(Applications: Prescribed Forms and Procedures) Regulations 2009 Kinmel No.4 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989) Malcolm Egerton Baron Caer Clawdd Groesffordd Marli ABERGELE Conwy Ll22 9DU (in respect of right to lay down and maintain a water pipe as stated in Conveyance dated 11 April 1972) Michael William Verity Park House 37 Lower Bridge Street CHESTER Cheshire CH1 1RS (in respect of sporting rights) Richard William Kenrick Price Rhiwlas Estate Rhiwlas Istate Rhiwlas Hall BALA	
		BALA Gwynedd LL23 7NP (in respect of sporting rights)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3 Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
Land Flans		(Applications: Prescribed Forms and Procedures) Regulations 2009	
08-151	310 square metres of agricultural land (south of the B5381)	Kinmel No.3 Company Limited	
		St. Johns Chambers	
		Love Street	
		CHESTER	
		Cheshire	
		CH1 1QN	
		(in respect of rights granted by the conveyance dated 4 October 1989)	
		Kinmel No.4 Company Limited	
		St. Johns Chambers	
		Love Street	
		CHESTER	
		Cheshire	
		CH1 1QN	
		(in respect of rights granted by the conveyance dated 4 October 1989)	
		Malcolm Egerton Baron	
		Caer Clawdd	
		Groesffordd Marli	
		ABERGELE	
		Conwy	
		LL22 9DU	
		(in respect of right to lay down and maintain a water pipe as stated in Conveyance dated 11 April 1972)	
		Michael William Verity	
		Park House	
		37 Lower Bridge Street	
		CHESTER	
		Cheshire	
		CH1 1RS	
		(in respect of sporting rights)	
I			

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-151 cont'd		Richard William Kenrick Price Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights)	
08-152	19 square metres of public highway and access splay (south of the B5381)	Unknown	
08-153	480 square metres of public highway and hedgerow (south of the B5381)	Unknown	
08-154	20598 square metres of agricultural land (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993) Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)	

		Mona Offshore Wind Limited Development Consent Order	
		BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
08-154		Kinmel No.4 Company Limited	
cont'd		St. Johns Chambers	
		Love Street	
		CHESTER	
		Cheshire CH1 1QN	
		(in respect of rights granted by the conveyance dated 4 October 1989)	
		Michael William Verity	
		Park House	
		37 Lower Bridge Street	
		CHESTER	
		Cheshire	
		CH1 1RS	
		(in respect of sporting rights)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)	
		Richard William Kenrick Price	
		Rhiwlas Estate	
		Rhiwlas Hall	
		BALA	
		Gwynedd	
		LL23 7NP	
		(in respect of sporting rights)	

	Mona Offshore Wind Limited Development Consent Order			
	BOOK OF REFERENCE - PART 3			
	Counties of Conwy and Denbighshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
08-154 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011) Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974)		
09-155	2642 square metres of hedgerow, access track and electricity pylon (south of the B5381)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993) Kinmel No.3 Company Limited St. Johns Chambers Love Street CHESTER Cheshire CH1 1QN (in respect of rights granted by the conveyance dated 4 October 1989)		

	1	Mona Offshore Wind Limited Development Consent Order	
		BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
09-155		Kinmel No.4 Company Limited	
cont'd		St. Johns Chambers	
		Love Street	
		CHESTER	
		Cheshire	
		CH1 1QN	
		(in respect of rights granted by the conveyance dated 4 October 1989)	
		Michael William Verity	
		Park House	
		37 Lower Bridge Street	
		CHESTER	
		Cheshire	
		CH1 1RS	
		(in respect of sporting rights)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)	
		Richard William Kenrick Price	
		Rhiwlas Estate	
		Rhiwlas Hall	
		BALA	
		Gwynedd	
		LL23 7NP	
		(in respect of sporting rights)	
		1	

		Mona Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
09-155 cont'd		(Applications: Prescribed Forms and Procedures) Regulations 2009 SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 87Z
09-156 39	square metres of hedgerow (south of the B5381)	(in respect of underground gas apparatus) Ian Alexander Thomas Johnson Ddol Farmhouse Groesffordd Marli Abergele LL29 9EB (in respect of the rights granted by a Transfer dated 19 January 2001)

		Mona Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-157	22005 square metres of agricultural land (south of the B5381)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Ian Alexander Thomas Johnson Ddol Farmhouse Groesffordd Marli Abergele LL29 9EB (in respect of the rights granted by a Transfer dated 19 January 2001) National Grid Electricity Transmission PLC 1 - 3 Strand
09-158	356 square metres of hedgerow (south of the B5381)	London WC2N 5EH (in respect of overhead electricity apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August 1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July 1993)

		Mona Offshore Wind Limited Development Consent Order	
		BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
09-158		Kinmel No.3 Company Limited	
cont'd		St. Johns Chambers	
		Love Street	
		CHESTER	
		Cheshire	
		CH1 1QN	
		(in respect of rights granted by the conveyance dated 4 October 1989)	
		Kinmel No.4 Company Limited	
		St. Johns Chambers	
		Love Street	
		CHESTER	
		Cheshire	
		CH1 1QN	
		(in respect of rights granted by the conveyance dated 4 October 1989)	
		Michael William Verity	
		Park House	
		37 Lower Bridge Street	
		CHESTER	
		Cheshire	
		CH1 1RS	
		(in respect of sporting rights)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)	

	Ν	Mona Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
09-158		(Applications: Prescribed Forms and Procedures) Regulations 2009 Richard William Kenrick Price
cont'd		Rhiwlas Estate
contru		Rhiwlas Hall
		BALA
		Gwynedd
		LL23 7NP
		(in respect of sporting rights)
		SP Manweb PLC
		3 Prenton Way
		North Cheshire Trading Estate
		PRENTON
		Merseyside
		CH43 3ET
		(in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July
		2011)
		Unknown
		(in respect of rights granted by a Deed of Grant dated 25 April 1974)
09-159	15703 square metres of agricultural land (south of the B5381)	Centrica PLC
		Millstream
		Maidenhead Road
		WINDSOR
		Berkshire
		SL4 5GD
		(in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August
		1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July
		1993)

	Mon	a Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	- · · · ·	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
09-159		Kinmel No.3 Company Limited
cont'd		St. Johns Chambers
		Love Street
		CHESTER
		Cheshire
		CH1 1QN
		(in respect of rights granted by the conveyance dated 4 October 1989)
		Kinmel No.4 Company Limited
		St. Johns Chambers
		Love Street
		CHESTER
		Cheshire CH1 1QN
		(in respect of rights granted by the conveyance dated 4 October 1989)
		Michael William Verity
		Park House
		37 Lower Bridge Street
		CHESTER
		Cheshire
		CH1 1RS
		(in respect of sporting rights)
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect of overhead electricity apparatus)
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)
		· · · · · · · · · · · · · · · · · · ·

	Mon	a Offshore Wind Limited Development Consent Order
	WOI	BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
09-159		Openreach Limited
cont'd		6 Gracechurch Street
		LONDON
		EC3V OAT
		(in respect of overhead and underground telecommunication apparatus)
		Richard William Kenrick Price
		Rhiwlas Estate
		Rhiwlas Estate
		BALA
		Gwynedd LL23 7NP
		(in respect of sporting rights)
		(in respect of sporting rights)
		SP Manweb PLC
		3 Prenton Way
		North Cheshire Trading Estate
		PRENTON
		Merseyside
		CH43 3ET
		(in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July
		2011)
		Unknown
		(in respect of rights granted by a Deed of Grant dated 25 April 1974)
09-160	51355 square metres of agricultural land (south of the B5381)	Centrica PLC
		Millstream
		Maidenhead Road
1		WINDSOR
		Berkshire
		SL4 5GD
		(in respect of the rights to access, maintain and repair the gas pipeline and restrictive covenants granted by the deed dated 5th August
		1992 and in respect to the right to access, lay and maintain cables and restrictive covenants as stated in the agreement dated 30th July
		1993)

		Mona Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
09-160		Kinmel No.3 Company Limited
cont'd		St. Johns Chambers
		Love Street
		CHESTER
		Cheshire
		CH1 1QN
		(in respect of rights granted by the conveyance dated 4 October 1989)
		Kinmel No.4 Company Limited
		St. Johns Chambers
		Love Street
		CHESTER
		Cheshire
		CH1 1QN
		(in respect of rights granted by the conveyance dated 4 October 1989)
		Michael William Verity
		Park House
		37 Lower Bridge Street
		CHESTER
		Cheshire
		CH1 1RS
		(in respect of sporting rights)
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect to the rights to erect, maintain, repair, inspect and renew electric lines as stated in a Deed dated 9th July 2012)
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect of overhead electricity apparatus)

		Mona Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
09.160		(Applications: Prescribed Forms and Procedures) Regulations 2009
09-160 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of overhead and underground telecommunication apparatus) Richard William Kenrick Price Rhiwlas Estate Rhiwlas Estate Rhiwlas Hall BALA Gwynedd LL23 7NP (in respect of sporting rights) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect to the right to lay, maintain and repair electrical lines and in respect to restrictive covenants granted by the deed dated 29 July 2011) SP Manweb PLC 3 Prenton Way
		North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)
		Unknown (in respect of rights granted by a Deed of Grant dated 25 April 1974)

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
09-160		Wales and West Utilities Limited	
cont'd		Wales & West House	
		Spooner Close	
		Coedkernew	
		CASNEWYDD	
		NP10 8FZ	
		(in respect of underground gas apparatus)	
09-161	2683 square metres of access track (south of B5381)	Gareth Hughes	
		Tan Dderwen	
		Glascoed	
		ABERGELE	
		LL22 9DE	
		(in respect of rights of access)	
		Gwen Vaughan Hughes	
		Tan Dderwen	
		Glascoed	
		ABERGELE	
		LL22 9DE	
		(in respect of rights of access)	
		Huw Lloyd Evans	
		Tan Y Graig	
		Cefn	
		ST. ASAPH	
		LL17 OHF	
		(in respect of rights of access)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of overhead electricity apparatus)	

		Mona Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Lond	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
09-161		Openreach Limited
cont'd		6 Gracechurch Street
		LONDON
		EC3V 0AT
		(in respect of overhead and underground telecommunication apparatus)
		Debart John Lloyd Gymra
		Robert John Lloyd Evans
		Bryn Hen Groesffordd Marli
		ABERGELE
		Clwyd LL22 9ED
		(in respect of rights of access)
		(in respect of rights of access)
		SP Manweb PLC
		3 Prenton Way
		North Cheshire Trading Estate
		PRENTON
		Merseyside
		CH43 3ET
		(in respect of overhead electricity apparatus)
		Change United Income
		Steven Lloyd Jones
		Bryn Y Pin Mawr
		Groesffordd Marli
		ABERGELE
		LL22 9DT
		(in respect of rights of access)
		Unknown

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-161 cont'd		Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)	
09-162	571 square metres of verge (south of B5381)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Unknown	
	810 square metres of public highway and verge (B5381, Roman Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus) Unknown	
09-165	234 square metres of hedgerow, verge and access splay (south of B5381)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	

	Mona Offshore Wind Limited Development Consent Order			
	BOOK OF REFERENCE - PART 3			
	Counties of Conwy and Denbighshire			
		Names of all those entitled to enjoy easements or other private rights over land		
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,		
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning		
		(Applications: Prescribed Forms and Procedures) Regulations 2009		
09-166	1199 square metres of public highway and verge (B5381)	Dŵr Cymru Cyfyngedig		
		Linea		
		Fortran Road		
		St. Mellons		
		CARDIFF CF3 0LT		
		(in respect of underground water apparatus)		
		Kinmel (H) No.1 Company Limited		
		St. Johns Chambers		
		Love Street		
		CHESTER		
		Cheshire		
		CH1 1QN		
		(in respect of rights reserved as contained in a Conveyance dated 6 February 1981)		
		Kinmel (H) No.2 Company Limited		
		St. Johns Chambers		
		Love Street		
		CHESTER		
		Cheshire		
		CH1 1QN		
		(in respect of rights reserved as contained in a Conveyance dated 6 February 1981)		
		Openreach Limited		
		6 Gracechurch Street		
		LONDON		
		EC3V OAT		
		(in respect of overhead telecommunication apparatus)		
		Zayo Group UK Limited		
		The Relay Building		
		114 Whitechapel High Street		
		E1 7PT /in report of underground telecommunication enparatue)		
		(in respect of underground telecommunication apparatus)		

	Mona Offsh	ore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
	C	counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-167	5273 square metres of public highway and verge (B5381, Glascoed Road, Roman Road and	
	Cae Onnen Road)	County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (in respect of street furniture) Dŵr Cymru Cyfyngedig Linea
		Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)
		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus)
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead and underground electricity apparatus)
		Unknown

Mona Offshore Wind Limited Development Consent Order			
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-167 cont'd		Zayo Group UK Limited The Relay Building 114 Whitechapel High Street London E1 7PT (in respect of underground telecommunication apparatus)	
09-168 585 squ	are metres of access track (off Cae Onnen)	Gareth Hughes Tan Dderwen Glascoed ABERGELE LL22 DDE (in respect of rights of access) Gwen Vaughan Hughes Tan Dderwen Glascoed ABERGELE LL22 9DE (in respect of rights of access) Huw Lloyd Evans Tan Y Graig Cefn ST. ASAPH LL17 0HF (in respect of rights of access) Openreach Limited 6 Graecchurch Street LONDON EC3V 0AT (in respect of overhead and underground telecommunication apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
09-168 cont'd		(Applications: Prescribed Forms and Procedures) Regulations 2009 Robert John Lloyd Evans Bryn Hen Groesffordd Marli ABERGELE Clwyd Ll22 9ED (in respect of rights of access) Steven Lloyd Jones Bryn Y Pin Mawr Groesffordd Marli ABERGELE LL22 9DT (in respect of rights of access) Unknown	
09-169	34835 square metres of agricultural land (north of Cae Onnen Road)	Unknown	
09-170	30 square metres of agricultural land (north of Cae Onnen Road)	Unknown	
	3859 square metres of agricultural land, hedgerow and access track (north of Cae Onnen Road)	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
09-172	1284 square metres of agricultural land (north of Cae Onnen Road)	SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of underground electricity apparatus)	
		Unknown	
09-173	1121 square metres of public highway (Cae Onnen Road)	SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
		Unknown	
00.472			
09-173a	13 square metres of public highway (Cae Onnen Road)	Dŵr Cymru Cyfyngedig	
		Linea	
		Fortran Road	
		St. Mellons CARDIFF	
		(in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-173a		SP Manweb PLC	
cont'd		3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground and overhead electricity apparatus)	
09-173b	31 square metres of public highway (Cae Onnen Road)	Dŵr Cymru Cyfyngedig	
		Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)	
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus)	
		Unknown	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
Counties of Conwy and Denbighshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
	4474 square metres of agricultural land and access track (south of Cae Onnen Road)	SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown	
	38922 square metres of agricultural land, access track and an air shaft (south of Cae Onnen Road) and public footpath (FP 105/5)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	
09-176	1109 square metres of access track (south of Cae Onnen Road) and public footpath (FP 105/5)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-176 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	
09-177	734 square metres of woodland (Coed Carreg-Dafydd)	Unknown	
	24186 square metres of agricultural land (south of Cae Onnen Road) and public footpath (FP 105/5)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	

	Mona Offsb	ore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
	(Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
10-179	64148 square metres of agricultural land, hedgerow and an electricity pylon (south of Plas Hafod) and public footpath (FP 105/5)	(Applications: Prescribed Forms and Procedures) Regulations 2009 Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET
10-180	11123 square metres of agricultural land, hedgerow and a pond (Maes Cefn)	(in respect of overhead and underground electricity apparatus) Unknown (in respect of rights stated in Lease dated 10 February 1932) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus)

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
10-180		SP Manweb PLC	
cont'd		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
10-182	696 square metres of public highway (east of Plas Hafod)	Openreach Limited	
	·····	6 Gracechurch Street	
		LONDON	
		EC3V DAT	
		(in respect of overhead and underground telecommunication apparatus)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of underground electricity apparatus)	
		Unknown	
10-183	16861 square metres of agricultural land and hedgerow (east of Plas Hafod)	Openreach Limited	
1		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of overhead telecommunication apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-184	5284 square metres of agricultural land and hedgerow (east of Plas Hafod)	Edward Durand Hotham	
		Ashcroft Furlong	
		West End	
		Chadlington	
		CHIPPING NORTON	
		OX7 3NJ	
		(in respect of rights granted by Conveyance dated 2 September 1966)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of overhead and underground telecommunication apparatus)	
		Robin Hugh Barlow	
		Carreg Dafydd	
		Groesffordd Marli	
		ABERGELE	
		Conwy LL22 9ED	
		(in respect of rights reserved by Conveyance dated 29 March 1978)	
		(in respect of rights reserved by conveyance dated 29 March 1978)	
		Susan Elizabeth Hotham	
		Plas Newydd	
		Groesffordd Marli	
		ABERGELE	
		Conwy LL22 9DS	
		(in respect of rights granted by Conveyance dated 2 September 1966)	
		(in respect of rights granited by conveyance dated 2 september 1900)	
		Unknown	
		(in respect of rights reserved by Conveyance dated 29 March 1978)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
10-185	49204 square metres of agricultural land and hedgerow (south of Glascoed Road)	Burbo Extension Ltd	
10 105		5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect to rights granted by deed dated 4th March 2017)	
		Richard Thomas Owen Williams	
		Dinam Hall	
		Llangaffo	
		GAERWEN	
		Gwynedd	
		LLGO 6LR	
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		СН43 ЗЕТ	
		(in respect of overhead electricity apparatus)	
		Timothy Michael Bell	
		Cefn Estate Office	
		Cefn	
		ST. ASAPH	
		Clwyd	
		LL17 OEY	
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		· · · · · · · · · · · · · · · · · · ·	
10-186	945 square metres of agricultural land (south of Glascoed Road)	Burbo Extension Ltd	
		5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect to rights granted by deed dated 4th March 2017)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-186		Richard Thomas Owen Williams	
cont'd		Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate	
		PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Timothy Michael Bell Cefn Estate Office	
		Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
10-187	1525 square metres of access track (south of Glascoed Road, B5831)	Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus) Unknown	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
	C	Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-188	29211 square metres of agricultural land and hedgerow (south of Glascoed Road, B5831)	Burbo Extension Ltd	
10-100	and and public footpath (FP 105/6)	5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect to rights granted by deed dated 4th March 2017)	
		Dŵr Cymru Cyfyngedig	
		Linea Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 OLT	
		(in respect of underground water apparatus)	
		Gwynt Y Mor OFTO PLC	
		Q14 Quorum Business Park	
		Benton Lane	
		NEWCASTLE UPON TYNE	
		NE12 8BU	
		(in respect of rights granted by deed dated 6th November 2009)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of overhead and underground telecommunication apparatus)	
		Richard Thomas Owen Williams	
		Dinam Hall	
		Llangaffo GAERWEN	
		GAERWEN Gwynedd	
		LL60 6LR	
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-188 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-189	797 square metres of public highway (south of the National Grid Bodelwyddan substation)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Unknown	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-190	3730 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd S Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CABDIFF CF3 0LT (in respect of underground water apparatus) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Witshire SNS 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NEL2 BBU (in respect of rights granted by deed dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-190		Openreach Limited	
cont'd		6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Richard Thomas Owen Williams	
		Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd	
		LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
	31865 square metres of agricultural land, pond and well (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Dŵr Cymru Cyfyngedig Linea	
		Fortran Road St. Mellons CARDIFF CF3 OLT (in respect of underground water apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-191		Gwynt Y Mor Offshore Wind Farm Limited	
cont'd		Windmill Hill Business Park	
		Whitehill Way	
		SWINDON	
		Wiltshire	
		SN5 6PB	
		(in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC	
		Q14 Quorum Business Park	
		Benton Lane	
		NEWCASTLE UPON TYNE	
		NE12 8BU	
		(in respect of rights granted by deed dated 6 November 2009)	
		National Crid Flashrish, Transmission DIC	
		National Grid Electricity Transmission PLC 1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of easements granted contained in a Lease dated 6 November 2009)	
		(in respect of easements granted contained in a lease dated o November 2009)	
		Richard Thomas Owen Williams	
		Dinam Hall	
		Llangaffo	
		GAERWEN	
		Gwynedd	
		LL60 6LR	
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
		(

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-191 cont'd		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-192	12141 square metres of agricultural land and hedgerow (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wittshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect of easements granted contained in a Lease dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-192 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 OEY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-193	1144 square metres of hedgerow (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order		
BOOK OF REFERENCE - PART 3			
		Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-193		National Grid Electricity Transmission PLC	
cont'd		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of easements granted contained in a Lease dated 6 November 2009)	
		Richard Thomas Owen Williams	
		Dinam Hall	
		Llangaffo	
		GAERWEN	
		Gwynedd	
		LL60 6LR	
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
		(
		Timothy Michael Bell	
		Cefn Estate Office	
		Cefn	
		ST. ASAPH	
		Clwyd	
		LL17 OEY	
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-194	544 square metres of hedgerow (south of the National Grid Bodelwyddan substation)	Unknown	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-195	1242 square metres of private road and hedgerow (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd S Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Craig Dakin Tan Y Bryn Uchaf Cefn LLANELWV LL17 DHG (in respect of rights of access) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Katherine Edwards Tan Y Bryn Uchaf Cefn LLANELWY LL17 OHG (in respect of rights of access) Richard Thomas Owen Williams Dinam Hall LLangaffo GAERWEN Gwynedd LE0 SLR	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	C	counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-195		SP Manweb PLC	
cont'd		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
1 1		(in respect of underground and overhead electricity apparatus)	
		Timothy Michael Bell	
		Cefn Estate Office	
		Cefn	
		ST. ASAPH	
		Clwyd	
		LL17 OEY	
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-195a	13 square metres of private road and hedgerow (south of the National Grid Bodelwyddan	Burbo Extension Ltd	
	substation)	5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect to rights granted by deed dated 4th March 2017)	
		Craig Dakin Teo V Demo Habef	
		Tan Y Bryn Uchaf Cefn	
		LLANELWY	
		LLANELWY LL17 OHG	
		(in respect of rights of access)	
		(in respect of rights of access)	
		Gwynt Y Mor OFTO PLC	
		Q14 Quorum Business Park	
		Benton Lane	
		NEWCASTLE UPON TYNE	
		NE12 8BU	
		(in respect of rights granted by deed dated 6th November 2009)	
II			

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-195a		Katherine Edwards	
cont'd		Tan Y Bryn Uchaf	
		Cefn	
		LLANELWY LL17 0HG	
		(in respect of rights of access)	
		(in respect of rights of access)	
		Richard Thomas Owen Williams	
		Dinam Hall	
		Llangaffo	
		GAERWEN	
		Gwynedd	
		LL60 6LR	
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of underground and overhead electricity apparatus)	
		Timothy Michael Bell	
		Cefn Estate Office	
		Cefn	
		ST. ASAPH	
		Clwyd	
		LL17 OEY	
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-196	238 square metres of private road and verges (south of the National Grid Bodelwyddan	Burbo Extension Ltd	
	substation)	5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect of rights granted by deed dated 4th March 2017)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
11.100		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-196 cont'd		Craig Dakin	
cont a		Tan Y Bryn Uchaf Cefn	
		LLANELWY	
		LLANELWY	
		(in respect of rights of access)	
		(in respect of rights of access)	
		Katherine Edwards	
		Tan Y Bryn Uchaf	
		Cefn	
		LLANELWY	
		LL17 OHG	
		(in respect of rights of access)	
		Richard Thomas Owen Williams	
		Dinam Hall	
		Llangaffo GAERWEN	
		GAERWEN Gwynedd	
		LL60 6LR	
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
		(in respect of the rights of access and sporting rights as contained in a transfer dated 5 may 2002)	
		Timothy Michael Bell	
		Cefn Estate Office	
		Cefn	
		ST. ASAPH	
		Clwyd	
		LL17 OEY	
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-197	180786 square metres of agricultural land, hedgerow, pond, access track and electricity	Burbo Extension Ltd	
	pylon (south of the National Grid Bodelwyddan substation)	5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect to rights granted by deed dated 4th March 2017)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Lond	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-197		Dŵr Cymru Cyfyngedig	
cont'd		Linea	
		Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 OLT	
		(in respect of underground water apparatus)	
		Gwynt Y Mor Offshore Wind Farm Limited	
		Windmill Hill Business Park	
		Whitehill Way	
		SWINDON	
		Wiltshire	
		SN5 6PB	
		(in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC	
		Q14 Quorum Business Park	
		Benton Lane NEWCASTLE UPON TYNE	
		NEWCASTLE OPON TINE	
		(in respect of rights granted by deed dated 6 November 2009)	
		(in respect of rights granted by deed dated o November 2005)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of easements granted contained in a Lease dated 6 November 2009)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of overhead electricity apparatus and underground telecommunication apparatus)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-197 cont'd		Openreach Limited 6 Grazechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 GLR (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL70 EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-197		Vodafone Limited	
cont'd		Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)	
11-197a	1530 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	
		Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (as beneficiary of an Option Agreement dated 3 April 2014)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-197a		National Grid Electricity Transmission PLC	
cont'd		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of easements granted contained in a Lease dated 6 November 2009)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of overhead electricity apparatus and underground telecommunication apparatus)	
		(
		Richard Thomas Owen Williams	
		Dinam Hall	
		Llangaffo	
		GAERWEN	
		Gwynedd	
		LL60 6LR	
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell	
		Cefn Estate Office	
		Cefn	
		ST. ASAPH	
		Clwyd	
		LL17 OEY	
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Vodafone Limited	
		Vodafone House	
		The Connection	
		NEWBURY	
		Berkshire	
		RG14 2FN	
		(in respect of underground telecommunication apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-197b	1082 square metres of private road (south of the National Grid Bodelwyddan substation)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus and underground telecommunication apparatus) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)	
11-197c	6354 square metres of agricultural land, hedgerow, pond, access track and electricity pylon (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 GPB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NEL2 8BU (in respect of rights granted by deed dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-197c cont'd	13315 square metres of agricultural land and access track (south of the National Grid Bodelwyddan substation)	(Applications: Prescribed Forms and Procedures) Regulations 2009 Richard Thomas Owen Williams Dinam Hall Llangaffo GARRWEN Gwynedd LL60 6LR (In respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn Cefn Estate Office Cefn Clwyd LL17 0EY (In respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Burbb Extension Ltd 5 Howick Place LONDON SWIP 1WG (In respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SNS 6PB (In respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Berton Lane NEWCASTLE UPON TYME NE12 28U (In respect of rights granted by deed dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order			
		BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire			
		Names of all those entitled to enjoy easements or other private rights over land		
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,		
Land Plans	Description of Land	suspended or interfered with under Regulation $7(1)(c)$ of the Infrastructure Planning		
		(Applications: Prescribed Forms and Procedures) Regulations 2009		
11-198		Mona Elizabeth Davies		
cont'd		Pentre Mawr Farm		
		Groesffordd Marli		
		ABERGELE		
		Conwy		
		LL22 9DR		
		(in respect of the rights granted by the Deed dated 26 January 1996)		
		National Grid Electricity Transmission PLC		
		1 - 3 Strand		
		London		
		WC2N 5EH		
		(in respect of easements granted contained in a Lease dated 6 November 2009)		
		National Grid Electricity Transmission PLC		
		1 - 3 Strand		
		London		
		WC2N 5EH		
		(in respect of overhead electricity apparatus and underground telecommunication apparatus)		
		Richard Thomas Owen Williams		
		Dinam Hall		
		Llangaffo		
		GAERWEN		
		Gwynedd		
		LL60 6LR		
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)		
		Timothy Michael Bell		
1		Cefn Estate Office		
		Cefn		
		ST. ASAPH		
		Clwyd		
		LL17 OEY		
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)		

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-198 cont'd		Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)	
11-199	5408 square metres of agricultural land and treeline (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd S Howick Place LONDON SW1P 1WG (In respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWIDDON Witshire SNS 6PB (In respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (In respect of rights granted by deed dated 6 November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesfford Marli ABERGELE Conwy LL22 9DR (In respect of the rights granted by the Deed dated 26 January 1996)	

	Mona Offshore Wind Limited Development Consent Order		
1		BOOK OF REFERENCE - PART 3	
	ſ	Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
Land Fians		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-199		National Grid Electricity Transmission PLC	
cont'd		1 - 3 Strand	
cont u		London	
		WC2N 5EH	
		(in respect of easements granted contained in a Lease dated 6 November 2009)	
		Richard Thomas Owen Williams	
		Dinam Hall	
		Llangaffo	
		GAERWEN	
		Gwynedd	
		LL60 6LR	
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		(in respect of the right to access and sporting rights as contained in a transfer dated 5 May 2002)	
		Timothy Michael Bell	
		Cefn Estate Office	
		Cefn	
		ST. ASAPH	
		Clwyd	
		LL17 0EY	
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-200	24600 square metres of agricultural land and electricity pylon (south of the National Grid	Burbo Extension Ltd	
	Bodelwyddan substation)	5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect to rights granted by deed dated 4th March 2017)	
		Diamond Transmission Partners BBE Limited	
		Mid City Place	
		71 High Holborn	
		LONDON	
		WC1V 6BA	
		(in respect of underground electricity apparatus)	

BORK OF EXPERTS Counters of Carwy and Densitytherine Number on Land Plans Names of all those entitled to enjoy essements or other private rights over land (including private rights of avagetion over avaer) which it is proposed shall be exclusibled, usuperator of interfered with under Pequilation 7(3)(4) of the Infrastructure Planning (Application Peacities Forms and Procedures) Requisitions 2009 11-200 Gwyst Y Mor Offshore Wind Far Infladd Windmill Hill against Park in United Windmill Hill against Park in United Windmill Hill against Park in United Windmill Way Windbork 2014 Gwyst Y Mor Offshore Wind Far Infladd Windmill Hill against Park in United Windbork 2015 Gwyst Y Mor Offshore Wind Far Windbork 2016 Gwyst Y Mor Offshore Wind Far Windbork 2017 Graph Y Mor Offshore Wind Far Windbork 2018 Graph Y Mor Offshore Wind Far Windbork 2014 Graph Y Mor Offshore Wind Far Windbork 2015 Graph Y Mor Offshore Wind Far Windbork 2016 Graph Y Mor Offshore Wind Far Windbork 2017 Graph Y Mor Offshore Wind Far Windbork 2018 Graph Y Mor Offshore Wind Far Windbork 2019 Graph Y Mor Offshore Wind Far Windbork 2019 Graph Y Mor Offshore Wind Far Windbork 2010		Mona Of	shore Wind Limited Development Consent Order
Countee of Convy and Debibying Number on Land Plans Description of Land 11:200 Countee of Convy and Debibying protein rights or available extinguished, supperded or interfore with und era regulation 7(L)(c) of the Informature Planning 11:200 Counter Wind Far Interfore with und era regulation 7(L)(c) of the Planning of Control Counter Wind Far Interfore with und era regulation 7(L)(c) of the Planning of Windowill Hill Business Park Counter Wind Far Interfore Wind Far Interfore Wind Far Interfore Windowill Hill Business Park SW 5678 (In respect of rights granted by the lease dated 6 November 2009) Gravit Y Mor Off DPLC CI 4 Quorum Business Park Berron Line NET 2801 Ner register of rights granted by ded dated 6th November 2009) Mora Sitabeth Davies Perter Mawr Farm Gravit Y Mar Off Davies Perter Mawr Farm Gravit Y Mar Off Davies Perter Mawr Farm Gravit Or The UPON TYNE NEI2 2801 Nora Sitabeth Davies Perter Mawr Farm Gravit Y Tarsmission PLC 1-3 Strand London NGC2N 55H National Grid Electricity Transmission PLC 1-3 Strand London 1-3 Strand London NGC2N 55H			
Number of Land Plan: Description of Land Infinite private rights or avagation over weakly which its proposed shall be extinguished, supended or interfered with under Replation 71(1) of the Infrastructure Planning (Applications: Proscibed Forms and Procedures) Resultations 2009 13-200 Cont'd Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Supines Sark Winteehill Way SWNNON SWNNON Winteehill Way SWNNON SWNNON SWNNON Winteehill Way SWNNON Winteehill Way SWNNON SWNNON SWNNON In respect of rights granted by the lease dated 6 November 2009) SWNNON In respect of rights granted by deed dated 6t November 2009) SWNNON In respect of rights granted by the Deed dated 5t November 2009)			
Land Plans Description of Land Supported or interfreed with under Regulation (1)(k) of the Infrastructure Planning:			Names of all those entitled to enjoy easements or other private rights over land
Land Plans Description of Land Supported or interfreed with under Regulation (1)(k) of the Infrastructure Planning:	Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,
Line Lappications, Prescribed Forms and Procedures) Regulations 2009 contd Gwynt Y Mar Offshore Wind Fam: Limited windmill Hill Business Park Windmill Hill Business Park SWNDON Gwynt Y Mar Offshore Bark Gwynt Y Mar Offshore Wind Fam: Limited Gwynt Y Mar OFIO PLC Clique Clique Gwynt Y Mar OFIO PLC Clique Clique Clique Gwynt Y Mar OFIO PLC Clique Clique Clique Michael Benton Lane NEVCASTL UPON TYME NEI 2880 (In respect of rights granted by deel dated 6th November 2009) Mona Elizabeth Davies Perter Marw Groestford Maril AERCELLE Conwy Liz 290R (In respect of rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC National Grid Electricity Transmission PLC 1 - 3 Strand London WCX 51H	Land Plans	Description of Land	
11-200 Gwynet War Griffensen Wind Farm Limited contd Windmill Business Park Windmill Wy SWINDON Willshire SNS 50 B Gin respect of rights granted by the lease dated 6 November 2009) Gwynet Y Mor OFID PLC Q14 Quorum Subiness Park Berloni Lane NEWCASTLE UPON TYNE NEI 28 U (In respect of rights granted by the dated 6th November 2009) Mona Elizabeth Davies Mona Elizabeth Davies Penter Mark Tarm Greesflordd Marii AdetRGLE Convy Liz2 90R In respect of rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand Lindon WCN SH Undan Grid Electricity Transmission PLC 1 - 3 Strand London WCN SH			
Whitehill Way SWINDOH Witshire SN5 6P8 (in respect of rights granted by the lease dated 6 November 2009) (in respect of rights granted by the lease dated 6 November 2009) (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by the Deed dated 26 January 1996) (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand (in respect of lectricity apparatus and telecommunication apparatus) (in respect of lectricity Transmission PLC 1 - 3 Strand (in respect of lectricity Transmission PLC 1 - 3 Strand (in Grespect of Strand) (in respect of lectricity Transmission PLC 1 - 3 Strand) (in Grespect of Strand) (in Grespect	11-200		
SWINDON Within's SN5 GPB (In respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OTP PLC Q14 Quorum Business Park Bertion Lane NEWCASTLE UPON TYNE NET2 88U (In respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Fram Groesffordd Marii ABERGELE Comw Lu22 90R (In respect of field fetriothy Transmission PLC 1 - 3 Strand London WC2N SEH (In respect of Electricity Transmission PLC 1 - 3 Strand London WC2N SEH	cont'd		Windmill Hill Business Park
Wilkhine SNS GP8 (in respect of rights granted by the lease dated 6 November 2009) (in respect of rights granted by the lease dated 6 November 2009) (in respect of rights granted by a lease dated 6 November 2009) (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted by the Deed dated 26th January 1996) (in respect of the rights granted b			Whitehill Way
SNS 6P8 (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC C14 Quorum Business Park. Benton Lane NEWCASTLE UPON TYNE NE1288U (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mayr Farm Greeffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) (in respect of leictricity Transmission PLC 1-3 Strand London WC2N SEH			SWINDON
(in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quoru Lane NEVXCASTLL UPON TYNE NE12 88 U (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marii ABERGELE Convay LL22 90R National Grid Electricity Transmission PLC 1 - 3 Strand London National Grid Electricity Transmission PLC 1 - 3 Strand London National Grid Electricity Transmission PLC 1 - 3 Strand London National Grid Electricity Transmission PLC 1 - 3 Strand London			Wiltshire
Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NEI 288U (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGLE Conwy L122 90R (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH			SN5 6PB
Qui Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NEXCASTLE UPON TYNE NEI2 88U (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesfford Marii ABERGCIE Conwy L12 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH Vin respect of electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			(in respect of rights granted by the lease dated 6 November 2009)
Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NEI2 88U (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesfford Marii ABERGCIE Conwy L122 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH Vin respect of electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			
Qui Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NEXCASTLE UPON TYNE NEI2 88U (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesfford Marii ABERGCIE Conwy L12 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH Vin respect of electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			Guard V Mor OFTO DLC
Benton Lane NEV228BU NE128BU (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy L122 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand Condon WC2N SEH National Grid Electricity Transmission PLC 1 - 3 Strand Condon WC2N SEH			
NEWCASTLE UPON TYNE NE12 88U (in respect of rights granted by deed dated 6th November 2009) (in respect of rights granted by deed dated 6th November 2009) National Bizabeth Davies Pentre Mawr Farm Groesffordd Marii ABERGELE Conwy L122 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect of electricity aparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH			
NE12 88U (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Greesffordd Marli ARERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH National Grid Electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			
(in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy Li22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH National Grid Electricity Transmission PLC 1 - 3 Strand London National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH			
Mona Elizabeth Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			
Pentre Mawr Farm Groesffordd Marii ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) 			
Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (In respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (In respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			Mona Elizabeth Davies
ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			
ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			
LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH National Grid Electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH			
LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH National Grid Electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH			Conwy
National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			
1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			(in respect of the rights granted by the Deed dated 26 January 1996)
1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			
London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			National Grid Electricity Transmission PLC
WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			1 - 3 Strand
(in respect of electricity apparatus and telecommunication apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			London
National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH			
1 - 3 Strand London WC2N 5EH			(in respect of electricity apparatus and telecommunication apparatus)
1 - 3 Strand London WC2N 5EH			
London WC2N 5EH			
WC2N 5EH			
(in respect of easements granted contained in a Lease dated 6 November 2009)			
			(in respect of easements granted contained in a Lease dated 6 November 2009)

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 3			
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
11.000		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-200 cont'd		Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 GLR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone Limited Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NF10 8FZ	

	Mona Offshore Wind Limited Development Consent Order			
	BOOK OF REFERENCE - PART 3			
	Counties of Conwy and Denbighshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
11-201	33659 square metres of agricultural land and hedgerow (west of the National Grid	Centrica PLC		
	Bodelwyddan substation)	Millstream		
		Maidenhead Road		
		WINDSOR		
		Berkshire		
		SL4 5GD		
		(in respect of the rights granted by a Deed of Grant dated 12 November 1992)		
		National Grid Electricity Transmission PLC		
		1 - 3 Strand		
		London		
		WC2N 5EH		
		(in respect of restrictive covenants and the rights granted by the Deed dated 06 November 2009)		
		National Grid Electricity Transmission PLC		
		1 - 3 Strand		
		London		
		WC2N 5EH (in respect of underground electricity apparatus and underground telecommunication apparatus)		
		Shell U.K. Limited		
		Shell Centre		
		York Road		
		SE1 7NA (in respect of the rights granted by a Deed of Grant dated 12 November 1992)		
		(in respect of the rights granted by a beed of Grant dated 12 November 1992)		
		The Executor of the Estate of the Late David Watkin Williams-Wynn BT		
		Forsters LLP		
		22 Baker Street		
		LONDON		
		W1U 3BW		
		(in respect of the rights as stated in the Deed dated 06 November 2009)		

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	0	Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-201 cont'd		Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)	
11-202	17087 square metres of agricultural land and hedgerow (west of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6 November 2009)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Early	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-202		Mona Elizabeth Davies	
cont'd		Pentre Mawr Farm	
		Groesffordd Marli	
		ABERGELE	
		Conwy	
		LL22 9DR	
		(in respect of the rights granted by the Deed dated 26 January 1996)	
		Richard Thomas Owen Williams	
		Dinam Hall	
		Llangaffo	
		GAERWEN	
		Gwynedd	
		LL60 6LR	
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Robert Bryn Davies	
		Pentre Mawr Farm	
		Groesffordd Marli	
		ABERGELE	
		Conwy	
		LL22 9DR	
		(in respect of the rights granted by the Deed dated 26 January 1996)	
		Shell U.K. Limited	
		Shell Centre	
		York Road	
		LONDON	
		SE1 7NA	
		(in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993)	
		(in respect of a right of access to lay pipelines contained in a beed dated 2nd July 1995)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-202 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWVDD NP10 8FZ (in respect of underground gas apparatus)	
11-203	9570 square metres of agricultural land (west of the National Grid Bodelwyddan substation)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of the rights granted by a Deed of Grant dated 12 November 1992)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-203		National Grid Electricity Transmission PLC	
cont'd		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of underground electricity apparatus and underground telecommunication apparatus)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of restrictive covenants and the rights granted by the Deed dated 06 November 2009)	
		Shell U.K. Limited	
		Shell Centre	
		York Road	
		LONDON SE1 7NA	
		(in respect of the rights granted by a Deed of Grant dated 12 November 1992)	
		(in respect of the rights granied by a beed of drant dated 12 November 1992)	
		The Executor of the Estate of the Late David Watkin Williams-Wynn BT	
		Forsters LLP	
		22 Baker Street	
		LONDON	
		W1U 3BW (in respect of the rights as stated in the Deed dated 06 November 2009)	
		(in respect of the rights as stated in the beed dated of November 2009)	
		Vodafone Limited	
		Vodafone House	
		The Connection	
		NEWBURY Berkshire	
		Berksnire RG14 2FN	
		(in respect of underground telecommunication apparatus)	
I			

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-203 cont'd 11-204	3021 square metres of agricultural land (west of the National Grid Bodelwyddan substation)	(Applications: Prescribed Forms and Procedures) Regulations 2009 Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of a Deed dated 10 June 2011) Wales & West House Spooner Close Coedkernew CASNEWYDD Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus) Burbo Extension Ltd S Howick Place LONDON SWIP 1WG	
		(in respect to rights granted by deed dated 4th March 2017) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (in respect of underground electricity apparatus)	

	Mona Off	shore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	- · · · ·	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
11-204		Gwynt Y Mor Offshore Wind Farm Limited
cont'd		Windmill Hill Business Park
		Whitehill Way
		SWINDON
		Wiltshire
		SN5 6PB
		(in respect of rights granted by the lease dated 6 November 2009)
		Gwynt Y Mor OFTO PLC
		Q14 Quorum Business Park
		Benton Lane
		NEWCASTLE UPON TYNE
		NEWCASTLE OPON TIME
		(in respect of rights granted by deed dated 6th November 2009)
		(in respect of rights granted by deed dated of november 2009)
		Mona Elizabeth Davies
		Pentre Mawr Farm
		Groesffordd Marli
		ABERGELE
		Conwy
		LL22 9DR
		(in respect of the rights granted by the Deed dated 26 January 1996)
		(in respect of the rights granted by the beed dated 20 January 1990)
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect of electricity apparatus and telecommunication apparatus)
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect of easements granted contained in a Lease dated 6 November 2009)

	Мс	ona Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
11-204 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT
		(in respect of proposed underground telecommunication apparatus)
		Richard Thomas Owen Williams Dinam Hall
		Llangaffo GAERWEN Gwynedd
		LL60 6LR
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
		Shell U.K. Limited Shell Centre
		York Road
		LONDON
		SE1 7NA (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993)
		Timothy Michael Bell Cefn Estate Office
		Cefn
		ST. ASAPH
		Clwyd
		LL17 OEY
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
		Vodafone Limited Vodafone House
		The Connection
		NEWBURY
		Berkshire
		RG14 2FN
		(in respect of underground telecommunication apparatus)

	Mona Offsh	nore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
	(Counties of Conwy and Denbighshire
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-204 cont'd		Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew CASNEWYDD NP10 8FZ (in respect of underground gas apparatus)
11-205	12 square metres of agricultural land (east of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wittshire SN5 6P8 (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NEUCASTLE UPON TYNE NEUCASTLE UPON TYNE NEUCASTLE UPON TYNE NEUCASTLE UPON TYNE NEI2 8BU (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Greesffordd Marli ABERGELE Conwy L122 9DR (in respect of the rights granted by the Deed dated 26 January 1996)

	Mona Offs	hore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-205 cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd Ll60 GLR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn ST. ASAPH Clwyd Ll17 OEY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
	252 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON
		WC1V 6BA (in respect of underground electricity apparatus)

	Mona C	Offshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation $7(1)(c)$ of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
11-206		Gwynt Y Mor Offshore Wind Farm Limited
cont'd		Windmill Hill Business Park
		Whitehill Way
		SWINDON
		Wiltshire
		SN5 6PB
		(in respect of rights granted by the lease dated 6 November 2009)
		Gwynt Y Mor OFTO PLC
		Q14 Quorum Business Park
		Benton Lane
		NEWCASTLE UPON TYNE
		NE12 8BU
		(in respect of rights granted by deed dated 6th November 2009)
		Mona Elizabeth Davies
		Pentre Mawr Farm
		Groesffordd Marli
		ABERGELE
		Conwy
		LL22 9DR
		(in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams
		Dinam Hall
		Llangaffo
		GAERWEN
		Gwynedd
		LL60 6LR
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
		(in respect of the rights of decess and sporting rights as contained in a transier dated 5 way 2002)

	Mona Offs	shore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-206 cont'd		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-207	1184 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SNS 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesfford Marlii ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Conwy and Denbighshire Names of all those entit	
Counties of Conwy and Denbighshire	
Numes of all chose circle	tled to enjoy easements or other private rights over land
Number on (including private rights of nav	vigation over water) which it is proposed shall be extinguished,
	with under Regulation 7(1)(c) of the Infrastructure Planning
	escribed Forms and Procedures) Regulations 2009
11-207 National Grid Electricity Transmission PLC	
cont'd 1 - 3 Strand	
London	
WC2N 5EH	
(in respect of underground telecommunication appar	ratus)
National Grid Electricity Transmission PLC	
1 - 3 Strand	
London	
WC2N 5EH	
(in respect of easements granted contained in a Lease	e dated 6 November 2009)
Richard Thomas Owen Williams	
Dinam Hall	
Llangaffo	
GAERWEN	
Gwynedd	
LL60 6LR	
(in respect of the rights of access and sporting rights a	as contained in a Transfer dated 3 May 2002)
Timothy Michael Bell	
Cefn Estate Office	
Cefn	
ST. ASAPH	
Clwyd	
LL17 OEY	
(in respect of the rights of access and sporting rights a	as contained in a Transfer dated 3 May 2002)
Vodafone Limited	
Vodafone House	
The Connection	
NEWBURY	
Berkshire	
RG14 2FN	
(in respect of underground telecommunication appar	ratus)

	Mona Offsh	nore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
	(Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
11-208	7985 square metres of woodland (Hendy Gorse, south of the National Grid Bodelwyddan	(Applications: Prescribed Forms and Procedures) Regulations 2009 Burbo Extension Ltd
	substation)	5 Howick Place LONDON SW1P 1WG (in respect of rights to have and use cables contained in a Deed dated 4 May 2017) Shell U.K. Limited Shell Centre
		York Road LONDON SE1 7NA (in respect of rights contained in a Deed dated 10 June 1977)
	1717 square metres of woodland (Hendy Gorse, south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights to have and use cables contained in a Deed dated 4 May 2017) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights contained in a Deed dated 10 June 1977)
11-210	1498 square metres of woodland (south of Glascoed Road)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights to have and use cables contained in a Deed dated 4 May 2017)

	Mona Of	fshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-210 cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of underground telecommunication apparatus) Shell U.K. Limited Shell Centre York Road LONDON SE1 7NA (in respect of rights contained in a Deed dated 10 June 1977) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)
11-211	2013 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009)

	Mona C	ffshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
11-211		Gwynt Y Mor OFTO PLC
cont'd		Q14 Quorum Business Park
		Benton Lane
		NEWCASTLE UPON TYNE
		NE12 8BU
		(in respect of rights granted by deed dated 6 November 2009)
		Mona Elizabeth Davies
		Pentre Mawr Farm
		Groesffordd Marli
		ABERGELE
		Conwy
		LL22 9DR
		(in respect of the rights granted by the Deed dated 26 January 1996)
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect of easements granted contained in a Lease dated 6 November 2009)
		, , , , , , , , , , , , , , , , , , ,
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect of overhead electricity apparatus and underground telecommunication apparatus)
		Richard Thomas Owen Williams
		Dinam Hall
		Llangaffo
		GAERWEN
		Gwynedd
		LL60 6LR
		(in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offsh	nore Wind Limited Development Consent Order
	BOOK OF REFERENCE - PART 3	
	(Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-211 cont'd	2656 square metres of agricultural land (south of the National Grid Bodelwyddan substation)	Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the right to access and sporting rights as contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus) Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor OFTO PLC
		Q1 ⁴ Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Ol	ffshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-212 cont'd		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-213	15147 square metres of agricultural land and pond (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL70 EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-214		Burbo Extension Ltd S Howick Place LONDON SWJP JWG (in respect to rights granted by deed dated 4th March 2017) Craig Dakin Tan Y Bryn Uchaf Cefn LLANELWY LLAY CWY LLAY CWY LLAY CWY LLAY CWY LLAY CWY LLAY DHG (in respect of rights of access) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NEI2 8BU (in respect of rights granted by deed dated 6th November 2009) Katherine Edwards Tan Y Bryn Uchaf Cefn LLANELWY LLAY CWY LLAY CWY	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on Land Plans	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-214		Richard Thomas Owen Williams	
cont'd		Dinam Hall	
		Llangaffo	
		GAERWEN	
		Gwynedd	
		LL60 6LR	
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell	
		Cefn Estate Office	
		Cefn	
		ST. ASAPH	
		Clwyd	
		LL17 OEY	
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
11-215	38142 square metres of agricultural land and pond (south of the National Grid	Burbo Extension Ltd	
	Bodelwyddan substation)	5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect to rights granted by deed dated 4th March 2017)	
		Gwynt Y Mor OFTO PLC	
		Q14 Quorum Business Park	
		Benton Lane	
		NEWCASTLE UPON TYNE	
		NE12 8BU	
		(in respect of rights granted by deed dated 6th November 2009)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of overhead electricity apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-215 cont'd	9552 square metres of agricultural land and hedgerow (south of the National Grid Bodelwyddan substation) and public bridleway (BR 208/3)	Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gewnedd LL50 GLR (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EV (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002) Burbo Extension Ltd S Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor OFTO PLC QL4 Quorum Business Park Benton Lane NEWCASTE UPON TYNE NEI2 8BU (in respect of rights granted by deed dated 6th November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LLangaffo GAERWEN Gwynedd LLangaffo GAERWEN Gwynedd LLangaffo GAERWEN Gwynedd LLANGEN GWYNE (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	

	Mona Offshore Wind Limited Development Consent Order	
	BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-216		Timothy Michael Bell
cont'd		Cefn State Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
11-217	822 square metres of agricultural land (south of the National Grid Bodelwyddan substation	Rurho Extension Itd
11 217		5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)
		National Grid Electricity Transmission PLC 1 - 3 Strand
		London
		WC2N 5EH
		(in respect of overhead electricity apparatus)
		Richard Thomas Owen Williams Dinam Hall
		Llangaffo
		GAERWEN
		Gwynedd
		LL60 6LR
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
		Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order	
	BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation $7(1)(c)$ of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
11-218	16202 square metres of agricultural land, hedgerow and pond (south of the National Grid	Mona Elizabeth Davies
	Bodelwyddan substation)	Pentre Mawr Farm
		Groesffordd Marli
		ABERGELE
		Conwy
		LL22 9DR
		(in respect of right of drainage and access for maintenance)
		Richard Thomas Owen Williams
		Dinam Hall
		Llangaffo
		GAERWEN
		Gwynedd
		LL60 6LR
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
		Robert Bryn Davies
		Pentre Mawr Farm
		Groesffordd Marli
		ABERGELE
		Conwy
		LL22 9DR
		(in respect of right of drainage and access for maintenance)
		Timothy Michael Bell
		Cefn Estate Office
		Cefn
		ST. ASAPH
		Clwyd
		LL17 OEY
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order	
	BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
-		(Applications: Prescribed Forms and Procedures) Regulations 2009
11-218a	14130 square metres of agricultural land, hedgerow and pond (south of the National Grid	Mona Elizabeth Davies
	Bodelwyddan substation)	Pentre Mawr Farm
		Groesffordd Marli
		ABERGELE
		Conwy
		LL22 9DR
		(in respect of right of drainage and access for maintenance)
		Richard Thomas Owen Williams
		Dinam Hall
		Llangaffo
		GAERWEN
		Gwynedd
		LL60 6LR
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)
		Robert Bryn Davies
		Pentre Mawr Farm
		Groesffordd Marli
		ABERGELE
		Conwy
		LL22 9DR
		(in respect of right of drainage and access for maintenance)
		· ····································
		Timothy Michael Bell
		Cefn Estate Office
		Cefn
		ST. ASAPH
		Clwyd LL17 OEY
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	C	ounties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Lond	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-219	5663 square metres of agricultural land and hedgerow (south of the Gwynt y Mor Offshore	Burbo Extension Ltd	
	Wind Farm Substation)	5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect to rights granted by deed dated 4th March 2017)	
		Diamond Transmission Partners BBE Limited	
		Mid City Place	
		71 High Holborn	
		LONDON	
		WC1V 6BA	
		(in respect of underground electricity apparatus)	
		Gwynt Y Mor Offshore Wind Farm Limited	
		Windmill Hill Business Park	
		Whitehill Way	
		SWINDON	
		Wiltshire	
		SN5 6PB	
		(in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC	
		Q14 Quorum Business Park	
		Benton Lane	
		NEWCASTLE UPON TYNE	
		NE12 8BU	
		(in respect of rights granted by deed dated 6th November 2009)	
		Mona Elizabeth Davies	
		Pentre Mawr Farm	
		Groesffordd Marli	
		ABERGELE	
		Conwy	
		LL22 9DR	
		(in respect of the rights granted by the Deed dated 26 January 1996)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Lond	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-219		National Grid Electricity Transmission PLC	
cont'd		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of underground telecommunication apparatus)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of easements granted contained in a Lease dated 6 November 2009)	
		Richard Thomas Owen Williams	
		Dinam Hall	
		Llangaffo	
		GAERWEN	
		Gwynedd	
		LL60 6LR	
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Timothy Michael Bell	
		Cefn Estate Office	
		Cefn	
		ST. ASAPH	
		Clwyd	
		LL17 OEY	
		(in respect of the rights of access and sporting rights as contained in a Transfer dated 3 May 2002)	
		Vodafone Limited	
		Vodafone House	
		The Connection	
		NEWBURY	
		Berkshire	
		RG14 2FN	
		(in respect of underground telecommunication apparatus)	
l		•	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation $7(1)(c)$ of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-220	3075 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm	Burbo Extension Ltd	
	Substation)	5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect of the rights as stated in the lease dated 30 October 2014)	
		ESP Electricity Limited	
		1st Floor	
		Bluebird House	
		Mole Business Park	
		Leatherhead	
		Surrey	
		KT22 7BA	
		(in respect of the rights as stated in the transfer dated 10 June 2016)	
		Gwynt Y Mor Offshore Wind Farm Limited	
		Windmill Hill Business Park	
		Whitehill Way	
		SWINDON	
		Wiltshire	
		SN5 6PB	
		(in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC	
1		Q14 Quorum Business Park	
		Benton Lane	
		NEWCASTLE UPON TYNE	
		NE12 8BU	
		(in respect of underground electricity apparatus)	
1		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect to the right to access as stated in the deed dated 6 November 2009)	
l			

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-220		Openreach Limited	
cont'd		Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of underground electricity apparatus) Wales and West Utilities Limited Wales & West House Spooner Close Coedkernew	
11-221	2146 square metres of private road and verges (east of the Gwynt y Mor Offshore Wind	CASNEWYDD NP10 8FZ (in respect of underground gas apparatus) Burbo Extension Ltd	
11-221	Farm Substation)	S Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons CARDIFF CF3 0LT (in respect of underground water apparatus and underground abandoned water apparatus)	

	Mona Offshore Wind Limited Development Consent Order	
		BOOK OF REFERENCE - PART 3
	Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
11-221		ESP Electricity Limited
cont'd		1st Floor
		Bluebird House Mole Business Park
		Leatherhead
		Surrey
		KT22 7BA
		(in respect of the rights as stated in the transfer dated 10 June 2016)
		County V Mary Offsham Wind Frank United
		Gwynt Y Mor Offshore Wind Farm Limited
		Windmill Hill Business Park Whitehill Way
		SWINDON
		Wiltshire
		SN5 6PB
		(in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February
		2010 and in respect of the rights as stated in the lease dated 6 November 2009)
		Gwynt Y Mor OFTO PLC
		Q14 Quorum Business Park
		Benton Lane
		NEWCASTLE UPON TYNE
		NE12 8BU
		(in respect of underground electricity apparatus)
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect to the right to access as stated in the deed dated 6 November 2009)
		Openreach Limited
		6 Gracechurch Street
		EC3V OAT
		(in respect of underground telecommunication apparatus)

BODK OF REERVICE - PART 3 Counties of Conwy and Dehighbrine Number on Land Plans Names of all those entitled to enjoy easements or other private rights own objective registric of navigation over water lymbs. The spring of the private rights of navigation over water lymbs. The spring of the spr	
Number on Land Plans Names of all those entitled to enjoy easements or other private rights or analyzation our value yikich is greated be exting usupended or interfered with under Regulations 2009 11-221 ont'd SP Manweb PLC 3 Pentor Mway Netron Way 11-221a Superative metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation) Burb Deticition agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation) 11-221a 127 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation) Burb Deticition the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebite House Mole Business Park Leatherhead Surrey Kit22 7GA (in respect of the rights as stated in the transfer dated 10 June 2016)	
Number on Land Plans Description of Land (Including private rights of navigation over water) which it is proposed shall be extinguing suspended or interfered with under Regulation 7(1)(c) of the infrastructure Planni (Applications: Prescribed Forms and Procedures) Regulations 2009 11-221 cont'd SP Manweb PLC SP Manweb PLC 3 Prention Way North Cheshire Trading Estate PRENTON PRENTON Merseyside CH43 3ET (In respect of underground electricity apparatus) Introduction of the Gwynt y Mor Offshore Wind Farm 11-221 Substation) Burbo Extension Ltd Substation) Substation) Substation 11-221 Substation) Substation 127 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Burbo Extension Ltd Substation) Substation Substation Substation) Earle Prescription of Land Superior Content of Superior C	
Land Plans Suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planni (Applications: Prescribed Forms and Procedures) Regulations 2009 11-221 SP Manweb PLC cont'd 3 Prenton Way 11-221a IZ7 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation) Burbo Extension Ltd S How Part Show Part Substation) Burbo Extension Ltd S How Part Show Part Substation) Sterification for the Gwynt y Mor Offshore Wind Farm Burbo Extension Ltd S How Part Substation) Sterification for the Gwynt y Mor Offshore Wind Farm Burbo Extension Ltd S How Part Substation) Sterest of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Burboit Mark Survey Kirz 2 ZBA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Gwynt Y Mor Offshore Wind Farm Limited Whitehill Way Whitehill Way	
11-221 Cont d SP Manweb PLC 12-221 SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3 ET (In respect of underground electricity apparatus) In respect of underground electricity apparatus) 11-221a 127 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Burbo Extension Ltd Substation) Stubstation) Burbo Extension Ltd Substation) Stubstation Stude as the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Burbo Sum Burbo Rave Mole Business Park Leatherhead Surrey Ki72 ZBA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Surrey Ki72 ZBA Min Business Park Leatherhead Surrey Ki72 ZBA (in respect of the rights as stated in the transfer dated 10 June 2016)	
11-221 SP Manweb PLC cont'd 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (In respect of underground electricity apparatus) In respect of underground electricity apparatus) 11-221a 127 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Burbo Extension Ltd Substation) Burbo Extension Ltd Substation) Extension Ltd Substation) Substation Extension Ltd Substation) Substation Substation) Extension Ltd Substation Substation) Extension Ltd Substation Substation) Substation Substation Substation) Substation Substation Substation Substation Substation Substatio	
Image: North Cheshire Trading Estate PRENTON North Cheshire Trading Estate PRENTON Nerseyside CH43 3ET (in respect of underground electricity apparatus) 11-221a 127 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation) Burbo Extension Ltd Studition) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Lattherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park	S Regulations 2005
PRENTON Merseyside CH-33 SET (In respect of underground electricity apparatus) 11-221a 127 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Burbo Extension Ltd Substation) Substation) Substation ESP Electricity Compared to the lease dated 30 October 2014) Substation ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Least of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Wintehill Business Park Winderfill Business Park Wintehill Business Park Wintehill Business Park	
Merseyside CH3 3ET (in respect of underground electricity apparatus) 11-221a 127 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Burbo Extension Ltd Substation) 5 Howick Place LONDON SWIP 1WG (in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park	
11-221a 127 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Burbo Extension Ltd Substation) 5 Howick Place LONDON SWIP 1WG (in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT2 Z ZBA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Windmill Hill Business Park Windmill Hill Business Park	
Image: Instant and the image: Imag	
11-221a 127 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation) Burbo Extension Ltd 11-221a Substation) 5 Howick Place LONDON SW1P WG (in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way	
Substation) Substation) Substation) Substation) Substation) Substation Substa	
Substation) Substation) Substation) Substation) Substation) Substation Substa	
Substation) Substation) Substation) Substation) Substation) Substation Substa	
LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way	
(in respect of the rights as stated in the lease dated 30 October 2014) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way	
ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way	
ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way	
1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Windmill Hill Business Park Windmill Hill Business Park Windmill Hill Business Park	
1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Windmill Hill Business Park Whitehill Way	
Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way	
Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way	
Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way	
KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way	
(in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way	
Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way	
Windmill Hill Business Park Whitehill Way	
Windmill Hill Business Park Whitehill Way	
Whitehill Way	
SWINDON	
Wiltshire	
SN5 6PB	
(in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the	of the rights as stated in the deed dated 16 February
2010 and in respect of the rights as stated in the lease dated 6 November 2009)	

	Mona Off	shore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-221a		National Grid Electricity Transmission PLC
cont'd		1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON
		EC3V 0AT (in respect of underground telecommunication apparatus)
11-222	174 square metres of private road (south of the Gwynt y Mor Offshore Wind Farm Substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014)
		ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016)
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February 2010 and in respect of the rights as stated in the lease dated 6 November 2009)

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-222 cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009)	
11-223	979 square metres of private road and verges (south of the Gwynt y Mor Offshore Wind Farm Substation)	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS (in respect of rights of access) Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 GPB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NES2 88U (in respect of rights granted by deed dated 6th November 2009)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
	(Counties of Conwy and Denbighshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-223		Mona Elizabeth Davies	
cont'd		Pentre Mawr Farm	
		Groesffordd Marli	
		ABERGELE	
		Conwy	
		LL22 9DR	
		(in respect of the rights granted by the Deed dated 26 January 1996)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of easements granted contained in a Lease dated 6 November 2009)	
		(in respect of easements granted contained in a lease dated o november 2005)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of electricity apparatus)	
		Richard Thomas Owen Williams	
		Dinam Hall	
		Llangaffo	
		GAERWEN	
		Gwynedd	
		LL60 6LR	
		(in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)	
		Robert Bryn Davies	
		Pentre Mawr Farm	
		Groesffordd Marli	
		ABERGELE	
		Conwy	
		LL22 9DR	
		(in respect of the rights granted by the Deed dated 26 January 1996)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-223 cont'd		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights granted by lease dated 16 September 2011) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)	
11-224	29 square metres of private road and verges (south of the Gwynt y Mor Offshore Wind Farm Substation)	Arthur Elwy Morris Owen Bodysgaw Isaf Llannefydd DENBIGH Denbighshire LL16 5DS (in respect of rights of access) Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-224		Gwynt Y Mor Offshore Wind Farm Limited	
cont'd		Windmill Hill Business Park	
		Whitehill Way	
		SWINDON	
		Wiltshire	
		SN5 6PB	
		(in respect of rights granted by the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC	
		Q14 Quorum Business Park	
		Benton Lane	
		NEWCASTLE UPON TYNE	
		NET 8BU	
		(in respect of rights granted by deed dated 6th November 2009)	
		(in respect of rights granted by deed dated of in November 2005)	
		Mona Elizabeth Davies	
		Pentre Mawr Farm	
		Groesffordd Marli	
		ABERGELE	
		Conwy	
		LL22 9DR	
		(in respect of the rights granted by the Deed dated 26 January 1996)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of easements granted contained in a Lease dated 6 November 2009)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of electricity apparatus)	

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-224 cont'd		Applications: Prescribed Forms and Procedures) Regulations 2009 Richard Thomas Owen Williams Dinam Hall Langaffo GAERWEN Gwynedd L60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002) Robert Bryn Davies Pentre Mawr Farm Groesffordd Marii ABERGELE Conwy L122 9DR (in respect of the rights granted by the Deed dated 26 January 1996) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of rights granted by lease dated 16 September 2011) Timothy Michael Bell Cefn Estate Office Cefn	
		ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		Counties of Conwy and Denbighshire	
Number o Land Plar	n Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-225	581 square metres of private road (south of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDOM SWUP 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SNS 6P8 (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesford Marli ABERGLE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3 Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-225 cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009) Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002) Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)	

	Mona Offshore Wind Limited Development Consent Order	
		BOOK OF REFERENCE - PART 3
	(Counties of Conwy and Denbighshire
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-225 cont'd		Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)
11-226	22 square metres of agricultural land (east of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Witshire SN5 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of rights granted by deed dated 6th November 2009) Mona Elizabeth Davies Pentre Mawr Farm Groesffrodd Marlii ABERGELE Conwy L122 9DR (in respect of the rights granted by the Deed dated 26 January 1996)

	Mona O	ffshore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
		Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
11-226		National Grid Electricity Transmission PLC
cont'd		1 - 3 Strand
		London
		WC2N 5EH
		(in respect of underground telecommunication apparatus)
		National Grid Electricity Transmission PLC
		1 - 3 Strand
		London
		WC2N 5EH
		(in respect of easements granted contained in a Lease dated 6 November 2009)
		Richard Thomas Owen Williams
		Dinam Hall
		Llangaffo
		GAEWEN
		Gwynedd
		LLGO 6LR
		(in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)
		Robert Bryn Davies
		Pentre Mawr Farm
		Groesffordd Marli
		ABERGELE
		Conwy
		LL22 9DR
		(in respect of the rights granted by the Deed dated 26 January 1996)
		Timothy Michael Bell
		Cefn Estate Office
		Cefn
		ST. ASAPH
		Clwyd
		LL17 OEY
		(in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)
		-

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
	C	Counties of Conwy and Denbighshire	
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-226 cont'd		Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)	
11-227	360 square metres of agricultural land (east of the National Grid Bodelwyddan substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect to rights granted by deed dated 4th March 2017) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SNS 6PB (in respect of rights granted by the lease dated 6 November 2009) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE122 BBU (in respect of rights granted by deed dated 6th November 2009)	

Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3	
	Counties of Conwy and Denbighshire	
Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
	Mona Elizabeth Davies	
	Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy	
	LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996)	
	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus and telecommunication apparatus)	
	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of easements granted contained in a Lease dated 6 November 2009)	
	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	
	Richard Thomas Owen Williams Dinam Hall Llangaffo GAERWEN Gwynedd LL60 6LR (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002)	
	Description of Land	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	(Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-227 cont'd		Robert Bryn Davies Pentre Mawr Farm Groesffordd Marli ABERGELE Conwy LL22 9DR (in respect of the rights granted by the Deed dated 26 January 1996) Shell U.K. Limited Shell Centre York Road LONDON SEI 7NA (in respect of a right of access to lay pipelines contained in a Deed dated 2nd July 1993) Timothy Michael Bell Cefn Estate Office Cefn ST. ASAPH Clwyd LL17 0EY (in respect of rights of access to right of sporting rights contained in a Transfer dated 3 May 2002) Vodafone Limited Vodafone Limited Vodafone Limited Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of underground telecommunication apparatus)	
11-228	1337 square metres of agricultural land (east of the Gwynt y Mor Offshore Wind Farm Substation)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the rights as stated in the lease dated 30 October 2014)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-228		ESP Electricity Limited	
cont'd		1st Floor	
		Bluebird House Mole Business Park	
		Leatherhead	
		Surrey	
		KT22 7BA	
		(in respect of the rights as stated in the transfer dated 10 June 2016)	
		County Vi Mara Offich and Milard France Lingths d	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park	
		Whitehill Way	
		SWINDON	
		Wiltshire	
		SN5 6PB	
		(in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February	
		2010 and in respect of the rights as stated in the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC	
		Q14 Quorum Business Park	
		Benton Lane	
		NEWCASTLE UPON TYNE	
		NE12 8BU	
		(in respect of underground electricity apparatus)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect to the right to access as stated in the deed dated 6 November 2009)	
		· ····································	
		Openreach Limited	
		6 Gracechurch Street	
		EC3V OAT	
		(in respect of underground telecommunication apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
11-228		(Applications: Prescribed Forms and Procedures) Regulations 2009 SP Manweb PLC	
cont'd		3 Prenton Way	
cont u		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of underground electricity apparatus)	
11-229	6518 square metres of agricultural land (south of Glascoed road, B5381)	Burbo Extension Ltd	
		5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect of the rights as stated in the lease dated 30 October 2014)	
		Dŵr Cymru Cyfyngedig	
		Linea	
		Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 OLT	
		(in respect of underground water apparatus and underground abandoned water apparatus)	
		ESP Electricity Limited	
		1st Floor	
		Bluebird House	
		Mole Business Park	
		Leatherhead	
		Surrey	
		кт22 7ВА	
		(in respect of the rights as stated in the transfer dated 10 June 2016)	

	Mona Offshore Wind Limited Development Consent Order			
	BOOK OF REFERENCE - PART 3			
	Counties of Conwy and Denbighshire			
		Names of all those entitled to enjoy easements or other private rights over land		
Number on	Description of land	(including private rights of navigation over water) which it is proposed shall be extinguished,		
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning		
		(Applications: Prescribed Forms and Procedures) Regulations 2009		
11-229		Gwynt Y Mor Offshore Wind Farm Limited		
cont'd		Windmill Hill Business Park		
		Whitehill Way		
		SWINDON		
		Wiltshire		
		SN5 6PB		
		(in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February		
		2010 and in respect of the rights as stated in the lease dated 6 November 2009)		
		Gwynt Y Mor OFTO PLC		
		Q14 Quorum Business Park		
		Benton Lane		
		NEWCASTLE UPON TYNE		
		NE12 8BU		
		(in respect of underground electricity apparatus)		
		National Grid Electricity Transmission PLC		
		1 - 3 Strand		
		London		
		WC2N 5EH		
		(in respect to the right to access as stated in the deed dated 6 November 2009)		
		Openreach Limited		
		6 Gracechurch Street		
		LONDON		
		EC3V DAT		
		(in respect of underground telecommunication apparatus)		
		SP Manweb PLC		
		3 Prenton Way		
		North Cheshire Trading Estate		
		PRENTON		
		Merseyside		
		CH43 3ET		
		(in respect of underground and overhead electricity apparatus)		

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
11.220		(Applications: Prescribed Forms and Procedures) Regulations 2009 Wales and West Utilities Limited	
11-229 cont'd		Wales & West Utilities Limited Wales & West House	
cont u		Spooner Close	
		Coedkernew	
		CASNEWYDD	
		NP10 8FZ	
		(in respect of underground gas apparatus)	
11-229a	313 square metres agricultural land (south of Glascoed road, B5381)	Burbo Extension Ltd	
		5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect of the rights as stated in the lease dated 30 October 2014)	
		Dŵr Cymru Cyfyngedig	
		Linea	
		Fortran Road	
		St. Mellons	
		CARDIFF	
		CF3 0LT	
		(in respect of underground water apparatus and underground abandoned water apparatus)	
		ESP Electricity Limited	
		1st Floor	
		Bluebird House	
		Mole Business Park	
		Leatherhead	
		Surrey	
		KT22 7BA	
		(in respect of the rights as stated in the transfer dated 10 June 2016)	
		·	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
Counties of Conwy and Denbighshire			
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-229a		Gwynt Y Mor Offshore Wind Farm Limited	
cont'd		Windmill Hill Business Park	
		Whitehill Way	
		SWINDON	
		Wiltshire	
		SN5 6PB	
		(in respect of the rights as stated in the lease dated 30 October 2014 and in respect of the rights as stated in the deed dated 16 February	
		2010 and in respect of the rights as stated in the lease dated 6 November 2009)	
		Gwynt Y Mor OFTO PLC	
		Q14 Quorum Business Park	
		Benton Lane	
		NEWCASTLE UPON TYNE	
		NE12 8BU	
		(in respect of underground electricity apparatus)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect to the right to access as stated in the deed dated 6 November 2009)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of underground and overhead electricity apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-230	44 square metres of agricultural land (south of Glascoed road, B5381) and public bridleway (BR 208/32)		
11-230a	38 square metres of agricultural land (south of Glascoed road, B5381) and public bridleway (BR 208/32)	Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights to install and use electric lines as stated in the Deed of Grant dated 6 November 2009) Unknown	
	33 square metres of private road (off Glascoed road, B5381) and public bridleway (BR 208/32)	Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of rights contained within a Lease dated 30 October 2014) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights in relation to the installation and use of electric lines within a Deed of Grant dated 6 November 2009)	

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Conwy and Denbighshire Number on Land Plans Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguish suspended the Regulation 7(1)(c) of the Informatructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 11-232 1147 square metres of private road (south of Glascoed road, B5381) Burbo Extension Ltd 5 Howick Place LONDON SW/1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as state October 2014) Dir Cyrrun Cyfyngedig Linea Fortran Road St. Mellons Dir Cyrrun Cyfyngedig Linea Fortran Road St. Mellons	
Counties of Conwy and Denbighshire Number on Land Plans Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguish suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 11-231 cont'd Unknown 11-232 1147 square metres of private road (south of Glascoed road, B5381) Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as state October 2014) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons Dŵr Cymru Cyfyngedig Linea	
Number on Land Plans Description of Land (including private rights of navigation over water) which it is proposed shall be extinguish suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning. (Applications: Prescribed Forms and Procedures) Regulations 2009 11-231 cont'd Unknown 11-232 1147 square metres of private road (south of Glascoed road, B5381) Burbo Extension Ltd 5 Howick Place LONDON SW1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated October 2014) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons Dŵr Cymru Cyfyngedig Linea	
cont'd I11-232 1147 square metres of private road (south of Glascoed road, B5381) Burbo Extension Ltd S Howick Place LONDON LONDON SW1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons	
5 Howick Place LONDON SW1P 1WG (in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as state October 2014) Dŵr Cymru Cyfyngedig Linea Fortran Road St. Mellons	
GF3 0LT (in respect of underground abandoned water apparatus) ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of the rights as stated in the transfer dated 10 June 2016) Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Witshire SNS 6PB (in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 16 January 2010 and janua	

	Mona Offsh	ore Wind Limited Development Consent Order
		BOOK OF REFERENCE - PART 3
	(Counties of Conwy and Denbighshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
Lallu Platis		(Applications: Prescribed Forms and Procedures) Regulations 2009
11-232		National Grid Electricity Transmission PLC
cont'd		1 - 3 Strand
cont u		London
		WC2N 5EH
		(in respect to the right to access as stated in the deed dated 6 November 2009)
		Openreach Limited
		6 Gracechurch Street
		LONDON
		EC3V DAT
		(in respect of underground proposed telecommunication apparatus)
		SP Manweb PLC
		3 Prenton Way
		North Cheshire Trading Estate
		PRENTON
		Merseyside
		CH43 3ET
		(in respect of underground electricity apparatus)
		The Executor of the Estate of the Late David Watkin Williams-Wynn BT
		Forsters LLP
		22 Baker Street
		LONDON
		W1U 3BW
		(in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)
11-233	1439 square metres of agricultural land and hardstanding (south of Glascoed road, B5381)	Burbo Extension Ltd
		5 Howick Place
		LONDON
		SW1P 1WG
		(in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated in the lease dated 30
		October 2014)

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-233		ESP Electricity Limited	
cont'd		1st Floor Bluebird House	
		Mole Business Park	
		Leatherhead	
		Surrey	
		КТ22 7ВА	
		(in respect of the rights as stated in the transfer dated 10 June 2016)	
		Gwynt Y Mor Offshore Wind Farm Limited Windmill Hill Business Park	
		Whitehill Way	
		SWINDON	
		Wiltshire	
		SN5 6PB	
		(in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 6 January 2009)	
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park	
		Benton Lane	
		NEWCASTLE UPON TYNE	
		NE12 8BU	
		(in respect of underground electricity apparatus)	
		National Grid Electricity Transmission PLC 1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect to the right to access as stated in the deed dated 6 November 2009)	
		Openreach Limited	
		6 Gracechurch Street	
		EC3V 0AT (in respect of underground telecommunication apparatus)	
		(in respect of underground teleconfindineation apparatus)	

	Mona Offshore Wind Limited Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire		
Number on	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	·	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
11.222		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-233 cont'd		SP Manweb PLC 3 Prenton Way	
cont u		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
		The Executor of the Estate of the Late David Watkin Williams-Wynn BT	
		Forsters LLP	
		22 Baker Street	
		LONDON	
		W1U 3BW	
		(in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)	
11-233a	783 square metres of agricultural land and hardstanding (south of Glascoed road, B5381)	Burbo Extension Ltd	
11 2550		5 Howick Place	
		LONDON	
		SW1P 1WG	
		(in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated in the lease dated 30	
		October 2014)	
		ESP Electricity Limited	
		1st Floor	
		Bluebird House	
		Mole Business Park	
		Leatherhead	
		Surrey	
		KT22 7BA	
		(in respect of the rights as stated in the transfer dated 10 June 2016)	

	Mona Offshore Wind Limited Development Consent Order			
		BOOK OF REFERENCE - PART 3		
	Counties of Conwy and Denbighshire			
		Names of all those entitled to enjoy easements or other private rights over land		
Number on	Description of Lond	(including private rights of navigation over water) which it is proposed shall be extinguished,		
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning		
		(Applications: Prescribed Forms and Procedures) Regulations 2009		
11-233a		Gwynt Y Mor Offshore Wind Farm Limited		
cont'd		Windmill Hill Business Park		
		Whitehill Way		
		SWINDON		
		Wiltshire		
		SN5 6PB		
		(in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 6 January 2009)		
		Gwynt Y Mor OFTO PLC		
		Q14 Quorum Business Park		
		Benton Lane		
		NEWCASTLE UPON TYNE		
		NE12 8BU		
		(in respect of underground electricity apparatus)		
		National Grid Electricity Transmission PLC		
		1 - 3 Strand		
		London		
		WC2N 5EH		
		(in respect to the right to access as stated in the deed dated 6 November 2009)		
		Openreach Limited		
		6 Gracechurch Street		
		LONDON		
		EC3V OAT		
		(in respect of underground telecommunication apparatus)		
		SP Manweb PLC		
		3 Prenton Way		
		North Cheshire Trading Estate		
		PRENTON		
		Merseyside		
		CH43 3ET		
		(in respect of overhead electricity apparatus)		

	Mona	a Offshore Wind Limited Development Consent Order			
	BOOK OF REFERENCE - PART 3				
	Counties of Conwy and Denbighshire Names of all those entitled to enjoy easements or other private rights over land				
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,			
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning			
Land Fians		(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-233a		The Executor of the Estate of the Late David Watkin Williams-Wynn BT			
cont'd		Forsters LLP			
		22 Baker Street			
		LONDON			
		W1U 3BW			
		(in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)			
11-233b	2013 square metres of agricultural land (south of Glascoed road, B5381)	Burbo Extension Ltd			
11-2550	2015 Square metres of agricultural land (south of Glascoed Toad, 65561)	5 Howick Place			
		LONDON			
		SW1P 1WG			
		(in respect of the right of easement as stated in the Deed dated 28 June 2017 and in respect of the rights as stated in the lease dated 30			
		October 2014)			
1st Floor		ESP Electricity Limited			
1st Floor Bluebird House					
		Mole Business Park			
		Leatherhead			
		Surrey			
		KT22 7BA			
		(in respect of the rights as stated in the transfer dated 10 June 2016)			
		Gwynt Y Mor Offshore Wind Farm Limited			
		Windmill Hill Business Park			
		Whitehill Way			
		SWINDON			
		Wiltshire			
		SN5 6PB			
		(in respect of the rights as stated in the deed dated 16 January 2010 and in respect to rights stated in the deed dated 6 January 2009)			

Mona Offshore Wind Limited Development Consent Order			
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-233b			
11-233b cont'd		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect to the right to access as stated in the deed dated 6 November 2009) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) SP Manweb PLC	
		SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus) The Executor of the Estate of the Late David Watkin Williams-Wynn BT Forsters LLP 22 Baker Street LONDON W1U 3BW (in respect to the right of entry and to shooting rights as stated in the conveyance dated 31 July 1981)	

	Mona Offsi	nore Wind Limited Development Consent Order	
		BOOK OF REFERENCE - PART 3	
		Counties of Conwy and Denbighshire	
Number of		Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,	
Number on Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
11-234	523 square metres of agricultural land and access splay (south of Glascoed road, B5381)	ESP Electricity Limited	
		1st Floor	
		Bluebird House	
		Mole Business Park	
		Leatherhead	
		Surrey	
		KT22 7BA	
		(in respect of right of way stated in Transfer dated 10 June 2016)	
		Gwynt Y Mor OFTO PLC	
		Q14 Quorum Business Park	
		Benton Lane	
		NEWCASTLE UPON TYNE	
(in		NE12 8BU	
		(in respect of underground electricity apparatus)	
		SP Manweb PLC	
		3 Prenton Way	
		North Cheshire Trading Estate	
		PRENTON	
		Merseyside	
		CH43 3ET	
		(in respect of overhead electricity apparatus)	
11-234a	79 square metres of agricultural land (south of Glascoed road, B5381)	ESP Electricity Limited	
		1st Floor	
		Bluebird House	
		Mole Business Park	
		Leatherhead	
		Surrey	
		KT22 7BA	
		(in respect of right of way stated in Transfer dated 10 June 2016)	
		1	

	Mona Offshore Wind Limited Development Consent Order			
	BOOK OF REFERENCE - PART 3			
	Counties of Conwy and Denbighshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
11-234a cont'd		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)		
11-234b	338 square metres of agricultural land (south of Glascoed road, B5381)	ESP Electricity Limited 1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of right of way stated in Transfer dated 10 June 2016) Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) SP Manweb PLC 3 Prenton Way North Cheshire Trading Estate PRENTON Merseyside CH43 3ET (in respect of overhead electricity apparatus)		

	Mona Offshore Wind Limited Development Consent Order				
	BOOK OF REFERENCE - PART 3				
	Counties of Conwy and Denbighshire				
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-235	194 square metres of private road (south of Glascoed road, B5381)	ESP Electricity Limited			
		1st Floor Bluebird House Mole Business Park Leatherhead Surrey KT22 7BA (in respect of right of way stated in Transfer dated 10 June 2016)			
		Gwynt Y Mor OFTO PLC Q14 Quorum Business Park Benton Lane NEWCASTLE UPON TYNE NE12 8BU (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON			
		EC3V 0AT (in respect of underground telecommunication apparatus)			
11-236	1769 square metres of public highway and verges (Glascoed road, B5381)	Denbighshire County Council County Hall Wynnstay Road RUTHIN Clwyd LL15 1YN (in respect of street furniture) Diamond Transmission Partners BBE Limited Mid City Place 71 High Holborn LONDON WC1V 6BA (in respect of underground electricity apparatus)			

	Mona Offshore Wind Limited Development Consent Order				
	BOOK OF REFERENCE - PART 3				
	Counties of Conwy and Denbighshire				
	Names of all those entitled to enjoy easements or other private rights over land				
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,			
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning			
		(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-236		Dŵr Cymru Cyfyngedig			
cont'd		Linea			
		Fortran Road			
		St. Mellons			
		CARDIFF			
		CF3 0LT			
		(in respect of underground water apparatus and underground abandoned water apparatus)			
		Gwynt Y Mor OFTO PLC			
		Q14 Quorum Business Park			
		Benton Lane			
		NEWCASTLE UPON TYNE			
		NE12 8BU			
		(in respect of underground electricity apparatus)			
		(
		Openreach Limited			
		6 Gracechurch Street			
		LONDON			
		EC3V 0AT			
		(in respect of underground telecommunication apparatus)			
		SP Manweb PLC			
		3 Prenton Way			
		North Cheshire Trading Estate			
		PRENTON			
		Merseyside			
		CH43 3ET			
		(in respect of underground and overhead electricity apparatus)			
		Unknown			

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 3 Counties of Conwy and Denbighshire			
Number on Land Plans	Number on Names of all those entitled to enjoy easements or other private rights over land Number on (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning			
11-236 cont'd				

	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 4				
La	Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 Counties of Conwy and Denbighshire				
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made		
01-001	Temporary Possession	177297 square metres of foreshore land seaward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH		
01-002	Temporary Possession	210377 square metres of foreshore land seaward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH		
01-003	Temporary Possession	10945 square metres of foreshore landward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)		
01-007	Temporary Possession	101 square metres of private road (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964)		
01-009	Temporary Possession	1931 square metres of private road and verges (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964)		
01-010	Temporary Possession	386 square metres of private road and verges (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of rights excepted and reserved as contained in a Conveyance dated 16 June 1964)		

Li	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 Counties of Conwy and Denbighshire			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-011	Temporary Possession	11530 square metres of private road and verges (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	
02-012	Acquisition of Rights	156831 square metres of foreshore land seaward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
02-013a	Temporary Possession	58 square metres of private road, hardstanding and grassland (Beach House Road) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	
02-016	Temporary Possession	15 square metres of hardstanding and grassland (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	
02-017	Temporary Possession	8927 square metres of grassland and track (north of Abergele Road, A547) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	
02-018	Temporary Possession	5529 square metres of hardstanding, grassland and track (north of Abergele Road, A547) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	
02-021	Acquisition of Rights	13245 square metres of foreshore land landward of the Mean High Water and cycle way (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	

Li	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 Counties of Conwy and Denbighshire			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
02-021a	Acquisition of Rights	3287 square metres of foreshore land landward of the Mean High Water and cycle way (north of the A55, Abergele) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	
02-02 4	Acquisition of Rights	2517 square metres of grassland (north of the A55, Abergele) (excluding all interests of the crown)	T he King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	
02-026	Acquisition of Rights	1011 square metres of grassland (north of the A55, Abergele) (excluding all interests of the crown)	The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ	
02-027	Acquisition of Rights	19178 square metres of public highway, verges and hedgerow (A55, Abergele) (excluding all interests of the crown)	The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ (as highways authority)	
02-029	Acquisition of Rights	3879 square metres of public highway and verges (Abergele road, A547) (excluding all interests of the crown)	The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ (in respect of subsoil beneath half width of public highway)	
02-030	Temporary Possession	509 square metres of public highway and verges (Abergele road, A547) (excluding all interests of the crown)	The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ (in respect of subsoil beneath half width of public highway)	

L	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 Counties of Conwy and Denbighshire			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
02-034	Acquisition of Rights	1323 square metres of woodland (south of Abergele Road, A547) (excluding all interests of the crown)	The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ	
02-036	Acquisition of Rights	33953 square metres of woodland and access track (south of Abergele Road, A547) (excluding all interests of the crown)	The Welsh Ministers Crown Building Cathays Park CARDIFF CF10 3NQ	
03-060	Acquisition of Rights	5211 square metres of agricultural land (north of Tan-Y-Gopa Road) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	
06-108	Acquisition of Rights	14113 square metres of agricultural land and hedgerow (south of the B5381) and public footpath (FP 19/15) (excluding all interests of the crown)		
07-109	Acquisition of Rights	407 square metres of agricultural land and hedgerow (south of the B5381) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	
07-110	Acquisition of Rights	1391 square metres of public highway and verges (south of the B5381) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	
07-112	Acquisition of Rights (Hedgerow)	734 square metres of public highway and hedgerow (south of the B5381) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	

Li	Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 Counties of Conwy and Denbighshire			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
07-120	Acquisition of Rights	13556 square metres of agricultural land (south of the B5381) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	
07-121	Acquisition of Rights (Hedgerow)	924 square metres of hedgerow (south of the B5381) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	
07-122	Acquisition of Rights (Hedgerow)	586 square metres of hedgerow (south of the B5381) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	
07-123	Acquisition of Rights (Hedgerow)	3859 square metres of public highway and hedgerow (south of the B5381) (excluding all interests of the crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (in respect of mines and minerals)	

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 5 Counties of Conwy and Denbighshire				
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land	
01-001	Temporary Possession	177297 square metres of foreshore land seaward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	Open Space	
01-002	Temporary Possession	210377 square metres of foreshore land seaward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	Open Space	
01-003	Temporary Possession	10945 square metres of foreshore landward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	Open Space	
01-004	Temporary Possession	167 square metres of foreshore landward of the Mean High Water (north of the A55, Abergele)	Open Space	
01-006	Temporary Possession	395 square metres of private road (north of the A55, Abergele)	Open Space	
01-007	Temporary Possession	101 square metres of private road (north of the A55, Abergele) (excluding all interests of the crown)	Open Space	
01-008	Temporary Possession	325 square metres of private road (north of the A55, Abergele)	Open Space	
01-009	Temporary Possession	1931 square metres of private road and verges (north of the A55, Abergele) (excluding all interests of the crown)	Open Space	
01-010	Temporary Possession	386 square metres of private road and verges (north of the A55, Abergele) (excluding all interests of the crown)	Open Space	
01-011	Temporary Possession	11530 square metres of private road and verges (north of the A55, Abergele) (excluding all interests of the crown)	Open Space	
02-012	Acquisition of Rights	156831 square metres of foreshore land seaward of the Mean High Water (north of the A55, Abergele) (excluding all interests of the crown)	Open Space	
02-021	Acquisition of Rights	13245 square metres of foreshore land landward of the Mean High Water and cycle way (north of the A55, Abergele) (excluding all interests of the crown)	Open Space	
02-021a	Acquisition of Rights	3287 square metres of foreshore land landward of the Mean High Water and cycle way (north of the A55, Abergele) (excluding all interests of the crown)	Open Space	
02-022	Acquisition of Rights	4484 square metres of grassland (lying to the north of the A55, Abergele)	Open Space	

Mona Offshore Wind Limited Development Consent Order BOOK OF REFERENCE - PART 5 Counties of Conwy and Denbighshire					
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land		
02-023	Acquisition of Rights	665 square metres of grassland (north of the A55, Abergele)	Open Space		